# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

EO	DIA	Ω	TZ
ΡU	$\mathbf{RM}$	Q.	-N

CURRENT REPORT
Pursuant to Section 13 or 15(d)
of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): June 25, 2021

# Origin Materials, Inc.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation) 001-39378 (Commission File Number) 87-1388928 (IRS Employer Identification No.)

930 Riverside Parkway, Suite 10 West Sacramento, CA (Address of principal executive offices)

95606 (Zip Code)

Registrant's telephone number, including area code: +1 (916) 231-9329

N/A

(Former Name or Former Address, if Changed Since Last Report)

	Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligations of the registrant under any of the following provisions:						
	Written communications pursuant to Rule 425 under t	he Securities Act (17 CFR 230.425)					
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)						
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))						
	☐ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))						
Seci	Securities registered pursuant to Section 12(b) of the Act:						
	Title of each class	Trading Symbol(s)	Name of each exchange on which registered				
C	ommon Stock, par value \$0.0001 per share	ORGN	The Nasdaq Stock Market LLC				
W	Warrants, each whole warrant exercisable for ORGNW The Nasdaq Stock Market LLC						
one	share of Common Stock at an exercise price						
	of \$11.50 per share						

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company ⊠

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

#### **Introductory Note**

#### **Domestication and Merger Transaction**

As previously announced, Micromidas, Inc., a Delaware corporation doing business as Origin Materials ("*Legacy Origin*"), Artius Acquisition Inc., a Cayman Islands exempted company ("*Artius*" and, after the Domestication as described below, the "*Company*"), and Zero Carbon Merger Sub, Inc., a Delaware corporation and wholly owned subsidiary of Artius ("*Merger Sub*"), entered into an Agreement and Plan of Merger and Reorganization, dated as of February 16, 2021 and amended by the letter agreement dated March 5, 2021 (the "*Merger Agreement*").

On June 24, 2021, as previously disclosed and as contemplated by the Merger Agreement and described in the final prospectus and definitive proxy statement, dated May 27, 2021 (the "*Proxy Statement/Prospectus*") and filed with the Securities and Exchange Commission (the "*SEC*") in the section titled "*Proposal No. 1—The Domestication Proposal*" beginning on page 259 of Proxy Statement/Prospectus, Artius filed a notice of deregistration with the Cayman Islands Registrar of Companies, together with the necessary accompanying documents, and filed a certificate of incorporation and a certificate of corporate domestication with the Secretary of State of the State of Delaware, under which Artius was domesticated and continues as a Delaware corporation, changing its name to "Origin Materials, Inc." (the "*Domestication*").

As a result of and upon the effective time of the Domestication, among other things:

- each then issued and outstanding Class A ordinary share, par value \$0.0001 per share, of Artius (the "Artius Class A Ordinary Shares") converted automatically by operation of law, on a one-for-one basis, into a share of Class A common stock, par value \$0.0001 per share, of Artius (the "Artius Class A Common Stock");
- each then issued and outstanding redeemable warrant that was registered pursuant to the Registration Statements on Form S-1 (SEC File Nos. 333-239421 and 333-239841) of Artius automatically became a redeemable warrant to acquire shares of Artius Class A Common Stock (no changes were made to the terms of any issued and outstanding public warrants as a result of the Domestication);
- each then issued and outstanding unit of Artius that had not been previously separated into the underlying Artius Class A Ordinary Share and underlying warrant upon the request of the holder thereof, automatically entitled the holder thereof to one share of Artius Class A Common Stock and one-third of one redeemable warrant to acquire one share of Artius Class A Common Stock;
- each then issued and outstanding Class B ordinary share, par value \$0.0001 per share, of Artius converted automatically by operation of law, on a one-for-one basis without giving effect to any rights of adjustment or other anti-dilution protections, into a share of Class B common stock, par value \$0.0001 per share, of Artius (the "Artius Class B Common Stock"); and
- the issued and outstanding warrants of Artius issued in a private placement automatically became warrants to acquire shares of Artius Class A Common Stock.

On June 25, 2021 (the "Closing Date") and following the approval at an extraordinary general meeting of the shareholders of Artius held on June 23, 2021 (the "Special Meeting"), as contemplated by the Merger Agreement and described in the Proxy Statement/Prospectus in the section titled "Proposal No. 2—The Transaction Proposal" beginning on page 273 of the Proxy Statement/Prospectus, the parties consummated the closing of the transactions contemplated by the Merger Agreement, whereby Merger Sub merged with and into Legacy Origin, the separate corporate existence of Merger Sub ceasing and Legacy Origin being the surviving corporation and a wholly owned subsidiary of the Company (the "Merger" and, together with the other transactions contemplated by the Merger Agreement, the "Business Combination"). Upon the closing of the Business Combination, the outstanding shares of Artius Class A Common Stock and Artius Class B Common Stock became shares of common stock of the Company ("Common Stock"), and the warrants to acquire shares of Artius Class A Common Stock became warrants to acquire Common Stock.

In connection with the Special Meeting and the Business Combination, the holders of 43,880,956 shares of Artius Class A Ordinary Shares, or 60.6% of the shares with redemption rights, exercised their right to redeem their shares for cash at a redemption price of approximately \$10.0040 per share, for an aggregate redemption amount of \$438,983,332.52.

At the effective time of the Merger (the "*Effective Time*"), by virtue of the Merger and without any action on the part of Artius, Merger Sub, Legacy Origin or the holders of any of Legacy Origin's securities, each outstanding share of Legacy Origin common stock, Series A preferred stock, Series B preferred stock and Series C Preferred Stock was automatically converted into the right to receive a number of shares of Common Stock equal to the Common Exchange Ratio, Series A Exchange Ratio, Series B Exchange Ratio and Series C Exchange Ratio, respectively (subject to certain adjustments as described in the Merger Agreement), as described below:

- The "*Common Exchange Ratio*" means (i) the quotient of 78,213,000 (minus adjustments as described in the Proxy Statement/Prospectus) minus the Aggregate Liquidation Preference Consideration (as defined in the Merger Agreement) (ii) divided by Legacy Origin's fully diluted shares outstanding (as defined in the Merger Agreement), as of immediately prior to the Effective Time.
- The "Series A Exchange Ratio" means (i) the quotient of the Series A Per Share Liquidation Preference (as defined in the Merger Agreement) divided by the per share value of Artius Class A Common Stock (as adjusted pursuant to the Merger Agreement) calculated in accordance with the Amended and Restated Certificate of Incorporation of Legacy Origin (the "Artius Trading Price") (plus adjustments as described in the Proxy Statement/Prospectus) (ii) plus the Per Share Residual Consideration (as defined in the Merger Agreement).
- The "Series B Exchange Ratio" means (i) the quotient of the Series B Per Share Liquidation Preference (as defined in the Merger Agreement) divided by the Artius Trading Price (plus adjustments as described in this proxy statement/prospectus) (ii) plus the Per Share Residual Consideration.
- The "Series C Exchange Ratio" means (i) the quotient of the Series C Per Share Liquidation Preference (as defined in the Merger Agreement) divided by the Artius Trading Price (plus adjustments as described in the Proxy Statement/Prospectus) (ii) plus the Per Share Residual Consideration.

On the Closing Date, the Common Exchange Ratio was approximately 2.1165, the Series A Exchange Ratio was approximately 2.5585, the Series B Exchange Ratio was approximately 3.1480 and the Series C Exchange Ratio was approximately 3.9520. In addition, certain holders of Legacy Origin's securities will be entitled to receive, in the aggregate, up to 25,000,000 shares of Common Stock subject to certain earnout provisions.

A description of the Business Combination and the terms of the Merger Agreement are included in the section titled "*The Merger Agreement and Related Agreements*" beginning on page 131 of the Proxy Statement/Prospectus. The foregoing description of the Merger Agreement is a summary only and is qualified in its entirety by the full text of the Merger Agreement, a copy of which is attached hereto as Exhibit 2.1 and incorporated herein by reference.

#### **PIPE Subscription Agreements**

On the Closing Date, certain purchasers (each, a "Subscriber") purchased from the Company an aggregate of 20,000,000 shares of Common Stock (the "PIPE Shares"), for a purchase price of \$10.00 per share and an aggregate purchase price of \$200.0 million, pursuant to separate subscription agreements, dated February 16, 2021 (collectively, the "Subscription Agreements"). Pursuant to the Subscription Agreements, the Company agreed to provide the Subscribers with certain registration rights with respect to the PIPE Shares.

A description of the Subscription Agreements is included in the Proxy Statement/Prospectus in the section titled "The Merger Agreement and Related Agreements—Other Agreements Related to the Merger Agreement—Subscription Agreements" beginning on page 146 of the Proxy Statement/Prospectus. The foregoing description of the Subscription Agreements is a summary only and is qualified in its entirety by the full text of the form of Subscription Agreement, a copy of which is filed hereto as Exhibit 10.1 and incorporated herein by reference.

#### **Backstop Agreement and Additional Subscription Agreements**

On the Closing Date, certain funds managed by affiliates of Apollo Capital Management, L.P. (collectively, the "*Apollo Entities*") purchased 3,000,000 shares of Common Stock (the "*Backstop Shares*"), for a purchase price of \$10.00 per share and an aggregate purchase price of \$30.0 million, pursuant to a purchase agreement dated June 14, 2021 (the "*Backstop Agreement*"). Pursuant to the Backstop Agreement, the Company agreed to provide certain registration rights to the Apollo Entities with respect to the Backstop Shares. In connection with and subject to satisfaction of the conditions contained in the Backstop Agreement, the Apollo Entities were paid a backstop premium of \$900,000.

On the Closing Date, certain purchasers (each, an "Additional Subscriber") purchased from the Company an aggregate of 1,300,001 shares of Common Stock (the "Additional Subscription Shares"), for a purchase price of \$10.00 per share and an aggregate purchase price of \$13.0 million, pursuant to a purchase agreement dated June 23, 2021 (collectively, the "Additional Subscription Agreements"). Pursuant to the Additional Subscription Agreements, the Company agreed to provide certain registration rights to the Additional Subscribers with respect to the Additional Subscription Shares.

The foregoing descriptions of the Backstop Agreement and the Additional Subscription Agreements are summaries only and are qualified in their entirety by the full text of the forms of Backstop Agreement and Additional Subscription Agreements, copies of which are filed hereto as 10.2 and 10.3, respectively, and are incorporated herein by reference.

# Item 1.01 Entry into a Material Definitive Agreement.

#### **Lock-Up Agreements**

In connection with the Business Combination, Artius, certain shareholders of Artius and certain stockholders, officers and directors of Legacy Origin entered into a Lock-Up Agreement (each, a "Lock-Up Agreement"). The terms of the Lock-up Agreement are described in the Proxy Statement/Prospectus in the section titled "The Merger Agreement and Related Agreements—Other Agreements Related to the Merger Agreement—Lock-Up Agreement" beginning on page 148 of the Proxy Statement/Prospectus. The holders of 35,765,099 shares of Common Stock are subject to a Lock-Up Agreement.

The foregoing description of the Lock-Up Agreement is qualified in its entirety by the full text of the form of Lock-Up Agreement, a copy of which is filed hereto as Exhibit 10.4 and incorporated herein by reference.

#### **Investor Rights Agreement**

Certain persons and entities receiving shares of Common Stock pursuant to the Merger Agreement and certain persons and entities holding securities of Artius prior to the closing entered into an investor rights agreement (the "Investor Rights Agreement"), which became effective on the Closing Date. The terms of the Investor Rights Agreement are described in the Proxy Statement/Prospectus in the section titled "The Merger Agreement and Related Agreements—Other Agreements Related to the Merger Agreement—Investor Rights Agreement" beginning on page 148 of the Proxy Statement/Prospectus. Following the closing, the holders of approximately 3.2 million shares of Common Stock (including up to 11,326,667 shares issuable upon the exercise of warrants to purchase Common Stock) are entitled to certain registration rights. Additionally, the Merger Agreement provides that these holders are eligible to receive up to 2,015,745 additional shares of Common Stock subject to certain earnout provisions described in the Merger Agreement and, if issued, such shares would be entitled to registration rights under the Investor Rights Agreement.

The foregoing description of the Investor Rights Agreement is qualified in its entirety by reference to the full text of the form of Investor Rights Agreement, a copy of which is attached hereto as Exhibit 10.5 and incorporated herein by reference.

# **Indemnification Agreements**

On the Closing Date, the Company entered into indemnification agreements with each of its directors and executive officers. These indemnification agreements require the Company to indemnify its directors and executive officers for certain expenses, including attorneys' fees, judgments, fines and settlement amounts incurred by a director or executive officer in any action or proceeding arising out of their services as one of the Company's directors or executive officers or any other company or enterprise to which the person provides services at the Company's request.

The foregoing description of the indemnification agreements is qualified in its entirety by the full text of the form of indemnification agreement, a copy of which is filed hereto as Exhibit 10.6 and incorporated herein by reference.

#### Item 2.01 Completion of Acquisition of Disposition of Assets.

The disclosure set forth in the "Introductory Note" above is incorporated by reference into this Item 2.01 of this Current Report on Form 8-K.

As of the Closing Date and following the closing of the Business Combination, the Company had the following outstanding securities:

- 141,248,470 shares of Common Stock; and
- 35,476,667 warrants, each exercisable for one share of Common Stock at a price of \$11.50 per share (the "*Warrants*").

#### **FORM 10 INFORMATION**

Item 2.01(f) of the Current Report on Form 8-K states that if the predecessor registrant was a shell company, as Artius was immediately before the Business Combination, then the registrant must disclose the information that would be required if the registrant were filing a general form for registration of securities on Form 10. Accordingly, the Company, as the successor registrant to Artius, is providing the information below that would be included in a Form 10 if it were to file a Form 10. Please note that the information provided below relates to the combined company after the closing of the Business Combination, unless otherwise specifically indicated or the context otherwise requires.

### **Forward-Looking Statements**

The Company makes forward-looking statements in this Current Report on Form 8-K and in documents incorporated herein by reference. Forward-looking statements relate to expectations, beliefs, projections, future plans and strategies, anticipated events or trends and similar expressions concerning matters that are not historical facts. These forward-looking statements include statements about future financial and operating results of the Company; benefits of the Business Combination; statements about the plans, strategies and objectives of management for future operations of the Company; statements regarding future performance; and other statements regarding the Business Combination. In some cases, you can identify these forward-looking statements by the use of terminology such as "outlook," "believes," "expects," "potential," "continues," "may," "will," "should," "could," "seeks," "approximately," "predicts," "intends," "plans," "estimates," "anticipates" or the negative version of these words or other comparable words or phrases.

These forward-looking statements are based on information available as of the date of this Current Report on Form 8-K, and current expectations, forecasts and assumptions, and involve a number of risks and uncertainties. Accordingly, forward-looking statements in this Current Report on Form 8-K and in any document incorporated herein by reference should not be relied upon as representing the Company's views as of any subsequent date, and the Company does not undertake any obligation to update forward-looking statements to reflect events or circumstances after the date they were made, whether as a result of new information, future events or otherwise, except as may be required under applicable securities laws.

As a result of a number of known and unknown risks and uncertainties, the Company's actual results or performance may be materially different from those expressed or implied by these forward-looking statements. Some factors that could cause actual results to differ include:

- changes in personnel and availability of qualified personnel;
- the effects of the ongoing coronavirus (COVID-19) pandemic or other infectious diseases, health epidemics, pandemics and natural disasters on the Company's business;
- the amount and timing of future sales;
- the Company's ability to secure additional project financing and government incentives;

- the Company's ability to complete construction of its plants in the expected timeframe and in a cost-effective manner;
- the Company's ability to procure necessary capital equipment and to produce its products in large commercial quantities;
- any decline in the value of carbon credits;
- increases or fluctuations in raw material costs;
- the ability to recognize the anticipated benefits of the Business Combination, which may be affected by, among other things, competition, and the ability of the Company to manage its growth and expand its business operations effectively following the consummation of the Business Combination;
- whether the concentration of the Company's stock ownership and voting power limits the stockholders of the Company's ability to influence corporate matters;
- the ability to maintain the listing of Common Stock on the Nasdaq following the Business Combination; and
- the increasingly competitive environment in which the Company will operate.

Please see the other risks and uncertainties set forth in the Proxy Statement/Prospectus in the section titled "*Risk Factors*" beginning on page 52 of the Proxy Statement/Prospectus and incorporated herein by reference.

In addition, statements that "Origin believes" or "Artius believes" and similar statements reflect Origin's or Artius's beliefs and opinions on the relevant subject. These statements are based upon information available to Origin or Artius, as the case may be, as of the date of the Proxy Statement/Prospectus, and while Origin or Artius, as the case may be, believes such information forms a reasonable basis for such statements, such information may be limited or incomplete, and such statements should not be read to indicate that such party has conducted an exhaustive inquiry into, or review of, all potentially available relevant information. These statements are inherently uncertain and investors are cautioned not to unduly rely upon these statements.

#### **Business and Properties**

The business and properties of Artius and Legacy Origin prior to the Business Combination are set forth in the Proxy Statement/Prospectus in the sections titled "Information About Artius" and "Information About Origin" beginning on pages 169 and 185, respectively, of the Proxy Statement/Prospectus and incorporated herein by reference.

#### Risk Factors

The risks associated with the Company's business are set forth in the Proxy Statement/Prospectus in the section titled "*Risk Factors*" beginning on page 52 of the Proxy Statement/Prospectus and incorporated herein by reference.

# **Selected Historical Financial Information**

The selected historical consolidated financial information and other data for the years ended December 31, 2019 and 2020 and for the three months ended March 31, 2021 for Legacy Origin are included in the Proxy Statement/Prospectus in the section titled "Selected Historical Financial Information of Origin" beginning on page 152 of the Proxy Statement/Prospectus and incorporated herein by reference.

#### **Unaudited Pro Forma Condensed Combined Financial Information**

The unaudited pro forma condensed combined financial information of the Company as of and for the year ended December 31, 2020 and as of and for the three months ended March 31, 2021 is attached hereto as Exhibit 99.2 and incorporated herein by reference.

# Management's Discussion and Analysis of Financial Condition and Results of Operations

Management's Discussion and Analysis of Financial Condition and Results of Operations of Legacy Origin prior to the Business Combination is set forth in the Proxy Statement/Prospectus in the section titled "Origin Management's Discussion and Analysis of Financial Condition and Results of Operations" beginning on page 208 of the Proxy Statement/Prospectus and incorporated herein by reference.

#### **Directors and Executive Officers**

The Company's directors and executive officers after the closing of the Business Combination are set forth in the Proxy Statement/Prospectus in the section titled "Management of the Combined Company" beginning on page 221 of the Proxy Statement/Prospectus and incorporated herein by reference.

#### **Executive Compensation**

Information with respect to the compensation of the Company's executive officers is described in the Proxy Statement/Prospectus in the section titled "Management of Origin" beginning on page 198 of the Proxy Statement/Prospectus and incorporated herein by reference. The description of the compensation of the Company's executive officers is qualified in its entirety by the full text of the employment agreements of Messrs. Riley, Lee and Whaley, copies of which are attached hereto as Exhibits 10.15, 10.16 and 10.17, respectively, and incorporated herein by reference.

#### **Director Compensation**

The compensation of Legacy Origin's non-employee directors is described in the Proxy Statement/Prospectus in the section titled "Management of the Combined Company—Non-Employee Director Compensation" beginning on page 227 of the Proxy Statement/Prospectus and incorporated herein by reference.

In connection with the consummation of the Business Combination, the board of directors of the Company adopted a non-employee director compensation policy, pursuant to which each non-employee director is entitled to a \$50,000 annual cash retainer. In addition, the members of the Audit Committee, Compensation Committee, and Nominating and Corporate Governance Committee will be entitled to an annual cash retainer of \$10,000, \$5,000, and \$5,000, respectively, with the chair of each such committee being entitled to an annual cash retainer of \$20,000, \$15,000, and \$15,000, respectively.

Following the closing of the Business Combination:

- Each non-employee director will also receive a one-time initial grant of restricted stock units with a grant date value of \$150,000, which
  will vest in three equal installments on the first, second and third anniversary of the grant date; and
- Each non-employee director will also receive an annual grant of restricted stock units with a grant date value of \$130,000, with the chairperson of the board of directors to receive an additional annual grant of restricted stock units with a grant date value of \$75,000, which awards will vest on the first anniversary of the grant date.

Each award described above is subject to the applicable individual continuing to serve on the Company's board of directors through the vesting date. In addition, each member of the board of directors is required to acquire and hold shares of common stock of the Company with a fair market value of at least \$250,000 by the later of the fifth anniversary of (i) the closing of the Business Combination and (ii) such director's election to the board of directors.

The foregoing description of the non-employee director compensation policy is qualified in its entirety by the full text of the policy, a copy of which is attached hereto as Exhibits 10.7 and incorporated herein by reference.

Each member of the Company's board of directors will also be covered by the Company's liability insurance and has entered into the Company's standard form of indemnification agreement. Please see "Item 1.01 - Entry into a Material Definitive Agreement – Indemnification Agreements" of this Current Report on Form 8-K for a description of such agreements.

# Security Ownership of Certain Beneficial Owners and Management

The following table sets forth information regarding the beneficial ownership of shares of Common Stock as of the Closing Date, after giving effect to the closing of the Business Combination, by:

- each person known by the Company to be the beneficial owner of more than 5% of Common Stock;
- · each of the Company's executive officers and directors; and
- all of the Company's executive officers and directors as a group.

Beneficial ownership is determined according to the rules of the SEC, which generally provide that a person has beneficial ownership of a security if he, she or it possesses sole or shared voting or investment power over that security, including options, warrants and restricted stock units that are currently exercisable or vested or that will become exercisable or vest within 60 days. This table is based upon information supplied by officers, directors and principal stockholders and Schedules 13G or 13D filed with the SEC. Unless otherwise indicated in the footnotes to this table and subject to community property laws where applicable, the Company believes that all persons named in the table have sole voting and investment power with respect to all shares of Common Stock beneficially owned by them. The beneficial ownership percentages set forth in the table below are based on 141,248,470 shares of Common Stock issued and outstanding as of the Closing Date.

Name and Address of Beneficial Owner(1)	Number of Shares	Percentage of Common Stock Outstanding
5% or Greater Stockholders:		
Artius Acquisition Partners LLC(2)	18,112,500	12.8%
Lior Amram(3)	10,142,100	7.2
Named Executive Officers and Directors:		
John Bissell(4)	1,649,515	1.1
Rich Riley <sup>(5)</sup>	2,013,190	1.4
Nate Whaley(6)	187,395	*
Joshua Lee(7)	105,823	*
William Harvey(8)	158,734	*
Karen Richardson	<del>_</del>	_
Boon Sim(2)	18,112,500	12.8
Charles D. Drucker(2)(9)	18,862,500	13.4
Kathleen B. Fish	<del>_</del>	_
Benno O. Dorer	<del>_</del>	_
Pia Heidenmark Cook	<del>-</del>	_
All current directors and executive officers as a group (12 persons)	23,937,116	17.0

<sup>\*</sup> Less than one percent.

- (1) Unless otherwise indicated, the business address of each of the directors, executive officers and unnamed officers of the Company is C/O Origin Materials, Inc., 930 Riverside Parkway, Suite 10, West Sacramento, CA 95606.
- (2) Artius Acquisition Partners LLC is the record holder of the shares reported herein. Mr. Sim and Mr. Drucker are the founding members of Artius Acquisition Partners LLC and together exercise voting and investment power with respect to the Common Stock held by Artius Acquisition Partners LLC. The shares beneficially owned by Artius Acquisition Partners LLC may also be deemed to be beneficially owned by Mr. Sim and Mr. Drucker.
- (3) Consists of (i) 33,843 shares of Common Stock held directly by Mr. Amram; (ii) 9,684,510 shares of Common Stock held by Evergreen InvestCo I, LLC ("Evergreen InvestCo I"); (iii) 59,373 shares of Common Stock held by JLA Construction LLC 401k Plan ("JLA Construction"); (iv) 186,612 shares of Common Stock held by Evergreen Capital, L.P. ("Evergreen Capital"); (v) 130,000 shares of Common Stock held by Evergreen Acquisition I Corp ("Evergreen Acquisition"); and (vi) 47,762 shares of Common Stock held by JLA Asset Management LLC ("JLA Asset Management"). Mr. Amram is the sole manager of each of Evergreen InvestCo I, Evergreen Capital, Evergreen Acquisition and JLA Construction, and the managing member of JLA Asset Management, and may be deemed to hold sole voting and dispositive power over the Common Stock shares held by these entities. With respect to the shares of Common Stock held by these entities, Mr. Amram disclaims beneficial ownership other than to the extent he may have a pecuniary interest therein, directly or indirectly. The principal business address for Mr. Amram is c/o Evergreen Capital, L.P. 551 Fifth Avenue, Suite 2100, New York, New York 10176.
- (4) Consists of (i) 634,943 shares of Common Stock held directly by Mr. Bissell and (ii) 1,014,572 shares of Common Stock issuable to Mr. Bissell pursuant to options exercisable within 60 days of June 25, 2021.
- (5) Consists of (i) 886,275 shares of Common Stock issuable to Mr. Riley pursuant to options exercisable within 60 days of June 25, 2021; (ii) 189,668 shares of Common Stock held by Riley Family Trust; (iii) 229,415 shares of Common Stock held by Riley Investment Trust I; and (iv) 707,832 shares of Common Stock held by Riley Separate Property Trust. Mr. Riley is co-trustee of the Riley Family Trust and by virtue of his shared control over Riley Family Trust, may be deemed to beneficially own the shares of Common Stock held by Riley Family Trust. Mr. Riley is sole trustee of each of Riley Investment Trust I and Riley Separate Property Trust and may be deemed to hold sole voting and dispositive power over the Common Stock shares held by Riley Investment Trust I and Riley Separate Property Trust.
- (6) Consists of 187,395 shares of Common Stock issuable to Mr. Whaley pursuant to options exercisable within 60 days of June 25, 2021.
- (7) Consists of 105,823 shares of Common Stock issuable to Mr. Lee pursuant to options exercisable within 60 days of June 25, 2021.
- (8) Consists of 158,734 shares of Common Stock issuable to Mr. Harvey pursuant to options exercisable within 60 days of June 25, 2021.
- (9) Includes 750,000 shares of Common Stock held directly by Mr. Drucker.

# Certain Relationships and Related Business Combination

Certain relationships and related party transactions are described in the Proxy Statement/Prospectus in the section titled "Certain Relationships and Related Party Transactions" beginning on page 249 of the Proxy Statement/Prospectus and incorporated herein by reference.

#### Legal Proceedings

There is no material litigation, arbitration or governmental proceeding currently pending against the Company or any members of its management team.

### Market Price of and Dividends on the Registrant's Common Equity and Related Stockholder Matters

#### **Market Information and Holders**

The Artius Class A Ordinary Shares, units and warrants were historically quoted on The Nasdaq Capital Market under the symbols "AACQ," "AACQU" and "AACQW," respectively. The Common Stock and Warrants began trading on The Nasdaq Capital Market under the new trading symbols "ORGN" and "ORGNW," respectively, on June 25, 2021.

In connection with the closing, each Artius unit was separated into its components, which consisted of one share of Artius Class A Common Stock and one-third of one redeemable warrant to acquire one share of Artius Class A Common Stock, and such units no longer exist. Upon the closing of the Business Combination, the outstanding shares of Artius Class A Common Stock became shares of Common Stock, and the warrants to acquire shares of Artius Class A Common Stock became Warrants to open purchase shares of Common Stock. As of the Closing Date, and following the closing of the Business Combination, the Company had 141,248,470 shares of the Common Stock issued and outstanding held of record by 209 holders, and 35,476,667 Warrants outstanding held of record by two holders.

#### Dividends

The Company has not paid dividends on its Common Stock to date and does not intend to pay cash dividends in the foreseeable future. The payment of cash dividends in the future will be dependent upon revenues and earnings, if any, capital requirements and general financial condition. The payment of any dividends will be within the discretion of the Company's board of directors. It is the present intention of the Company's board of directors to retain all earnings, if any, for use in the Company's business operations and, accordingly, the board of directors does not anticipate declaring any dividends in the foreseeable future.

#### **Recent Sales of Unregistered Securities**

Reference is made to the disclosure set forth under Item 3.02 of this Current Report on Form 8-K concerning recent sales of unregistered securities.

# **Description of Registrant's Securities**

#### Common Stock

A description of the Common Stock is included in the Proxy Statement/Prospectus in the section titled "Description of Securities—Authorized and Outstanding Stock" beginning on page 229 of the Proxy Statement/Prospectus and incorporated herein by reference.

#### Warrants

A description of the Warrants is included in the Proxy Statement/Prospectus in the section titled "Description of Securities—Warrants" beginning on page 230 of the Proxy Statement/Prospectus and incorporated herein by reference.

# **Indemnification of Directors and Officers**

In connection with the Business Combination, the Company entered into indemnification agreements with each of its directors and executive officers. These indemnification agreements provide such directors and executive officers with contractual rights to indemnification and expense advancement.

The foregoing summary is qualified in its entirety by reference to the text of the form of Indemnification Agreement, a copy of which is attached hereto as Exhibit 10.6 and incorporated herein by reference.

#### **Financial Statements and Supplementary Data**

Reference is made to the disclosure set forth under Item 9.01 of this Current Report on Form 8-K concerning the Company's consolidated financial statements and supplementary data.

### Changes in and Disagreements with Accountants on Accounting and Financial Disclosure

Reference is made to the disclosure set forth under Item 4.01 of this Current Report on Form 8-K concerning the changes in certifying accountant.

# **Financial Statements and Exhibits**

The information set forth in Item 9.01 of this Current Report on Form 8-K is incorporated herein by reference.

# Item 3.02 Unregistered Sales of Equity Securities.

The disclosures set forth in the "Introductory Note" above with respect to the Subscription Agreements, the Backstop Agreement and the Additional Subscription Agreements are incorporated by reference into this Item 3.02 of this Current Report on Form 8-K.

The securities issued in connection with the Subscription Agreements, the Backstop Agreement and the Additional Subscription Agreements have not been registered under the Securities Act of 1933, as amended (the "*Securities Act*") in reliance on the exemption from registration provided by Section 4(a)(2) of the Securities Act.

# Item 3.03 Material Modification to Rights of Security Holders.

The information set forth in Item 5.03 of this Current Report on Form 8-K is incorporated herein by reference.

#### Item 4.01 Changes in Registrant's Certifying Accountant.

On June 25, 2021, the Audit Committee of the Company's board of directors approved the engagement of Grant Thornton LLP ("*Grant Thornton*") as the Company's independent registered public accounting firm to audit the Company's consolidated financial statements for the year ending December 31, 2021. Grant Thornton previously served as the independent registered public accounting firm of Legacy Origin prior to the Business Combination. Accordingly, Marcum LLP ("*Marcum*"), Artius's independent registered public accounting firm prior to the Business Combination, was informed on June 25, 2021 that it would be replaced by Grant Thornton as the Company's independent registered public accounting firm.

Marcum's report of independent registered public accounting firm dated March 4, 2021, except for the effects of the restatement discussed in Note 2 to the financial statements in Amendment No. 3 to Registration Statement on Form S-4, dated May 25, 2021, filed by the Company with the SEC, as to which the date is April 30, 2021, on the Artius consolidated balance sheet as of December 31, 2020, the related consolidated statements of operations, changes in stockholders' equity and cash flows for the period from January 24, 2020 (Artius's inception) through December 31, 2020 and the related notes to the financial statements did not contain any adverse opinion or disclaimer of opinion, and were not qualified or modified as to uncertainties, audit scope or accounting principles.

During the period from January 24, 2020 (Artius's inception) through December 31, 2020 and the subsequent interim period through June 25, 2021, there were no "disagreements" (as such term is defined in Item 304(a)(1)(iv) of Regulation S-K) with Marcum on any matter of accounting principles or practices, financial statement disclosure, or auditing scope or procedures, which disagreements, if not resolved to the satisfaction of Marcum, would have caused Marcum to make reference thereto in its reports on Artius's financial statements for such periods. During the period from January 24, 2020 (Artius's inception) through December 31, 2020 and the subsequent interim period through June 25, 2021, there have been no "reportable events" (as such term is defined in Item 304(a)(1)(v) of Regulation S-K), other than the material weakness in internal controls identified by management related to the accounting for warrants issued in connection with Artius' initial public offering, which resulted in the restatement of Artius's financial statements as set forth Amendment No. 1 to Artius' Form 10-K for the year ended December 31, 2020, as filed with the SEC on May 3, 2021.

During the period from January 24, 2020 (Artius's inception) through December 31, 2020 and the subsequent interim period through June 25, 2021, (i) the Company did not both (a) consult with Grant Thornton as to the application of accounting principles to a specified transaction, either completed or proposed, or the type of audit opinion that might be rendered on the Company's consolidated financial statements and (b) receive a written report or oral advice that Grant Thornton concluded was an important factor considered by the Company in reaching a decision as to such accounting, auditing, or financial reporting issue; and (ii) the Company did not consult Grant Thornton on any matter that was either the subject of a "disagreement" (as that term is defined in Item 304(a)(1)(iv) of Regulation S-K and the related instructions) or a "reportable event" (as that term is defined in Item 304(a)(1)(v) of Regulation S-K).

The Company has provided Marcum with a copy of the disclosures made by the registrant in this Item 4.01 in response to Item 304(a) of Regulation S-K under the Exchange Act of 1934, as amended (the "*Exchange Act*") and requested that Marcum furnish the Company with a letter addressed to the SEC stating whether it agrees with the statements made by the registrant in this Item 4.01 in response to Item 304(a) of Regulation S-K under the Exchange Act and, if not, stating the respects in which it does not agree. A letter from Marcum is attached hereto as Exhibit 16.1.

#### Item 5.01 Changes in Control of Registrant.

The information set forth in the section titled "*Introductory Note*" and in the section titled "*Security Ownership of Certain Beneficial Owners and Management*" in Item 2.01 of this Current Report on Form 8-K is incorporated herein by reference.

As a result of the completion of the Business Combination pursuant to the Merger Agreement, a change of control of Artius has occurred, and the shareholders of Artius as of immediately prior to the closing held 33.0% of the outstanding shares of Common Stock immediately following the closing.

#### Item 5.02 Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

The information set forth in the sections titled "Directors and Executive Officers" and "Certain Relationships and Related Transactions" in Item 2.01 of this Current Report on Form 8-K is incorporated herein by reference.

# 2021 Equity Incentive Plan

At the Special Meeting, the Artius shareholders considered and approved the 2021 Equity Incentive Plan (the "*Equity Incentive Plan*"). The Equity Incentive Plan was previously approved, subject to shareholder approval, by Artius's board of directors on May 25, 2021. The Equity Incentive Plan became effective upon the closing of the Business Combination.

A summary of the terms of the Equity Incentive Plan is set forth in the Proxy Statement/Prospectus in the section titled "*Proposal No. 7—The Equity Incentive Plan Proposal*" beginning on page 283 of the Proxy Statement/Prospectus, which is incorporated herein by reference. Such summary and the foregoing description are qualified in their entirety by reference to the text of the Equity Incentive Plan, a copy of which is attached hereto as Exhibit 10.10 and incorporated herein by reference.

### 2021 Employee Stock Purchase Plan

At the Special Meeting, the Artius shareholders considered and approved the 2021 Employee Stock Purchase Plan (the "ESPP"). The ESPP was previously approved, subject to stockholder approval, by Artius's board of directors on May 25, 2021. The ESPP became effective upon the closing.

A summary of the terms of the ESPP is set forth in the Proxy Statement/Prospectus in the section titled "*Proposal No. 8—The ESPP Proposal*" beginning on page 291 of the Proxy Statement/Prospectus, which is incorporated herein by reference. Such summary and the foregoing description are qualified in their entirety by reference to the text of the ESPP, a copy of which is attached hereto as Exhibit 10.12 and incorporated herein by reference.

# Item 5.03 Amendments to Articles of Incorporation or Bylaws; Change in Fiscal Year.

At the Special Meeting, the Artius shareholders considered and approved, among other things, Proposal No. 4—The Interim Charter Proposal, Proposal No. 5—The Charter Proposal and Proposal No. 6—The Organizational Documents Proposals (together, the "*Charter Proposals*"), which are described in greater detail in the Proxy Statement/Prospectus beginning on pages 277, 278 and 279, respectively, of the Proxy Statement/Prospectus.

The Certificate of Incorporation of the Company (the "*Certificate of Incorporation*") and the Bylaws of the Company (the "*Bylaws*"), each of which became effective on June 24, 2021 in connection with the Domestication, and the Amended and Restated Certificate of Incorporation of the Company (the "*A&R Certificate of Incorporation*"), which became effective on June 25, 2021, include the amendments proposed by the Charter Proposals.

Copies of the Certificate of Incorporation, the Bylaws and the A&R Certificate of Incorporation are attached hereto as Exhibit 3.2, Exhibit 3.4 and Exhibit 3.3, respectively, and incorporated herein by reference.

The description of the A&R Certificate of Incorporation and the general effect of the Certificate of Incorporation, the Bylaws and the A&R Certificate of Incorporation upon the rights of holders of the Company's capital stock are included in the Proxy Statement/Prospectus in the section titled "Description of Securities" beginning on page 229 of the Proxy Statement/Prospectus and incorporated herein by reference.

#### Item 5.05 Amendments to the Registrant's Code of Ethics, or Waiver of a Provision of the Code of Ethics.

In connection with the closing of the Business Combination, the Company's board of directors approved and adopted a new Code of Business Conduct and Ethics applicable to all employees, officers and directors of the Company. A copy of the Code of Business Conduct and Ethics can be found in the Investors section of the Company's website at <a href="https://www.originmaterials.com">www.originmaterials.com</a>.

# Item 5.06 Change in Shell Company Status.

As a result of the Business Combination, the Company ceased being a shell company. Reference is made to the disclosure in the Proxy Statement/Prospectus in the section titled "*The Merger Agreement and Related Agreements*" beginning on page 131 of the Proxy Statement/Prospectus and incorporated herein by reference. Further reference is made to the information contained in Item 2.01 of this Current Report on Form 8-K.

#### Item 7.01 Regulation FD Disclosure.

On June 25, 2021, the Company issued a press release announcing the Closing. A copy of the press release is filed hereto as Exhibit 99.1 and incorporated herein by reference.

The information in this Item 7.01, including Exhibit 99.1, is furnished and shall not be deemed "filed" for purposes of Section 18 of the Exchange Act, or otherwise subject to liabilities under that section, and shall not be deemed to be incorporated by reference into the filings of the registrant under the Securities Act of 1933, as amended, or the Exchange Act, regardless of any general incorporation language in such filings. This Current Report on Form 8-K will not be deemed an admission as to the materiality of any information contained in this Item 7.01, including Exhibit 99.1.

# Item 8.01 Other Events.

In April 2020, Legacy Origin received an unsecured loan in the amount of \$905,838 under the Paycheck Protection Program (the "*PPP Loan*"). The PPP Loan has a two-year term and bears interest at a rate of 1.00% per annum. In connection with consummation of the Business Combination, on June 24, 2021, Legacy Origin repaid all amounts outstanding under the PPP Loan, which totaled \$916,607.40 in principal and interest.

#### Item 9.01 Financial Statement and Exhibits.

#### (a) Financial Statements of Business Acquired.

The audited consolidated financial statements of Artius as of and for the period from January 24, 2020 (Artius' inception) to December 31, 2020 and the related notes are included in the Proxy Statement/Prospectus beginning on page F-2 of the Proxy Statement/Prospectus and incorporated herein by reference.

The unaudited consolidated financial statements of Artius as of and for the period from January 24, 2020 to March 31, 2020 and the three months ended March 31, 2021 and the related notes are included in the Proxy Statement/Prospectus beginning on page F-26 of the Proxy Statement/Prospectus and incorporated herein by reference.

The audited consolidated financial statements of Legacy Origin as of and for the years ended December 31, 2019 and 2020 and the related notes are included in the Proxy Statement/Prospectus beginning on page F-49 of the Proxy Statement/Prospectus and incorporated herein by reference.

The unaudited consolidated financial statements of Legacy Origin as of and for the three months ended March 31, 2020 and 2021 and the related notes are included in the Proxy Statement/Prospectus beginning on page F-75 of the Proxy Statement/Prospectus and are incorporated herein by reference.

# (b) Pro Forma Financial Information.

The unaudited pro forma condensed combined financial information of the Company as of and for the year ended December 31, 2020 and as of and for the three months ended March 31, 2021 is attached hereto as Exhibit 99.2 and incorporated herein by reference.

#### (d) Exhibits.

		Incorporated by Reference			
Exhibit Number	Description	Schedule/ Form	File No.	Exhibit	Filing Date
2.1+	Agreement and Plan of Merger, dated February 16, 2021.	S-4/A	333-254012	2.1	May 25, 2021
2.2	Letter Agreement, dated March 5, 2021.	S-4/A	333-254012	2.2	May 25, 2021
3.1	Artius Amended and Restated Memorandum and Articles of Association.	S-4/A	333-254012	3.1	May 25, 2021
3.2	Interim Certificate of Incorporation of Artius.	8-K	001-39378	3.1	June 29, 2021
3.3*	Amended and Restated Certificate of Incorporation of the Company				
3.4	Bylaws of the Company.	8-K	001-39378	3.2	June 29, 2021
4.1	Specimen Common Stock Certificate of the Company.	S-4/A	333-254012	4.4	May 25, 2021
4.2	Specimen Warrant Certificate of the Company.	S-1/A	333-239421	4.3	July 2, 2020
4.3	Warrant Agreement between the Company and Continental Stock Transfer & Trust Company, dated July 13, 2020.	8-K	001-39378	4.1	July 16, 2020
4.4	Certificate of Corporate Domestication of Artius.	S-4/A	333-254012	4.6	May 25, 2021

			Incorporated by Reference		
Exhibit <u>Number</u>	Description	Schedule/ Form	File No.	Exhibit	Filing Date
10.1	Form of Subscription Agreement.	8-K	001-39378	10.1	February 17, 2021
10.2	Form of Backstop Agreement.	8-K	001-39378	10.1	June 15, 2021
10.3	Form of Additional Subscription Agreement.	8-K	001-39378	10.1	June 29, 2021
10.4	Form of Lock-Up Agreement.	S-4/A	333-254012	10.26	February 17, 2021
10.5*	Investor Rights Agreement, by and between the Company and certain stockholders,				
	dated June 25, 2021.				
10.6*#	Form of Indemnification Agreement.				
10.7*#	Non-Employee Director Compensation Policy.				
10.8#	Micromidas, Inc. 2010 Stock Incentive Plan, as amended.	S-4/A	333-254012	10.1	May 25, 2021
10.9#	Forms of Incentive Stock Option Award Notice, Incentive Stock Option Award	S-4/A	333-254012	10.2	May 25, 2021
	Agreement, Exercise Notice and Investment Representation Statement under the 2010 Stock Incentive Plan.				
10.10#	Micromidas, Inc. 2020 Equity Incentive Plan.	S-4/A	333-254012	10.3	May 25, 2021
10.11#	Forms of Stock Option Grant Notice, Option Agreement and Exercise Notice under the 2020 Equity Incentive Plan.	S-4/A	333-254012	10.4	May 25, 2021
10.12*#	Origin Materials 2021 Equity Incentive Plan.				
10.13*#	Form of Stock Option Grant Notice, Stock Option Agreement, Restricted Stock				
	<u>Unit Grant Notice and Restricted Stock Unit Award Agreement under the 2021</u> <u>Equity Incentive Plan.</u>				
10.14*#	Origin Materials 2021 Employee Stock Purchase Plan.				
10.15#	Offer Letter, dated October 28, 2020, by and between Micromidas, Inc. and Rich	S-4/A	333-254012	10.7	May 25, 2021
	Riley.				
10.16#	Offer Letter, dated January 9, 2018, by and between Micromidas, Inc. and Joshua Lee.	S-4/A	333-254012	10.8	May 25, 2021
10.17#	Offer Letter, dated August 11, 2020, by and between Micromidas, Inc. and Nate	S-4/A	333-254012	10.9	May 25, 2021
	Whaley.				
10.18	<u>Standard Industrial/Commercial Multi-Tenant Lease for 930 Riverside Parkway,</u> <u>Suites 10-30, West Sacramento, CA 95605, by and between Harsch Investment</u>	S-4/A	333-254012	10.10	May 25, 2021
	Properties, LLC and Micromidas, Inc., dated May 22, 2020.				

			Incorporated by Reference		
Exhibit Number	Description	Schedule/ Form	File No.	Exhibit	Filing Date
10.19	Standard Industrial/Commercial Multi-Tenant Lease for 970 Riverside Parkway, Suite 40, West Sacramento, CA 95605, by and between Harsch Investment Properties, LLC and Micromidas, Inc., dated February 28, 2013.	S-4/A	333-254012	10.11	May 25, 2021
10.20	Second Amendment to Lease for 970 Riverside Parkway, Suite 40, West Sacramento, CA 95605, by and between Harsch Investment Properties, LLC and Micromidas, Inc., dated May 11, 2015.	S-4/A	333-254012	10.12	May 25, 2021
10.21	Third Amendment to Lease for 970 Riverside Parkway, Suite 40, West Sacramento, CA 95605, by and between Harsch Investment Properties, LLC and Micromidas, Inc., dated May 22, 2020.	S-4/A	333-254012	10.13	May 25, 2021
10.22	Form of Sponsor Letter Agreement.	S-4/A	333-254012	10.16	May 25, 2021
10.23	Form of Company Stockholder Support Agreement.	S-4/A	333-254012	10.17	May 25, 2021
10.24	<u>Promissory Note issued to Artius Acquisition Partners LLC, dated February 4, 2020.</u>	S-1	333-239421	10.1	June 25, 2020
10.25	<u>Letter Agreement among the Registrant and its directors, director nominees and officers and the Company.</u>	8-K	001-39378	10.5	July 16, 2020
10.26	Form of Investment Management Trust Agreement between Continental Stock Transfer & Trust Company and the Registrant.	8-K	001-39378	10.1	July 16, 2020
10.27	Registration Rights Agreement among the Company and certain security holders.	8-K	001-39378	10.2	July 16, 2020
10.28	Securities Subscription Agreement between the Company and Artius Acquisition Partners LLC, dated February 4, 2020.	S-1	333-239421	10.5	June 25, 2020
10.29	Private Placement Warrants Purchase Agreement between the Company and Artius Acquisition Partners LLC.	8-K	001-39378	10.3	July 16, 2020
10.30	Administrative Services Agreement between the Company and Artius Management LLC.	8-K	001-39378	10.4	July 16, 2020
10.31†	Note Purchase Agreement, by and among Micromidas, Inc. and certain persons and entities named on the Schedule of Purchasers attached therein, dated November 8, 2019.	S-4/A	333-254012	10.28	May 25, 2021
10.32	First Amendment to Note Purchase Agreement, by and between Micromidas, Inc. and certain noteholders, dated February 3, 2020.	S-4/A	333-254012	10.29	May 25, 2021

		Incorporated by Reference			
Exhibit <u>Number</u>	Description	Schedule/ Form	File No.	Exhibit	Filing Date
10.33	Form of Senior Secured Convertible Promissory Note, by and between Micromidas, Inc. and certain noteholders thereof.	S-4/A	333-254012	10.30	May 25, 2021
10.34	First Amendment to Senior Secured Convertible Promissory Note, by and between Micromidas, Inc. and certain noteholders, dated May 21, 2020.	S-4/A	333-254012	10.31	May 25, 2021
10.35	Second Amendment to Senior Secured Convertible Promissory Note, by and between Micromidas, Inc. and certain noteholders, dated January 21, 2021.	S-4/A	333-254012	10.32	May 25, 2021
10.36	Amended and Restated Secured Promissory Note, by and among Micromidas, Inc., Origin Materials Canada Holding Limited, Origin Materials Canada Pioneer Limited and Danone Asia Pte Ltd, dated May 17, 2019.	S-4/A	333-254012	10.33	May 25, 2021
10.37	First Amendment to Amended and Restated Secured Promissory Note, by and among Micromidas, Inc., Origin Materials Canada Holding Limited, Origin Materials Canada Pioneer Limited and Danone Asia Pte Ltd, dated November 8, 2019.	S-4/A	333-254012	10.34	May 25, 2021
10.38	Second Amendment to Amended and Restated Secured Promissory Note, by and among Micromidas, Inc., Origin Materials Canada Holding Limited, Origin Materials Canada Pioneer Limited and Danone Asia Pte Ltd, dated May 21, 2020.	S-4/A	333-254012	10.35	May 25, 2021
10.39	Third Amendment to Amended and Restated Secured Promissory Note, by and among Micromidas, Inc., Origin Materials Canada Holding Limited, Origin Materials Canada Pioneer Limited and Danone Asia Pte Ltd, dated January 22, 2021.	S-4/A	333-254012	10.36	May 25, 2021
10.40	Amended and Restated Secured Promissory Note, by and among Micromidas, Inc., Origin Materials Canada Holding Limited, Origin Materials Canada Pioneer Limited and Nestle Waters Management & Technology, dated May 23, 2019.	S-4/A	333-254012	10.37	May 25, 2021
10.41	First Amended and Restated Secured Promissory Note, by and among Micromidas, Inc., Origin Materials Canada Holding Limited, Origin Materials Canada Pioneer Limited and Nestle Waters Management & Technology, dated November 8, 2019.	S-4/A	333-254012	10.38	May 25, 2021

		Incorporated by Reference		ference	
Exhibit Number	Description	Schedule/ Form	File No.	Exhibit	Filing Date
10.42	Second Amended and Restated Secured Promissory Note, by and among Micromidas, Inc., Origin Materials Canada Holding Limited, Origin Materials Canada Pioneer Limited and Nestle Waters Management & Technology, dated May 21, 2020.	S-4/A	333-254012	10.39	May 25, 2021
10.43	Third Amended and Restated Secured Promissory Note, by and among Micromidas, Inc., Origin Materials Canada Holding Limited, Origin Materials Canada Pioneer Limited and Nestle Waters Management & Technology, dated January 27, 2021.	S-4/A	333-254012	10.40	May 25, 2021
10.44	Form of Convertible Promissory Note Series 2021A, by and between Micromidas, Inc. and certain noteholders thereof.	S-4/A	333-254012	10.41	May 25, 2021
10.45†^	Offtake Supply Agreement, by and between Micromidas, Inc. and Pepsi-Cola Advertising and Marketing, Inc., dated August 3, 2018.	S-4/A	333-254012	10.42	May 25, 2021
10.46^	Amendment No. 1 to Offtake Supply Agreement, by and between Micromidas, Inc. and Pepsi-Cola Advertising and Marketing, Inc., dated October 24, 2019.	S-4/A	333-254012	10.43	May 25, 2021
10.47†^	Amended and Restated Offtake Supply Agreement, by and between Micromidas, Inc. and Danone Asia Pte Ltd, dated May 17, 2019.	S-4/A	333-254012	10.44	May 25, 2021
10.48†^	Amended and Restated Offtake Supply Agreement, by and between Micromidas, Inc. and Nestle Waters Management & Technology, dated May 23, 2019.	S-4/A	333-254012	10.45	May 25, 2021
10.49†^	Offtake Supply Agreement, by and between Micromidas, Inc. and Packaging Equity Holdings, LLC, dated December 13, 2020.	S-4/A	333-254012	10.46	May 25, 2021
16.1*	Letter from Marcum LLP.				
21.1*	List of Subsidiaries.				
99.1**	Press Release dated June 25, 2021.				
99.2*	<u>Unaudited pro forma condensed combined financial information as of and for the year ended December 31, 2020 and as of and for the three months ended March 31, 2021.</u>				

<sup>#</sup> Indicates management contract or compensatory plan or arrangement.

- \* Filed herewith.
- \*\* Furnished herewith.
- + Schedules and exhibits have been omitted pursuant to Item 601(b)(2) of Regulation S-K. The Company agrees to furnish supplementally a copy of any omitted schedule or exhibit to the SEC upon request.
- † Certain schedules and exhibits to this exhibit have been omitted pursuant to Item 601(a)(5) of Regulation S-K. A copy of any omitted schedule and/or exhibit will be furnished to the SEC upon request.
- ^ Certain portions of this exhibit (indicated by asterisks) have been excluded pursuant to Item 601(b)(10) of Regulation S-K because they are both not material and are the type that the Company treats as private or confidential.

# **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the Company has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

ORIGIN MATERIALS, INC.

Dated: July 1, 2021 By: /S/ NATE WHALEY

Nate Whaley

Chief Financial Officer

# AMENDED AND RESTATED CERTIFICATE OF INCORPORATION OF ORIGIN MATERIALS, INC.

Origin Materials, Inc., a corporation organized and existing under the General Corporation Law of the State of Delaware (as it now exists or may hereafter be amended and supplemented, the "*DGCL*"), does hereby certify that:

**ONE:** The original Certificate of Incorporation of Origin Materials, Inc. was filed with the Secretary of State of the State of Delaware on June 24, 2021 under the name Origin Materials, Inc. (the "*Certificate of Incorporation*") and effective at 5:15 p.m. on June 24, 2021.

TWO: This Amended and Restated Certificate of Incorporation restates, integrates and amends the provisions of the Certificate of Incorporation

THREE: This Amended and Restated Certificate of Incorporation shall become effective as of 12:01 a.m. Eastern Time on June 25, 2021.

**FOUR:** The Certificate of Incorporation is hereby amended and restated to read as follows:

#### ARTICLE I

The name of this corporation is Origin Materials, Inc. (the "Company").

#### **ARTICLE II**

The address of the registered office of the Company in the State of Delaware is Corporation Trust Center, 1209 Orange Street, in the City of Wilmington, County of New Castle, 19801. The name of its registered agent at such address is The Corporation Trust Company.

#### **ARTICLE III**

The purpose of the Company is to engage in any lawful act or activity for which a corporation may be organized under the DGCL.

#### ARTICLE IV

- **A.** The Company is authorized to issue two classes of stock to be designated, respectively, "*Common Stock*" and "*Preferred Stock*." The total number of shares which the Company is authorized to issue is 1,010,000,000 shares, consisting of (i) 1,000,000,000 shares of Common Stock, having a par value per share of \$0.0001 and (ii) 10,000,000 shares of Preferred Stock, having a par value per share of \$0.0001.
- **B.** Effective immediately upon the effectiveness of this Amended and Restated Certificate of Incorporation (the time of such effectiveness, the "Effective Time"), each share of the Company's Class A Common Stock, par value \$0.0001 per share (the "Class A Common Stock") and Class B Common Stock, par value \$0.0001 per share (the "Class B Common Stock"), that was issued and outstanding immediately prior to the Effective Time shall automatically be reclassified, redesignated and changed into one validly issued, fully paid and non-assessable share of Common Stock of the Company, par value \$0.0001 per share (the "Common Stock"), without any further action by the Company or any stockholder thereof. Each certificate that immediately prior to the Effective Time represented shares of Class A Common Stock or Class B Common Stock (each, a "Prior Certificate") shall, until surrendered to the Company in exchange for a certificate representing the same number of shares of Common Stock, automatically represent that number of shares of Common Stock into which the shares of Class A Common Stock or Class B Common Stock represented by the Prior Certificate shall have been reclassified and redesignated.

- **C.** The Preferred Stock may be issued from time to time in one or more series. The Board of Directors of the Company (the "**Board of Directors**") is hereby expressly authorized to provide for the issue of all or any of the shares of the Preferred Stock in one or more series, and to fix the number of shares and to determine or alter for each such series, such voting powers, full or limited, or no voting powers, and such designation, preferences, and relative, participating, optional, or other rights and such qualifications, limitations, or restrictions thereof, as shall be stated and expressed in the resolution or resolutions adopted by the Board of Directors providing for the issuance of such shares and as may be permitted by the DGCL. The Board of Directors is also expressly authorized to increase or decrease the number of shares of any series subsequent to the issuance of shares of that series, but not below the number of shares of such series then outstanding. In case the number of shares of any series shall be decreased in accordance with the foregoing sentence, the shares constituting such decrease shall resume the status that they had prior to the adoption of the resolution originally fixing the number of shares of such series. The number of authorized shares of Preferred Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the voting power of the stock of the Company entitled to vote thereon, without a separate vote of the holders of the Preferred Stock, or of any series thereof, unless a vote of any such holders is required pursuant to the terms of any certificate of designation filed with respect to any series of Preferred Stock.
- **D.** Each outstanding share of Common Stock shall entitle the holder thereof to one vote on each matter properly submitted to the stockholders of the Company for their vote; *provided*, *however*, that, except as otherwise required by law, holders of Common Stock shall not be entitled to vote on any amendment to this Amended and Restated Certificate of Incorporation (including any certificate of designation filed with respect to any series of Preferred Stock) that relates solely to the terms of one or more outstanding series of Preferred Stock if the holders of such affected series are entitled, either separately or together as a class with the holders of one or more other such series, to vote thereon by law or pursuant to this Amended and Restated Certificate of Incorporation (including any certificate of designation filed with respect to any series of Preferred Stock) or the DGCL.
- **E.** Subject to the rights of the holders of any outstanding series of Preferred Stock, the holders of shares of Common Stock shall be entitled to receive any dividends to the extent permitted by law when, as and if declared by the Board of Directors of the Company.
- **F.** Upon the dissolution, liquidation or winding up of the Company, subject to the rights of the holders of any outstanding series of Preferred Stock, the holders of shares of Common Stock shall be entitled to receive the assets of the Company available for distribution to its stockholders ratably in proportion to the number of shares held by them. For the avoidance of doubt, a dissolution, liquidation or winding up shall not be deemed to be occasioned by or to include, without limitation, any voluntary consolidation, reorganization, conversion or merger of the Company with or into any other corporation or entity or other corporation or entities or a sale, lease, transfer, exchange or conveyance of all or a part of the Company's assets.

#### ARTICLE V

For the management of the business and for the conduct of the affairs of the Company, and in further definition, limitation and regulation of the powers of the Company, of its directors and of its stockholders or any class thereof, as the case may be, it is further provided that:

**A. MANAGEMENT OF BUSINESS.** The management of the business and the conduct of the affairs of the Company shall be vested in its Board of Directors. The number of directors which shall constitute the Board of Directors shall be fixed exclusively by resolutions adopted by a majority of the authorized number of directors constituting the Board of Directors.

#### B. ELECTION OF BOARD OF DIRECTORS.

- 1. Subject to the rights of the holders of any series of Preferred Stock to elect additional directors under specified circumstances, from the effective date of this Amended and Restated Certificate of Incorporation (the "Effective Date") until the completion of the fourth annual meeting of stockholders to occur after the Effective Date, the directors shall be divided into three classes designated as Class I, Class II and Class III, respectively. Each class shall consist, as nearly as may be possible, of one-third of the total number of directors constituting the entire Board of Directors. If the number of directors is changed, any increase or decrease shall be apportioned among the classes so as to maintain the number of directors in each class as nearly equal as possible, and any additional director of any class elected to fill a vacancy resulting from an increase in such class or from the removal from office, death, disability, resignation or disqualification of a director or other cause shall hold office for a term that shall coincide with the remaining term of that class, but in no case will a decrease in the number of directors have the effect of removing or shortening the term of any incumbent director. The Board of Directors is authorized to assign members of the Board of Directors already in office to such classes at the time the classification becomes effective. The term of the initial Class I directors shall terminate on the date of the first annual meeting of stockholders to occur after the Effective Date; the term of the initial Class II directors shall terminate on the date of the second annual meeting of stockholders to occur after the Effective Date; and the term of the initial Class III directors shall terminate on the date of the third annual meeting of stockholders to occur after the Effective Date, or, in each case, upon such director's earlier death, resignation or removal. Each Class I director elected at the first annual meeting of stockholders to occur after the Effective Date shall hold office until the fourth annual meeting of stockholders to occur after the Effective Date and each Class I director elected at the fourth annual meeting of stockholders to occur after the Effective Date shall hold office until the fifth annual meeting of stockholders to occur after the Effective Date and, in each case, until his or her respective successor shall have been duly elected and qualified or until his or her earlier resignation or removal. Each Class II director elected at the second annual meeting of stockholders to occur after the Effective Date and each Class III director elected at the third annual meeting of stockholders to occur after the Effective Date shall hold office until the fifth annual meeting of stockholders to occur after the Effective Date and, in each case, until his or her respective successor shall have been duly elected and qualified or until his or her earlier resignation or removal. Commencing with the fifth annual meeting of stockholders to occur after the Effective Date, each director shall be elected annually and shall hold office until the next annual meeting of stockholders and until his or her respective successor shall have been duly elected and qualified or until his or her earlier resignation or removal. Pursuant to such procedures, effective as of the conclusion of the fourth annual meeting of stockholders to occur after the Effective Date, the Board of Directors will no longer be classified under Section 141(d) of the DGCL and directors shall no longer be divided into three classes.
- **2.** At any time that applicable law prohibits a classified board as described in Section B.1. of this Article V, all directors shall be elected at each annual meeting of stockholders to hold office until the next annual meeting. The directors of the Company need not be elected by written ballot unless the Bylaws so provide.
- 3. No stockholder entitled to vote at an election for directors may cumulate votes to which such stockholder is entitled unless required by applicable law at the time of such election. During such time or times that applicable law requires cumulative voting, every stockholder entitled to vote at an election for directors may cumulate such stockholder's votes and give one candidate a number of votes equal to the number of directors to be elected multiplied by the number of votes to which such stockholder's shares are otherwise entitled, or distribute the stockholder's votes on the same principle among as many candidates as such stockholder thinks fit. No stockholder, however, shall be entitled to so cumulate such stockholder's votes unless (i) the names of such candidate or candidates have been placed in nomination

prior to the voting and (ii) the stockholder has given notice at the meeting, prior to the voting, of such stockholder's intention to cumulate such stockholder's votes. If any stockholder has given proper notice to cumulate votes, all stockholders may cumulate their votes for any candidates who have been properly placed in nomination. Under cumulative voting, the candidates receiving the highest number of votes, up to the number of directors to be elected, are elected.

- **4.** Notwithstanding the foregoing provisions of this section, each director shall serve until his or her successor is duly elected and qualified or until his or her earlier death, resignation or removal. No decrease in the number of directors constituting the Board of Directors shall shorten the term of any incumbent director.
- 5. Notwithstanding the foregoing, whenever the holders of any one or more classes or series of Preferred Stock issued by the Company shall have the right, voting separately by class or series, to elect directors at an annual or special meeting of stockholders, the election, term of office, filling of vacancies and other features of such directorships shall be governed by the terms of this Amended and Restated Certificate of Incorporation and the resolution or resolutions adopted by the Board of Directors providing for the issuance of such class or series, and such directors so elected shall not be divided into classes pursuant to this Article V unless expressly provided by such terms.

#### C. REMOVAL OF DIRECTORS.

- **1.** Prior to the fifth annual meeting of stockholders to occur after the Effective Date, except as required by applicable law and subject to the rights of any series of Preferred Stock to elect additional directors under specified circumstances, the Board of Directors or any individual director may be removed from office at any time, but only for cause by the affirmative vote of the holders of at least sixty-six and two-thirds percent (66 2/3%) of the voting power of all then-outstanding shares of capital stock of the Company entitled to vote generally at an election of directors.
- **2.** Beginning with the fifth annual meeting of stockholders to occur after the Effective Date, subject to the rights of any series of Preferred Stock to elect additional directors under specified circumstances, the Board of Directors or any individual director may be removed from office at any time, with or without cause, by the affirmative vote of the holders of at least sixty-six and two-thirds percent (66 2/3%) of the voting power of all then-outstanding shares of capital stock of the Company entitled to vote generally at an election of directors.
- **D.** VACANCIES. Subject to any limitations imposed by applicable law and subject to the rights of the holders of any series of Preferred Stock, any vacancies on the Board of Directors resulting from death, resignation, disqualification, removal or other causes and any newly created directorships resulting from any increase in the number of directors, shall, unless the Board of Directors determines by resolution that any such vacancies or newly created directorships shall be filled by the stockholders and except as otherwise provided by applicable law, be filled only by the affirmative vote of a majority of the directors then in office, even though less than a quorum of the Board of Directors, and not by the stockholders. Any director elected in accordance with the preceding sentence shall hold office for the remainder of the full term of the director for which the vacancy was created or occurred and until such director's successor shall have been elected and qualified.

#### E. BYLAW AMENDMENTS.

**1.** The Board of Directors is expressly empowered to adopt, amend or repeal the Bylaws of the Company. Any adoption, amendment or repeal of the Bylaws of the Company by the Board of Directors shall require the approval of a majority of the authorized number of directors. The stockholders shall also have power to adopt, amend or repeal the Bylaws of the Company; *provided*, *however*, that, in addition to any vote of the holders of any class or series of stock of the Company required by law or by this

Amended and Restated Certificate of Incorporation, such action by stockholders shall require the affirmative vote of the holders of at least sixty-six and two-thirds percent (66 2/3%) of the voting power of all of the then-outstanding shares of the capital stock of the Company entitled to vote generally in the election of directors, voting together as a single class.

- 2. The directors of the Company need not be elected by written ballot unless the Bylaws so provide.
- **3.** No action shall be taken by the stockholders of the Company except at an annual or special meeting of stockholders called in accordance with the Bylaws, and no action shall be taken by the stockholders by written consent or electronic transmission.
- **4.** Advance notice of stockholder nominations for the election of directors and of business to be brought by stockholders before any meeting of the stockholders of the Company shall be given in the manner provided in the Bylaws of the Company.
- 5. In the event that a member of the Board of Directors of the Company who is not an employee of the Company, or any partner, member, director, stockholder, employee or agent of such member, other than someone who is an employee of the Company (collectively, the "Covered Persons"), acquires knowledge of any business opportunity matter, potential transaction, interest or other matter, unless such matter, transaction or interest is presented to, or acquired, created or developed by, or otherwise comes into the possession of, a Covered Person expressly and solely in connection with such individual's service as a member of the Board of Directors of the Company (a "Corporate Opportunity"), then the Company, pursuant to Section 122(17) of the DGCL and to the maximum extent permitted from time to time under Delaware law, (i) renounces any expectancy that such Covered Person offer an opportunity to participate in such Corporate Opportunity to the Company and (ii) to the fullest extent permitted by law, waives any claim that such opportunity constituted a Corporate Opportunity that should have been presented by such Covered Person to the Company or any of its affiliates. No amendment or repeal of this paragraph shall apply to or have any effect on the liability or alleged liability of any officer, director or stockholder of the Company for or with respect to any opportunities of which such officer, director or stockholder becomes aware prior to such amendment or repeal.

# ARTICLE VI

No director of the Company shall have any personal liability to the Company or its stockholders for monetary damages for any breach of fiduciary duty as a director, except to the extent such exemption from liability or limitation thereof is not permitted under the DGCL as the same exists or hereafter may be amended. Any amendment, repeal or modification of this Article VI, or the adoption of any provision of the Certificate of Incorporation of the Company inconsistent with this Article VI, shall not adversely affect any right or protection of a director of the Company with respect to any act or omission occurring prior to such amendment, repeal, modification or adoption. If the DGCL is amended after approval by the stockholders of this Article VI to authorize corporate action further eliminating or limiting the personal liability of directors, then the liability of a director of the Company shall be eliminated or limited to the fullest extent permitted by the DGCL as so amended.

# **ARTICLE VII**

**A.** The Company shall indemnify its directors and executive officers to the fullest extent authorized or permitted by applicable law, as now or hereafter in effect, and such right to indemnification shall continue as to a person who has ceased to be a director or executive officer of the Company and shall inure to the benefit of his or her heirs, executors and personal and legal representatives; *provided*, *however*, that the Company shall not be required to indemnify any director or executive officer in connection with any proceeding (or part thereof) initiated by such person unless (i) such indemnification is expressly

required to be made by law, (ii) the proceeding was authorized by the Board of Directors, (iii) such indemnification is provided by the Company, in its sole discretion, pursuant to the powers vested in the corporation under the DGCL or any other applicable law or (iv) such indemnification is required to be made under the Bylaws.. The right to indemnification conferred by this Article VII shall include the right to be paid by the Company the expenses incurred in defending or otherwise participating in any proceeding in advance of its final disposition upon receipt by the Company of an undertaking by or on behalf of the director or executive officer receiving advancement to repay the amount advanced if it shall ultimately be determined that such person is not entitled to be indemnified by the Company under this Article VII. The Company may, to the extent authorized from time to time by the Board of Directors, provide rights to indemnification and to the advancement of expenses to other officers, employees and agents of the Company similar to those conferred in this Article VII to directors and executive officers of the Company. The rights to indemnification and to the advancement of expenses conferred in this Article VII shall not be exclusive of any other right which any person may have or hereafter acquire under this Certificate of Incorporation, the Bylaws, any statute, agreement, vote of stockholders or disinterested directors or otherwise. Any repeal or modification of this Article VII by the stockholders of the Company shall, unless otherwise required by law, be prospective only (except to the extent such amendment or change in law permits the Company to provide broader indemnification rights on a retroactive basis than permitted prior thereto), and shall not adversely affect any rights to indemnification and to the advancement of expenses of a director, executive officer, other officer, employee or agent of the Company (collectively, the "Indemnified Persons") existing at the time of such repeal or modification in respect of any proceeding (regardless of when such proceeding is first threatened, commenced or completed) arising out of, or related to, any act or omission occurring prior to such repeal or modification of such inconsistent provision.

**B.** The Company hereby acknowledges that certain Indemnified Persons may have rights to indemnification and advancement of expenses (directly or through insurance obtained by any such entity) provided by one or more third parties (collectively, the "Other Indemnitors"), and which may include third parties for whom such Indemnified Person serves as a manager, member, officer, employee or agent. The Company hereby agrees and acknowledges that notwithstanding any such rights that a Indemnified Person may have with respect to any Other Indemnitor(s), (i) the Company is the indemnitor of first resort with respect to all Indemnified Persons in respect of all obligations to indemnify and provide advancement of expenses to Indemnified Persons, (ii) the Company shall be required to indemnify and advance the full amount of expenses incurred by the Indemnified Persons, to the fullest extent required by law, the terms of this Certificate of Incorporation, the Bylaws, any agreement to which the Company is a party, any vote of the stockholders or the Board of Directors, or otherwise, without regard to any rights the Indemnified Persons may have against the Other Indemnitors and (iii) to the fullest extent permitted by law, the Company irrevocably waives, relinquishes and releases the Other Indemnitors from any and all claims for contribution, subrogation or any other recovery of any kind in respect thereof. The Company further agrees that no advancement or payment by the Other Indemnitors with respect to any claim for which the Indemnified Persons have sought indemnification from the Company shall affect the foregoing and the Other Indemnitors shall have a right of contribution and/or be subrogated to the extent of any such advancement or payment to all of the rights of recovery of the Indemnified Persons against the Company. These rights shall be a contract right, and the Other Indemnitors are express third party beneficiaries of the terms of this paragraph. Notwithstanding anything to the contrary herein, the obligations

#### **ARTICLE VIII**

**A.** Unless the Company consents in writing to the selection of an alternative forum, the Court of Chancery of the State of Delaware (or, if and only if the Court of Chancery of the State of Delaware lacks subject matter jurisdiction, any state court located within the State of Delaware or, if and only if all such state courts lack subject matter jurisdiction, the federal district court for the District of Delaware) and any appellate court therefrom shall be the sole and exclusive forum for the following claims or causes of action under the Delaware statutory or common law: (i) any derivative claim or cause of action brought on

behalf of the Company; (ii) any claim or cause of action for breach of a fiduciary duty owed by any current or former director, officer or other employee of the Company or the Company's stockholders; (iii) any claim or cause of action against the Company or any current or former director, officer or other employee of the Company, arising out of or pursuant to any provision of the DGCL, this Amended and Restated Certificate of Incorporation or the Bylaws of the Company (as each may be amended from time to time); (iv) any claim or cause of action seeking to interpret, apply, enforce or determine the validity of this Amended and Restated Certificate of Incorporation or the Bylaws of the Company (as each may be amended from time to time, including any right, obligation, or remedy thereunder); (v) any claim or cause of action as to which the DGCL confers jurisdiction on the Court of Chancery of the State of Delaware; and (vi) any claim or cause of action against the Company or any current or former director, officer or other employee of the Company, governed by the internal-affairs doctrine or otherwise related to the Company's internal affairs, in all cases to the fullest extent permitted by law and subject to the court having personal jurisdiction over the indispensable parties named as defendants. This Section A of Article VIII shall not apply to claims or causes of action brought to enforce a duty or liability created by the Securities Act or the Securities Exchange Act of 1934, as amended, or any other claim for which the federal courts have exclusive jurisdiction.

- **B.** Unless the Company consents in writing to the selection of an alternative forum, to the fullest extent permitted by law, the federal district courts of the United States of America shall be the exclusive forum for the resolution of any complaint asserting a cause of action arising under the Securities Act, including all causes of action asserted against any defendant named in such complaint. For the avoidance of doubt, this provision is intended to benefit and may be enforced by the Company, the Company's officers and directors, the underwriters to any offering giving rise to such complaint, and any other professional entity whose profession gives authority to a statement made by that person or entity and who has prepared or certified any part of the documents underlying the offering.
- C. Any person or entity purchasing or otherwise acquiring any interest in shares of capital stock of the Company shall be deemed to have notice of and to have consented to the provisions of this Article VIII.

# ARTICLE IX

- **A.** The Company reserves the right to amend, alter, change or repeal any provision contained in this Amended and Restated Certificate of Incorporation, in the manner now or hereafter prescribed by statute, except as provided in paragraph B. of this Article IX, and all rights conferred upon the stockholders herein are granted subject to this reservation.
- **B.** Notwithstanding any other provisions of this Amended and Restated Certificate of Incorporation or any provision of applicable law which might otherwise permit a lesser vote or no vote, but in addition to any affirmative vote of the holders of any particular class or series of the capital stock of the Company required by law or by this Amended and Restated Certificate of Incorporation or any certificate of designation filed with respect to a series of Preferred Stock, the affirmative vote of the holders of at least sixty-six and two-thirds percent (66 2/3%) of the voting power of all of the then outstanding shares of capital stock of the Company entitled to vote generally in the election of directors, voting together as a single class, shall be required to alter, amend or repeal Articles V, VI, VII, VIII and IX.

\* \* \* \*

FIVE: This Amended and Restated Certificate of Incorporation has been duly approved by the Board of Directors of the Company.

**SIX:** This Amended and Restated Certificate of Incorporation was approved by the holders of the requisite number of shares of the Company in accordance with Section 228 of the DGCL. This Amended and Restated Certificate of Incorporation has been duly adopted in accordance with the provisions of Sections 242 and 245 of the DGCL by the stockholders of the Company.

**IN WITNESS WHEREOF**, the Company has caused this Amended and Restated Certificate of Incorporation to be signed by its Chief Executive Officer this 24th day of June, 2021.

By: /s/ Boon Sim
Boon Sim
Chief Executive Officer

[Signature page to Amended and Restated Certificate of Incorporation of Origin Materials, Inc.]

#### **INVESTOR RIGHTS AGREEMENT**

THIS INVESTOR RIGHTS AGREEMENT (this "Agreement"), dated as of June 25, 2021, is made and entered into by and among Origin Materials, Inc., a Delaware corporation (the "Company") (formerly known as Artius Acquisition Inc., a Cayman Islands exempted company prior to its domestication as a Delaware corporation), Artius Acquisition Partners LLC, a Delaware limited liability company (the "Sponsor"), and the undersigned parties listed under New Holders on the signature page hereto (each such party, together with any person or entity deemed a "New Holder" who hereafter becomes a party to this Agreement pursuant to Section 5.2 of this Agreement, a "New Holder" and collectively the "New Holders"). Capitalized terms used but not otherwise defined in this Agreement shall have the meaning ascribed to such term in the Merger Agreement (as defined below).

#### **RECITALS**

**WHEREAS**, the Company and the Sponsor are party to that certain Registration Rights Agreement dated July 13, 2020 (the "*Existing Registration Rights Agreement*"), pursuant to which the Company granted the Sponsor certain registration rights with respect to certain securities of the Company;

**WHEREAS**, the Company has entered into that certain Agreement and Plan of Merger (the "*Merger Agreement*"), dated as of February 16, 2021, by and among the Company, Micromidas, Inc., a Delaware corporation, and Zero Carbon Merger Sub Inc., a Delaware corporation and a wholly owned subsidiary of Artius;

**WHEREAS**, on the date hereof, pursuant to the transactions contemplated by the Merger Agreement, the New Holders received shares of the Company's Common Stock upon the closing of such transactions (the "*Closing*");

**WHEREAS**, the New Holders may receive additional shares of Common Stock as Earnout Shares pursuant to the earn out provisions in the Merger Agreement;

WHEREAS, the Sponsor holds 18,112,500 shares of Common Stock as of the date hereof (the "Founder Shares");

**WHEREAS**, on July 13, 2020, the Company and the Sponsor entered into that certain Private Placement Warrants Purchase Agreement, pursuant to which the Sponsor purchased 11,326,667 warrants in connection with the Company's initial public offering (the "*Private Placement Warrants*"), in a private placement transaction occurring simultaneously with the closing of the Company's initial public offering;

**WHEREAS**, the Company and the Sponsor have entered into that certain Amended & Restated Sponsor Letter Agreement (the "*Sponsor Agreement*"), dated as of February 16, 2021, wherein the Company and the Sponsor agreed, in connection with the Closing, to subject the Founder Shares held by the Sponsor to certain vesting requirements and to surrender to the Company certain Founder Shares under certain circumstances, in accordance with the terms of the Sponsor Agreement;

**WHEREAS**, pursuant to Section 5.5 of the Existing Registration Rights Agreement, the provisions, covenants and conditions set forth therein may be amended or modified upon the written consent of the Company and the holders of a majority-in-interest of the "Registrable Securities" (as such term was defined in the Existing Registration Rights Agreement) at the time in question;

**WHEREAS**, the Company and the Sponsor desire to terminate the Existing Registration Rights Agreement and all other registration rights that might exist with respect to the equity securities of the Company and to enter into this Agreement in order to provide the Sponsor and the New Holders certain registration rights with respect to certain securities of the Company, as set forth in this Agreement.

**NOW**, **THEREFORE**, in consideration of the representations, covenants and agreements contained herein, and certain other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

# ARTICLE I DEFINITIONS

- 1.1. Definitions. The terms defined in this Article I shall, for all purposes of this Agreement, have the respective meanings set forth below:
- "Adverse Disclosure" shall mean any public disclosure of material non-public information, which disclosure, in the good faith judgment of any Chief Executive Officer or Chief Financial Officer of the Company, after consultation with counsel to the Company, (i) would be required to be made in any Registration Statement or Prospectus in order for the applicable Registration Statement or Prospectus not to contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements contained therein (in the case of any prospectus and any preliminary prospectus, in the light of the circumstances under which they were made) not misleading, (ii) would not be required to be made at such time if the Registration Statement were not being filed, and (iii) the Company has a bona fide business purpose for not making such information public.
  - "Agreement" shall have the meaning given in the Preamble.
- "*Block Trade*" means an offering and/or sale of Registrable Securities by any Holder on a block trade or underwritten basis (whether firm commitment or otherwise) without substantial marketing efforts prior to pricing, including, without limitation, a same day trade, overnight trade or similar transaction.
  - "Board" shall mean the Board of Directors of the Company.
  - "Bylaws" means the amended and restated bylaws of the Company, as the same may be amended or amended and restated from time to time.

- "Charter" means the amended and restated certificate of incorporation of the Company, as the same may be amended or amended and restated from time to time.
- "Common Stock" shall mean (i) following the Domestication, the Class A common stock of Artius, par value one ten-thousandth of one dollar (\$0.0001) per share, to be authorized pursuant to the Interim Artius Certificate of Incorporation and (ii) following the Effective Time, the Common Stock of Artius, par value one ten-thousandth of one dollar (\$0.0001) per share, to be authorized pursuant to the Artius Certificate of Incorporation.
  - "Commission" shall mean the Securities and Exchange Commission.
- "Commission Guidance" means (i) any publicly-available written or oral guidance of the Commission staff, or any comments, requirements or requests of the Commission staff and (ii) the Securities Act.
  - "Company" shall have the meaning given in the Preamble.
  - "Company Shelf Takedown Notice" shall have the meaning given in subsection 2.1.3.
  - "Demand Registration" shall have the meaning given in subsection 2.2.1.
  - "Demanding Holders" shall have the meaning given in subsection 2.2.1.
  - "Effectiveness Deadline" shall have the meaning given in subsection 2.1.1.
  - "Exchange Act" shall mean the Securities Exchange Act of 1934, as it may be amended from time to time.
  - "Existing Registration Rights Agreement" shall have the meaning given in the Recitals hereto.
  - "Form S-1 Shelf" shall have the meaning given in subsection 2.1.1.
  - "Form S-3 Shelf" shall have the meaning given in subsection 2.1.1.
  - "Founder Shares" shall have the meaning given in the Recitals hereto.
- "Holders" shall mean the Sponsor and the New Holders and any person or entity who hereafter becomes a party to this Agreement pursuant to Section 5.2.
- "Lock-Up Period" shall mean, with respect to the Registrable Securities held by the Holders or their Permitted Transferees, from the date hereof and through the earliest to occur of (A) 365 days after the date hereof; (B) the first day after the date on which the closing price of the Common Stock equals or exceeds \$12.00 per share (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like) for any 20 trading days within any 30-trading day period commencing at least 150 days after the date hereof; and (C) the date on which the Company completes a liquidation, merger, capital stock exchange, reorganization or other similar transaction that results in all of the Company's public stockholders having the right to exchange their Common Stock for cash, securities or other property.

- "Lock-Up Agreement" shall mean that certain Lock-Up Agreement, dated as of February 16, 2021, by and among the Company, the Sponsor and certain stockholders signatory thereto.
  - "Maximum Number of Securities" shall have the meaning given in subsection 2.2.4.
  - "Merger Agreement" shall have the meaning given in the Recitals hereto.
- "Misstatement" shall mean an untrue statement of a material fact or an omission to state a material fact required to be stated in a Registration Statement or Prospectus, or necessary to make the statements in a Registration Statement or Prospectus (in the case of the Prospectus, in the light of the circumstances under which they were made) not misleading.
  - "New Holders" shall have the meaning given in the Preamble.
  - "Organizational Documents" means the Charter and Bylaws.
- "Permitted Transferees" shall mean (i) a person or entity to whom a Holder of Registrable Securities is permitted to transfer such Registrable Securities prior to the expiration of the Lock-Up Period under the Lock-Up Agreement and (ii) after expiration of the Lock-Up Period, a person or entity to whom a Holder of Registrable Securities is permitted to transfer such Registrable Securities under this Agreement, the Sponsor Agreement and any other applicable agreement between such Holder and the Company, and to any transferee thereafter.
  - "Piggyback Registration" shall have the meaning given in subsection 2.3.1.
  - "Private Placement Warrants" shall have the meaning given in the Recitals hereto.
  - "Pro Rata" shall have the meaning given in subsection 2.2.4.
- "*Prospectus*" shall mean the prospectus included in any Registration Statement, as supplemented by any and all prospectus supplements and as amended by any and all post-effective amendments and including all material incorporated by reference in such prospectus.
- "Registrable Security" shall mean (a) the Founder Shares, (b) the Private Placement Warrants (including any Common Stock issued or issuable upon the exercise of any such Private Placement Warrants), (c) any other issued and outstanding share of Common Stock or any other equity security (including the shares of Common Stock issued or issuable upon the exercise of any other equity security, units comprising Common Stock and warrants, and warrants) of the Company held by the Sponsor as of the date of this Agreement, (d) any equity securities (including the shares of Common Stock issued or issuable upon the exercise of any such equity security) of the Company issuable upon conversion of any working capital loans in an amount up to \$1,500,000 made to the Company by a Holder, (e) any outstanding shares of Common Stock or any other equity security of the Company held by a New Holder as of the date of this Agreement (including shares transferred to a Permitted Transferee and the shares of Common Stock issued or issuable upon the exercise of any such other equity security, units comprising Common Stock and warrants,

and warrants), (f) any shares of Common Stock issued or issuable as Earnout Shares to a New Holder and (g) any other equity security of the Company issued or issuable with respect to any such share of Common Stock described in the foregoing clauses (a) through (f) by way of a stock dividend or stock split or in connection with a combination of shares, recapitalization, merger, consolidation or reorganization; provided, however, that, as to any particular Registrable Security, such securities shall cease to be Registrable Securities when: (A) a Registration Statement with respect to the sale of such securities shall have become effective under the Securities Act and such securities shall have been sold, transferred, disposed of or exchanged in accordance with such Registration Statement; (B) such securities shall have been otherwise transferred, new certificates or book entries for such securities not bearing a legend restricting further transfer shall have been delivered by the Company and subsequent public distribution of such securities shall not require registration under the Securities Act; (C) such securities shall have ceased to be outstanding; (D) such securities may be sold without registration and without limitations, including restrictions on volume, manner of sale or other limitations or restrictions pursuant to Rule 144 promulgated under the Securities Act (or any successor rule promulgated thereafter by the Commission) ("Rule 144"); or (E) such securities have been sold to, or through, a broker, dealer or underwriter in a public distribution or other public securities transaction.

"*Registration*" shall mean a registration effected by preparing and filing a registration statement or similar document in compliance with the requirements of the Securities Act, and the applicable rules and regulations promulgated thereunder, and such registration statement becoming effective.

"Registration Expenses" shall mean the documented, out-of-pocket expenses of a Registration, including, without limitation, the following:

- (A) all registration and filing fees (including fees with respect to filings required to be made with the Financial Industry Regulatory Authority, Inc.) and any securities exchange on which the Common Stock is then listed;
- (B) fees and expenses of compliance with securities or blue sky Laws (including reasonable fees and disbursements of outside counsel for the Underwriters in connection with blue sky qualifications of Registrable Securities);
  - (C) printing, messenger, telephone and delivery expenses;
  - (D) reasonable fees and disbursements of counsel for the Company;
- (E) reasonable fees and disbursements of all independent registered public accountants of the Company incurred specifically in connection with such Registration; and
- (F) reasonable fees and expenses of one (1) legal counsel selected by the majority-in-interest of the Demanding Holders initiating a Demand Registration to be registered for offer and sale in the applicable Registration.

"Registration Statement" shall mean any registration statement that covers the Registrable Securities pursuant to the provisions of this Agreement, including the Prospectus included in such registration statement, amendments (including post-effective amendments) and supplements to such registration statement, and all exhibits to and all material incorporated by reference in such registration statement.

- "Removed Shares" shall have the meaning given in Section 2.6.
- "Requesting Holder" shall have the meaning given in subsection 2.2.1.
- "Rule 144" shall have the meaning given in the definition of "Registrable Security."
- "Rule 415" shall have the meaning given in subsection 2.1.1.
- "SEC" shall mean the United States Securities and Exchange Commission.
- "Securities Act" shall mean the Securities Act of 1933, as amended from time to time.
- "Shelf Takedown Notice" shall have the meaning given in subsection 2.1.3.
- "Shelf Underwritten Offering" shall have the meaning given in subsection 2.1.3.
- "Sponsor" shall have the meaning given in the Preamble hereto.
- "Sponsor Agreement" shall have the meaning given in the Recitals hereto.
- "*Transfer*" shall mean the (a) sale or assignment of, offer to sell, contract or agreement to sell, hypothecate, pledge, grant of any option to purchase or otherwise dispose of or agreement to dispose of, directly or indirectly, or establishment or increase of a put equivalent position or liquidation with respect to or decrease of a call equivalent position within the meaning of Section 16 of the Exchange Act and the rules and regulations of the Commission promulgated thereunder with respect to, any security, (b) entry into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any security, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (c) public announcement of any intention to effect any transaction specified in clause (a) or (b).
- "*Underwriter*" shall mean a securities dealer who purchases any Registrable Securities as principal in an Underwritten Offering and not as part of such dealer's market-making activities.
- "*Underwritten Registration*" or "*Underwritten Offering*" shall mean a Registration in which securities of the Company are sold to an Underwriter in a firm commitment underwriting for distribution to the public.

# ARTICLE II REGISTRATIONS

# 2.1. Shelf Registration.

2.1.1. <u>Initial Registration</u>. The Company shall, as soon as practicable, but in no event later than fifteen (15) Business Days after the Closing Date, use its reasonable best efforts to file a Registration Statement under the Securities Act to permit the public resale of all the Registrable Securities held by the Holders from time to time as permitted by Rule 415 under the Securities Act

(or any successor or similar provision adopted by the Commission then in effect) ("Rule 415") on the terms and conditions specified in this subsection 2.1.1 and shall use its reasonable best efforts to cause such Registration Statement to be declared effective as soon as practicable after the filing thereof, but in no event later than sixty (60) days following the filing deadline (the "Effectiveness Deadline"); provided, that the Effectiveness Deadline shall be extended to ninety (90) days after the filing deadline if the Registration Statement is reviewed by, and receives comments from, the Commission. The Registration Statement filed with the Commission pursuant to this subsection 2.1.1 shall be a shelf registration statement on Form S-3 (a "Form S-3") Shelf") or, if Form S-3 is not then available to the Company, on Form S-1 (a "Form S-1 Shelf") or such other form of registration statement as is then available to effect a registration for resale of such Registrable Securities, covering such Registrable Securities, and shall contain a Prospectus in such form as to permit any Holder to sell such Registrable Securities pursuant to Rule 415 at any time beginning on the effective date for such Registration Statement. A Registration Statement filed pursuant to this subsection 2.1.1 shall provide for the resale pursuant to any method or combination of methods legally available to, and requested by, the Holders. The Company shall use its reasonable best efforts to cause a Registration Statement filed pursuant to this <u>subsection 2.1.1</u> to remain effective, and to be supplemented and amended to the extent necessary to ensure that such Registration Statement is available or, if not available, that another Registration Statement is available, for the resale of all the Registrable Securities held by the Holders until all such Registrable Securities have ceased to be Registrable Securities. As soon as practicable following the effective date of a Registration Statement filed pursuant to this subsection 2.1.1, but in any event within one (1) Business Day of such date, the Company shall notify the Holders of the effectiveness of such Registration Statement. When effective, a Registration Statement filed pursuant to this subsection 2.1.1 (including the documents incorporated therein by reference) will comply as to form in all material respects with all applicable requirements of the Securities Act and the Exchange Act and will not contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading (in the case of any Prospectus contained in such Registration Statement, in the light of the circumstances under which such statement is made).

- 2.1.2. Form S-3 Shelf. If the Company files a Form S-3 Shelf and thereafter the Company becomes ineligible to use Form S-3 for secondary sales, the Company shall use its reasonable best efforts to file a Form S-1 Shelf as promptly as practicable to replace the shelf registration statement that is a Form S-3 Shelf and have the Form S-1 Shelf declared effective as promptly as practicable and to cause such Form S-1 Shelf to remain effective, and to be supplemented and amended to the extent necessary to ensure that such Registration Statement is available or, if not available, that another Registration Statement is available, for the resale of all the Registrable Securities held by the Holders until all such Registrable Securities have ceased to be Registrable Securities. In the event the Company files a Form S-1 Shelf, the Company shall use its commercially reasonable efforts to convert the Form S-1 Shelf (and any Subsequent Shelf Registration) to a Form S-3 Shelf as soon as practicable after the Company is eligible to use Form S-3, or any similar short-form registration.
- 2.1.3. <u>Requests for Underwritten Shelf Takedowns</u>. At any time and from time to time following the effectiveness of the shelf registration statement required by subsections 2.1.1 or 2.1.2, any Holder may request to sell all or a portion of their Registrable Securities in an underwritten offering that is registered pursuant to such shelf registration statement, including a

Block Trade (a "Shelf Underwritten Offering") provided that such Holder(s) reasonably expect aggregate gross proceeds in excess of \$75,000,000 from such Shelf Underwritten Offering. All requests for a Shelf Underwritten Offering shall be made by giving written notice to the Company (the "Shelf Takedown Notice"). Each Shelf Takedown Notice shall specify the approximate number of Registrable Securities proposed to be sold in the Shelf Underwritten Offering and the expected price range (net of underwriting discounts and commissions) of such Shelf Underwritten Offering. Within three (3) Business Days after receipt of any Shelf Takedown Notice (or twenty-four (24) hours thereafter in connection with an underwritten Block Trade), the Company shall give written notice of such requested Shelf Underwritten Offering to all other Holders of Registrable Securities (the "Company Shelf Takedown Notice") and, subject to reductions consistent with the Pro Rata calculations in subsection 2.2.4, shall include in such Shelf Underwritten Offering all Registrable Securities with respect to which the Company has received written requests for inclusion therein, within five (5) days after sending the Company Shelf Takedown Notice, or, in the case of a Block Trade, as provided in Section 2.5. The Company shall enter into an underwriting agreement in a form as is customary in Underwritten Offerings of securities by the Company with the managing Underwriter or Underwriters selected by the Company, subject to the prior approval of the initiating Holders (such approval not to be unreasonably withheld, conditioned or delayed) and shall take all such other reasonable actions as are requested by the managing Underwriter or Underwriters in order to expedite or facilitate the disposition of such Registrable Securities. In connection with any Shelf Underwritten Offering contemplated by this subsection 2.1.3, subject to Section 2.3 and Article IV, the underwriting agreement into which each Holder and the Company shall enter shall contain such representations, covenants, indemnities and other rights and obligations of the Company and the selling stockholders as are customary in underwritten offerings of securities by the Company. No Holder may demand more than two (2) Shelf Underwritten Offerings in any twelve (12) month period.

2.1.4. Holder Information Required for Participation in Shelf Registration. At least ten (10) Business Days prior to the first anticipated filing date of a Registration Statement pursuant to this Article II, the Company shall use reasonable efforts to notify each Holder in writing (which may be by email) of the information reasonably necessary about the Holder to include such Holder's Registrable Securities in such Registration Statement. Notwithstanding anything else in this Agreement, the Company shall not be obligated to include such Holder's Registrable Securities to the extent the Company has not received such information, and received any other reasonably requested agreements or certificates, on or prior to the fifth business day prior to the first anticipated filing date of a Registration Statement pursuant to this Article II.

## 2.2. Demand Registration.

2.2.1. <u>Request for Registration</u>. Subject to the provisions of <u>subsection 2.2.4</u> and <u>Section 2.4</u> hereof and provided that the Company does not have an effective Registration Statement pursuant to <u>subsection 2.1.1</u> outstanding covering Registrable Securities, (a) the Sponsor or (b) the New Holders of at least a majority-in-interest of the then-outstanding number of Registrable Securities held by the New Holders (the "*Demanding Holders*"), in each case, may make a written demand for Registration of all or part of their Registrable Securities, which written demand shall describe the amount and type of securities to be included in such Registration and the intended method(s) of distribution thereof (such written demand a "*Demand Registration*"). The Company shall, within three (3) Business Days of the Company's receipt of the Demand

Registration, notify, in writing, all other Holders of Registrable Securities of such demand, and each Holder of Registrable Securities who thereafter wishes to include all or a portion of such Holder's Registrable Securities in a Registration pursuant to a Demand Registration (each such Holder that includes all or a portion of such Holder's Registrable Securities in such Registration, a "Requesting Holder") shall so notify the Company, in writing, within five (5) Business Days after the receipt by the Holder of the notice from the Company. Upon receipt by the Company of any such written notification from a Requesting Holder(s) to the Company, which to be deemed timely hereunder shall include all information reasonably requested by the Company from such Requesting Holder(s) with respect to such Registration, such Requesting Holder(s) shall be entitled to have their Registrable Securities included in a Registration pursuant to a Demand Registration and the Company shall use its reasonable best efforts to effect, as soon thereafter as practicable, the Registration of all Registrable Securities requested by the Demanding Holders and Requesting Holders pursuant to such Demand Registration. Under no circumstances shall the Company be obligated to effect (x) more than an aggregate of two (2) Registrations pursuant to a Demand Registration by the Sponsor under this subsection 2.2.1 with respect to any or all Registrable Securities held by the Sponsor and its Permitted Transferees and (y) more than an aggregate of two (2) Registrations pursuant to a Demand Registration by the New Holders with respect to any or all Registrable Securities held by such New Holders and its Permitted Transferees; provided, however, that a Registration pursuant to a Demand Registration shall not be counted for such purposes unless a Form S-1 or any similar long form Registration Statement that may be available at such time has become effective and all of the Registrable Securities requested by the Requesting Holders and the Demanding Holders to be registered on behalf of the Requesting Holders and the Demanding Holders in such Form S-1 Registration Statement have been sold or have ceased to be Registrable Securities, in accordance with Section 3.1 of this Agreement.

- 2.2.2. Effective Registration. Notwithstanding the provisions of subsection 2.2.1 above or any other part of this Agreement, a Registration pursuant to a Demand Registration shall not count as a Registration unless and until (i) the Registration Statement filed with the Commission with respect to a Registration pursuant to a Demand Registration has been declared effective by the Commission and (ii) the Company has complied with all of its obligations under this Agreement in all material respects with respect thereto; provided, that if, after such Registration Statement has been declared effective, an offering of Registrable Securities in a Registration pursuant to a Demand Registration is subsequently interfered with by any stop order or injunction of the Commission, federal or state court or any other Governmental Entity, the Registration Statement with respect to such Registration shall be deemed not to have been declared effective unless and until (i) such stop order or injunction is removed, rescinded or otherwise terminated, and (ii) a majority-in-interest of the Demanding Holders initiating such Demand Registration thereafter affirmatively elect to continue with such Registration and accordingly notify the Company in writing, but in no event later than five (5) days, of such election; provided, further, that the Company shall not be obligated or required to file another Registration Statement until the Registration Statement that has been previously filed with respect to a Registration pursuant to a Demand Registration becomes effective or is subsequently terminated.
- 2.2.3. <u>Underwritten Offering</u>. Subject to the provisions of <u>subsection 2.2.4</u> and <u>Section 2.4</u> hereof, if a majority-in-interest of the Demanding Holders so advise the Company as part of their Demand Registration that the offering of the Registrable Securities pursuant to such Demand Registration shall be in the form of an Underwritten Offering, then the right of such

Demanding Holder or Requesting Holder (if any) to include its Registrable Securities in such Registration shall be conditioned upon such Holder's participation in such Underwritten Offering and the inclusion of such Holder's Registrable Securities in such Underwritten Offering to the extent provided herein. All such Holders proposing to distribute their Registrable Securities through an Underwritten Offering under this <u>subsection 2.2.3</u> shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by the majority-in-interest of the Demanding Holders initiating the Demand Registration, which Underwriter(s) shall be reasonably satisfactory to the Company.

- 2.2.4. Reduction of Underwritten Offering. If the managing Underwriter or Underwriters in an Underwritten Registration, in good faith, advises the Company, the Demanding Holders and the Requesting Holders (if any) in writing that the dollar amount or number of Registrable Securities that the Demanding Holders and the Requesting Holders (if any) desire to sell, taken together with all other Common Stock or other equity securities that the Company desires to sell and the Common Stock, if any, as to which a Registration has been requested pursuant to separate written contractual piggyback registration rights held by any other stockholders who desire to sell, exceeds the maximum dollar amount or maximum number of equity securities that can be sold in the Underwritten Offering without adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of such offering (such maximum dollar amount or maximum number of such securities, as applicable, the "Maximum Number of Securities"), then the Company shall include in such Underwritten Offering, as follows: (i) first, the Registrable Securities of the Demanding Holders (pro rata based on the respective number of Registrable Securities that each Demanding Holder has requested be included in such Underwritten Registration and the aggregate number of Registrable Securities that the Demanding Holders have requested be included in such Underwritten Registration (such proportion is referred to herein as "Pro Rata")) that can be sold without exceeding the Maximum Number of Securities; (ii) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (i), the Registrable Securities of Requesting Holders (Pro Rata, based on the respective number of Registrable Securities that each Requesting Holder has so requested) exercising their rights to register their Registrable Securities pursuant to subsection 2.2.1 hereof, without exceeding the Maximum Number of Securities; and (iii) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i) and (ii), the Common Stock or other equity securities of other persons or entities that the Company is obligated to register in a Registration pursuant to separate written contractual arrangements with such persons and that can be sold without exceeding the Maximum Number of Securities; and (iv) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i), (ii) and (iii), the Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities.
- 2.2.5. <u>Demand Registration Withdrawal</u>. A majority-in-interest of the Demanding Holders initiating a Demand Registration or a majority-in-interest of the Requesting Holders (if any) pursuant to a Registration under <u>subsection 2.2.1</u> shall have the right to withdraw from a Registration pursuant to such Demand Registration or a Shelf Underwritten Offering pursuant to <u>subsection 2.1.3</u> for any or no reason whatsoever upon written notification to the Company and the Underwriter or Underwriters (if any) of their intention to withdraw from such Registration at least one (1) business day prior to the effectiveness of the Registration Statement filed with the Commission with respect to the Registration of their Registrable Securities pursuant to such

Demand Registration (or in the case of an Underwritten Registration pursuant to Rule 415, at least five (5) Business Days prior to the time of pricing of the applicable offering). Notwithstanding anything to the contrary in this Agreement, (i) the Company may effect any Underwritten Registration pursuant to any then effective Registration Statement, including a Form S-3 Shelf, that is then available for such offering, (ii) the Company shall be responsible for the Registration Expenses incurred in connection with a Registration pursuant to a Demand Registration or a Shelf Underwritten Offering prior to its withdrawal under this <u>subsection 2.2.5</u> and (iii) any such withdrawn Demand Registration shall constitute a completed Demand Registration for purposes of determining the number of Demand Registrations that may be requested by the Holders pursuant to <u>subsection 2.2.1</u>.

#### 2.3. Piggyback Registration.

2.3.1. Piggyback Rights. If the Company proposes to file a Registration Statement under the Securities Act with respect to an offering of equity securities, or securities or other obligations exercisable or exchangeable for, or convertible into, equity securities, for its own account or for the account of stockholders of the Company (or by the Company and by the stockholders of the Company), other than a Registration Statement (i) filed pursuant to Section 2.2, (ii) filed in connection with any employee stock option or other benefit plan, (iii) for a rights offering or an exchange offer or offering of securities solely to the Company's existing stockholders, (iv) for an offering of debt that is convertible into equity securities of the Company (v) pursuant to a Registration Statement on Form S-4 (or similar form that relates to a transaction subject to Rule 145 under the Securities Act or any successor rule thereto) or (vi) for a dividend reinvestment plan, then the Company shall give written notice of such proposed filing to all of the Holders of Registrable Securities as soon as practicable but not less than three (3) Business Days before the anticipated filing date of such Registration Statement, which notice shall (A) describe the amount and type of securities to be included in such offering, the intended method(s) of distribution, and the name of the proposed managing Underwriter or Underwriters, if any, in such offering, and (B) offer to all of the Holders of Registrable Securities the opportunity to register the sale of such number of Registrable Securities as such Holders may request in writing within five (5) Business Days after receipt of such written notice (such Registration a "Piggyback Registration"); provided, that each Holder of Registrable Securities agrees that the fact that such a notice has been delivered shall constitute confidential information; provided further, that the exercise of any piggy-back rights with respect to any block trade should be done no later than twenty four (24) hours following receipt of any written notice regarding such Block Trade. The Company shall, in good faith, cause such Registrable Securities to be included in such Piggyback Registration and shall use its reasonable best efforts to cause the managing Underwriter or Underwriters of a proposed Underwritten Offering to permit the Registrable Securities requested by the Holders pursuant to this subsection 2.3.1 to be included in a Piggyback Registration on the same terms and conditions as any similar securities of the Company included in such Registration and to permit the sale or other disposition of such Registrable Securities in accordance with the intended method(s) of distribution thereof. All such Holders proposing to distribute their Registrable Securities through an Underwritten Offering under this subsection 2.3.1 shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by the Company.

- 2.3.2. <u>Reduction of Piggyback Registration</u>. If the managing Underwriter or Underwriters in an Underwritten Registration that is to be a Piggyback Registration, in good faith, advises the Company and the Holders of Registrable Securities participating in the Piggyback Registration in writing that the dollar amount or number of Common Stock that the Company desires to sell, taken together with (i) the Common Stock, if any, as to which Registration has been demanded pursuant to separate written contractual arrangements with persons or entities other than the Holders of Registrable Securities hereunder, (ii) the Registrable Securities as to which registration has been requested pursuant to <u>Section 2.3</u> hereof, and (iii) the Common Stock, if any, as to which Registration has been requested pursuant to separate written contractual piggy-back registration rights of other stockholders of the Company, exceeds the Maximum Number of Securities, then:
  - (a) If the Registration is undertaken for the Company's account, the Company shall include in any such Registration (A) first, the Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to subsection 2.3.1 hereof, Pro Rata, which can be sold without exceeding the Maximum Number of Securities and (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), the Common Stock, if any, as to which Registration has been requested or demanded pursuant to written contractual piggy-back registration rights of other stockholders of the Company, which can be sold without exceeding the Maximum Number of Securities;
  - (b) If the Registration is pursuant to a request by persons or entities other than the Holders of Registrable Securities, then the Company shall include in any such Registration (A) first, the Common Stock or other equity securities, if any, of such requesting persons or entities, other than the Holders of Registrable Securities, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to <a href="subsection 2.3.1">subsection 2.3.1</a>, Pro Rata, which can be sold without exceeding the Maximum Number of Securities; and (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities.
- 2.3.3. <u>Piggyback Registration Withdrawal</u>. Any Holder of Registrable Securities shall have the right to withdraw from a Piggyback Registration for any or no reason whatsoever upon written notification to the Company and the Underwriter or Underwriters (if any) of his, her or its intention to withdraw from such Piggyback Registration prior to the effectiveness of the Registration Statement filed with the Commission with respect to such Piggyback Registration (or in the case of an Underwritten Registration pursuant to Rule 415, at least two (2) Business Days prior to the time of pricing of the applicable offering). The Company (whether on its own good faith determination or as the result of a request for withdrawal by persons pursuant to separate

written contractual obligations) may withdraw a Registration Statement filed with the Commission in connection with a Piggyback Registration at any time prior to the effectiveness of such Registration Statement. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with the Piggyback Registration prior to its withdrawal under this <u>subsection 2.3.3</u>.

- 2.3.4. <u>Unlimited Piggyback Registration Rights</u>. For purposes of clarity, any Registration effected pursuant to <u>Section 2.3</u> hereof shall not be counted as a Registration pursuant to a Demand Registration effected under <u>Section 2.2</u> hereof or a Shelf Underwritten Offering effected under subsection 2.1.3.
- 2.4. Restrictions on Registration Rights. If (A) during the period starting with the date sixty (60) days prior to the Company's good faith estimate of the date of the filing of, and ending on a date one hundred and twenty (120) days after the effective date of, a Company initiated Registration and provided that the Company has delivered written notice to the Holders prior to receipt of a Demand Registration pursuant to subsection 2.2.1 and it continues to actively employ, in good faith, all reasonable efforts to cause the applicable Registration Statement to become effective; (B) the Holders have requested an Underwritten Registration and the Company and the Holders are unable to obtain the commitment of underwriters to firmly underwrite the offer; or (C) in the good faith judgment of the Board such Registration would be seriously detrimental to the Company and the Board concludes as a result that it is essential to defer the filing of such Registration Statement at such time, then in each case the Company shall furnish to such Holders a certificate signed by the Chairman of the Board stating that in the good faith judgment of the Board it would be seriously detrimental to the Company for such Registration Statement to be filed in the near future and that it is therefore essential to defer the filing of such Registration Statement, then the Company shall have the right to defer such filing for a period of not more than sixty (60) days; provided, however, that the Company shall not defer its obligation in this manner more than twice in any 12-month period (the "Aggregate Blocking Period").
- 2.5. <u>Block Trades</u>. Notwithstanding any other provision of this Article II, but subject to <u>Sections 2.4</u> and <u>3.4</u>, if the Holders desire to effect a Block Trade, the Holders shall provide written notice to the Company at least five (5) Business Days prior to the date such Block Trade will commence. As expeditiously as possible, the Company shall use its reasonable best efforts to facilitate such Block Trade. The Holders shall use reasonable best efforts to work with the Company and the Underwriter(s) (including by disclosing the maximum number of Registrable Securities proposed to be the subject of such Block Trade) in order to facilitate preparation of the Registration Statement, Prospectus and other offering documentation related to the Block Trade and any related due diligence and comfort procedures.
- 2.6. <u>Rule 415; Removal</u>. If at any time the Commission takes the position that the offering of some or all of the Registrable Securities in a Registration Statement on Form S-3 filed pursuant to this <u>Article II</u>, is not eligible to be made on a delayed or continuous basis under the provisions of Rule 415 under the Securities Act (<u>provided</u>, <u>however</u>, the Company shall be obligated to use diligent efforts to advocate with the Commission for the registration of all of the Registrable Securities in accordance with the Commission Guidance, including without limitation, Compliance and Disclosure Interpretation 612.09) or requires a Holder to be named as an "underwriter," the Company shall (i) promptly notify each holder of Registrable Securities thereof

(or in the case of the Commission requiring a Holder to be named as an "underwriter," the Holders) and (ii) use reasonable best efforts to persuade the SEC that the offering contemplated by such Registration Statement is a valid secondary offering and not an offering "by or on behalf of the issuer" as defined in Rule 415 and that none of the Holders is an "underwriter." The Holders shall have the right to select one legal counsel designated by the holders of a majority of the Registrable Securities subject to such Registration Statement to review and oversee any registration or matters pursuant to this Section 2.6, including participation in any meetings or discussions with the Commission regarding the Commission's position and to comment on any written submission made to the Commission with respect thereto. No such written submission with respect to this matter shall be made to the Commission to which the applicable Holders' counsel reasonably objects. In the event that, despite the Company's reasonable best efforts and compliance with the terms of this Section 2.6, the Commission refuses to alter its position, the Company shall (i) remove from such Registration Statement such portion of the Registrable Securities (the "Removed Shares") and/or (ii) agree to such restrictions and limitations on the registration and resale of the Registrable Securities as the Commission may require to assure the Company's compliance with the requirements of Rule 415; provided, however, that the Company shall not agree to name any Holder as an "underwriter" in such Registration Statement without the prior written consent of such Holder. In the event of a share removal pursuant to this Section 2.6, the Company shall give the applicable Holders at least three (3) Business Days prior written notice along with the calculations as to such Holder's allotment. Any removal of shares of the Holders pursuant to this Section 2.6 shall be allocated between the Holders on a pro rata basis based on the aggregate amount of Registrable Securities held by the Holders. In the event of a share removal of the Holders pursuant to this Section 2.6, the Company shall promptly register the resale of any Removed Shares pursuant to subsection 2.1.2 hereof and in no event shall the filing of such Registration Statement on Form S-1 or subsequent Registration Statement on Form S-3 filed pursuant to the terms of <u>subsection 2.1.2</u> be counted as a Demand Registration hereunder.

# ARTICLE III COMPANY PROCEDURES

- 3.1. <u>General Procedures</u>. If the Company is required to effect the Registration of Registrable Securities, the Company shall use its reasonable best efforts to effect such Registration to permit the sale of such Registrable Securities in accordance with the intended plan of distribution thereof, and pursuant thereto the Company shall, as expeditiously as reasonably possible:
- 3.1.1. prepare and file with the Commission as soon as reasonably practicable a Registration Statement with respect to such Registrable Securities and use its reasonable best efforts to cause such Registration Statement to become effective and remain effective until all Registrable Securities covered by such Registration Statement have been sold or have ceased to be Registrable Securities;
- 3.1.2. prepare and file with the Commission such amendments and post-effective amendments to the Registration Statement, and such supplements to the Prospectus, as may be reasonably requested by any majority-in-interest of the Holders with Registrable Securities registered on such Registration Statement or any Underwriter of Registrable Securities or as may be required by the rules, regulations or instructions applicable to the registration form used by the Company or by the Securities Act or rules and regulations thereunder to keep the Registration Statement effective until all Registrable Securities covered by such Registration Statement are sold in accordance with the intended plan of distribution set forth in such Registration Statement or supplement to the Prospectus;

- 3.1.3. prior to filing a Registration Statement or Prospectus, or any amendment or supplement thereto, furnish without charge to the Underwriters, if any, and each Holder of Registrable Securities included in such Registration, and such Holders' legal counsel, copies of such Registration Statement as proposed to be filed, each amendment and supplement to such Registration Statement (in each case including all exhibits thereto and documents incorporated by reference therein), the Prospectus included in such Registration Statement (including each preliminary Prospectus), and such other documents as the Underwriters and each Holder of Registrable Securities included in such Registration or the legal counsel for any such Holders may reasonably request in order to facilitate the disposition of the Registrable Securities owned by such Holders;
- 3.1.4. prior to any public offering of Registrable Securities, use its reasonable best efforts to (i) register or qualify the Registrable Securities covered by the Registration Statement under such securities or "blue sky" Laws of such jurisdictions in the United States as any Holder of Registrable Securities included in such Registration Statement (in light of their intended plan of distribution) may reasonably request (or provide evidence satisfactory to such Holders that the Registrable Securities are exempt from such registration or qualification) and (ii) take such action necessary to cause such Registrable Securities covered by the Registration Statement to be registered with or approved by such other Governmental Entities as may be necessary by virtue of the business and operations of the Company and do any and all other acts and things that may be necessary or advisable to enable the Holders of Registrable Securities included in such Registration Statement to consummate the disposition of such Registrable Securities in such jurisdictions; provided, however, that the Company shall not be required to qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify or take any action to which it would be subject to general service of process or taxation in any such jurisdiction where it is not then otherwise so subject;
- 3.1.5. cause all such Registrable Securities to be listed on each securities exchange or automated quotation system on which similar securities issued by the Company are then listed;
- 3.1.6. provide a transfer agent or warrant agent, as applicable, and registrar for all such Registrable Securities no later than the effective date of such Registration Statement;
- 3.1.7. advise each seller of such Registrable Securities, promptly after it shall receive notice or obtain knowledge thereof, of the issuance of any stop order by the Commission suspending the effectiveness of such Registration Statement or the initiation or threatening of any proceeding for such purpose and promptly use its reasonable best efforts to prevent the issuance of any stop order or to obtain its withdrawal if such stop order should be issued;
- 3.1.8. at least five (5) days prior to the filing of any Registration Statement or Prospectus or any amendment or supplement to such Registration Statement or Prospectus (or such shorter period of time as may be (a) necessary in order to comply with the Securities Act, the Exchange Act and the rules and regulations promulgated thereunder or (b) advisable in order to reduce the

number of days that sales are suspended pursuant to <u>Section 3.4</u>), furnish a copy thereof to each seller of such Registrable Securities and its counsel, including, without limitation, providing copies promptly upon receipt of any comment letters received with respect to any such Registration Statement or Prospectus;

- 3.1.9. notify the Holders at any time when a Prospectus relating to such Registration Statement is required to be delivered under the Securities Act, of the happening of any event as a result of which the Prospectus included in such Registration Statement, as then in effect, includes a Misstatement, and then to correct such Misstatement as set forth in Section 3.4 hereof;
- 3.1.10. permit a representative of the Holders (such representative to be selected by a majority of the participating Holders), the Underwriter(s), if any, and any attorney or accountant retained by such Holders or Underwriter(s) to participate, at each such person's own expense, in the preparation of the Registration Statement, and cause the Company's officers, directors and employees to supply all information reasonably requested by any such representative, Underwriter, attorney or accountant in connection with the Registration; <u>provided</u>, <u>however</u>, that such representative or Underwriter enters into a confidentiality agreement, in form and substance reasonably satisfactory to the Company, prior to the release or disclosure of any such information;
- 3.1.11. obtain a "cold comfort" letter from the Company's independent registered public accountants in the event of an Underwritten Registration, in customary form and covering such matters of the type customarily covered by "cold comfort" letters as the managing Underwriter(s) may reasonably request;
- 3.1.12. on the date the Registrable Securities are delivered for sale pursuant to such Registration, obtain an opinion, dated such date, of counsel representing the Company for the purposes of such Registration, addressed to the placement agent or sales agent, if any, and the Underwriter(s), if any, covering such legal matters with respect to the Registration in respect of which such opinion is being given as the Underwriter(s), placement agent(s) or sales agent(s) may reasonably request and as are customarily included in such opinions and negative assurance letters;
- 3.1.13. in the event of any Underwritten Offering, enter into and perform its obligations under an underwriting agreement, in usual and customary form, with the managing Underwriter of such offering;
- 3.1.14. make available to its security holders, as soon as reasonably practicable, an earnings statement covering the period of at least twelve (12) months beginning with the first day of the Company's first full calendar quarter after the effective date of the Registration Statement which satisfies the provisions of Section 11(a) of the Securities Act and Rule 158 thereunder (or any successor rule promulgated thereafter by the Commission);
- 3.1.15. in the event of an Underwritten offering in which the Registration involves the Registration of Registrable Securities involving gross proceeds in excess of \$50,000,000, use its reasonable efforts to make available senior executives of the Company to participate in customary "road show" presentations that may be reasonably requested by the Underwritter(s) in any Underwritten Offering; and

- 3.1.16. otherwise, in good faith, cooperate reasonably with, and take such customary actions as may reasonably be requested by the participating Holders, consistent with the terms of this Agreement, in connection with such Registration, including, without limitation, making available senior executives of the Company to participate in any due diligence sessions that may be reasonably requested by the Underwriter in any Underwritten Offering.
- 3.2. <u>Registration Expenses</u>. Except as otherwise provided herein, the Registration Expenses of all Registrations shall be borne by the Company. It is acknowledged by the Holders that the Holders shall bear all incremental selling expenses relating to the sale of Registrable Securities, such as Underwriters' commissions and discounts, brokerage fees, Underwriter marketing costs and, other than as set forth in the definition of "Registration Expenses," all reasonable fees and expenses of any legal counsel representing the Holders.
- 3.3. Requirements for Participation in Registrations. Notwithstanding anything in this Agreement to the contrary, if any Holder does not provide the Company with the information reasonably necessary about the Holder to include such Holder's Registrable Securities in a Registration Statement, the Company may exclude such Holder's Registrable Securities from such Registration Statement or related Prospectus if the Company determines, based on the advice of counsel, that such information is necessary to effect the registration and such Holder continues thereafter to withhold such information. No person may participate in any Underwritten Offering for equity securities of the Company pursuant to a Registration initiated by the Company hereunder unless such person (i) agrees to sell such person's securities on the basis provided in any underwriting, sales, placement or distribution arrangements approved by the Company and (ii) completes and executes all customary questionnaires, powers of attorney, indemnities, lock-up agreements, underwriting or other agreements and other customary documents as may be reasonably required under the terms of such underwriting, sales, distribution or placement arrangements.
- 3.4. Suspension of Sales; Adverse Disclosure. Upon receipt of written notice from the Company that a Registration Statement or Prospectus contains a Misstatement, each of the Holders shall forthwith discontinue disposition of Registrable Securities until he, she or it has received copies of a supplemented or amended Prospectus correcting the Misstatement (it being understood that the Company hereby covenants to prepare and file such supplement or amendment as soon as reasonably practicable after the time of such notice), or until he, she or it is advised in writing by the Company that the use of the Prospectus may be resumed. If the filing, initial effectiveness or continued use of a Registration Statement in respect of any Registration at any time would require the Company to make an Adverse Disclosure or would require the inclusion in such Registration Statement of financial statements that are unavailable to the Company for reasons beyond the Company's reasonable control, the Company may, upon giving prompt written notice of such action to the Holders, delay the filing or initial effectiveness of, or suspend use of, such Registration Statement for the shortest period of time, but in no event more than ninety (90) days, determined in good faith by the Company to be necessary for such purpose; provided, that each day of any such suspension pursuant to this Section 3.4 shall correspondingly decrease the Aggregate Blocking Period available to the Company during any twelve (12) month period pursuant to Section 2.4 hereof. In the event the Company exercises its rights under the preceding sentence, the Holders agree to suspend, immediately upon their receipt of the notice referred to above, their use of the Prospectus relating to any Registration in connection with any sale or offer to sell Registrable Securities. The Company shall notify the Holders within one (1) Business Day of the expiration of any period during which it exercised its rights under this Section 3.4.

3.5. Reporting Obligations. As long as any Holder shall own Registrable Securities, the Company, at all times while it shall be a reporting company under the Exchange Act, covenants to file timely (or obtain extensions in respect thereof and file within the applicable grace period) all reports required to be filed by the Company after the date hereof pursuant to Sections 13(a) or 15(d) of the Exchange Act and to promptly furnish the Holders with true and complete copies of all such filings; provided that any documents publicly filed or furnished with the Commission pursuant to the Electronic Data Gathering, Analysis and Retrieval System shall be deemed to have been furnished or delivered to the Holders pursuant to this Section 3.5. The Company further covenants that it shall take such further action as any Holder may reasonably request, all to the extent required from time to time to enable such Holder to sell shares of the Common Stock held by such Holder without registration under the Securities Act within the limitation of the exemptions provided by Rule 144. Upon the request of any Holder, the Company shall deliver to such Holder a written certification of a duly authorized officer as to whether it has complied with such requirements.

# ARTICLE IV INDEMNIFICATION AND CONTRIBUTION

#### 4.1. Indemnification.

- 4.1.1. The Company agrees to indemnify, to the extent permitted by Law, each Holder of Registrable Securities, its officers and directors and each person who controls such Holder (within the meaning of the Securities Act) against all losses, claims, damages, liabilities and out-of-pocket expenses (including without limitation reasonable attorneys' fees) resulting from or based upon any untrue or alleged untrue statement of material fact contained in any Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, except insofar as the same are caused by or contained in any information furnished in writing to the Company by such Holder expressly for use therein. The Company shall indemnify the Underwriters, their officers and directors and each person who controls such Underwriters (within the meaning of the Securities Act) to the same extent as provided in the foregoing with respect to the indemnification of the Holder.
- 4.1.2. In connection with any Registration Statement in which a Holder of Registrable Securities is participating, such Holder shall furnish to the Company in writing such information and affidavits as the Company reasonably requests for use in connection with any such Registration Statement or Prospectus and, to the extent permitted by Law, shall indemnify the Company, its directors and officers and agents and each person who controls the Company (within the meaning of the Securities Act) against any losses, claims, damages, liabilities and expenses (including without limitation reasonable attorneys' fees) resulting from or based upon any untrue statement of material fact contained in the Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, but only to the extent that such untrue statement or omission is contained in any information or affidavit so

furnished in writing by such Holder expressly for use therein; <u>provided</u>, <u>however</u>, that the obligation to indemnify shall be several, not joint and several, among such Holders of Registrable Securities, and the liability of each such Holder of Registrable Securities shall be in proportion to and limited to the net proceeds received by such Holder from the sale of Registrable Securities pursuant to such Registration Statement. The Holders of Registrable Securities shall indemnify the Underwriters, their officers, directors and each person who controls such Underwriters (within the meaning of the Securities Act) to the same extent as provided in the foregoing with respect to indemnification of the Company.

- 4.1.3. Any person entitled to indemnification herein shall (i) give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification (provided, that the failure to give prompt notice shall not impair any person's right to indemnification hereunder to the extent such failure has not materially prejudiced the indemnifying party) and (ii) unless in such indemnified party's reasonable judgment a conflict of interest between such indemnified and indemnifying parties may exist with respect to such claim, permit such indemnifying party to assume the defense of such claim with counsel reasonably satisfactory to the indemnified party. If such defense is assumed, the indemnifying party shall not be subject to any liability for any settlement made by the indemnified party without its consent (but such consent shall not be unreasonably withheld). An indemnifying party who is not entitled to, or elects not to, assume the defense of a claim shall not be obligated to pay the fees and expenses of more than one counsel for all parties indemnified by such indemnifying party with respect to such claim, unless in the reasonable judgment of any indemnified party a conflict of interest may exist between such indemnified party and any other of such indemnified parties with respect to such claim. No indemnifying party shall, without the consent of the indemnified party, consent to the entry of any judgment or enter into any settlement which cannot be settled in all respects by the payment of money (and such money is so paid by the indemnified party pursuant to the terms of such settlement) or which settlement includes a statement or admission of fault and culpability on the part of such indemnified party or which settlement does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnified party of a release from all liability in respect to such claim or litigation.
- 4.1.4. The indemnification provided for under this Agreement shall remain in full force and effect regardless of any investigation made by or on behalf of the indemnified party or any officer, director or controlling person of such indemnified party and shall survive the Transfer of securities. The Company and each Holder of Registrable Securities participating in an offering also agrees to make such provisions as are reasonably requested by any indemnified party for contribution to such party in the event the Company's or such Holder's indemnification is unavailable for any reason.
- 4.1.5. If the indemnification provided under Section 4.1 hereof from the indemnifying party is unavailable or insufficient to hold harmless an indemnified party in respect of any losses, claims, damages, liabilities and out-of-pocket expenses referred to herein, then the indemnifying party, in lieu of indemnifying the indemnified party, shall contribute to the amount paid or payable by the indemnified party as a result of such losses, claims, damages, liabilities and out-of-pocket expenses in such proportion as is appropriate to reflect the relative fault of the indemnifying party and the indemnified party, as well as any other relevant equitable considerations. The relative fault of the indemnifying party and indemnified party shall be determined by reference to, among other

things, whether any action in question, including any untrue or alleged untrue statement of a material fact or omission or alleged omission to state a material fact, was made (or not made by, in the case of an omission) by, or relates to information supplied by (or not supplied by, in the case of an omission), such indemnifying party or indemnified party, and the indemnifying party's and indemnified party's relative intent, knowledge, access to information and opportunity to correct or prevent such action; provided, however, that the liability of any Holder under this subsection 4.1.5 shall be limited to the amount of the net proceeds received by such Holder in such offering giving rise to such liability. The amount paid or payable by a party as a result of the losses or other liabilities referred to above shall be deemed to include, subject to the limitations set forth in subsections 4.1.1, 4.1.2 and 4.1.3 above, any legal or other fees, charges or out-of-pocket expenses reasonably incurred by such party in connection with any investigation or proceeding. The parties hereto agree that it would not be just and equitable if contribution pursuant to this subsection 4.1.5 were determined by Pro Rata allocation or by any other method of allocation, which does not take account of the equitable considerations referred to in this subsection 4.1.5. No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution pursuant to this subsection 4.1.5 from any person who was not guilty of such fraudulent misrepresentation.

## ARTICLE V MISCELLANEOUS

5.1. Notices. Any notice or communication under this Agreement must be in writing and given by (i) deposit in the United States mail, addressed to the party to be notified, postage prepaid and registered or certified with return receipt requested, (ii) delivery in person or by courier service providing evidence of delivery, or (iii) transmission by hand delivery, electronic mail or facsimile. Each notice or communication that is mailed, delivered, or transmitted in the manner described above shall be deemed sufficiently given, served, sent, and received, in the case of mailed notices, on the third business day following the date on which it is mailed and, in the case of notices delivered by courier service, hand delivery, electronic mail or facsimile, at such time as it is delivered to the addressee (with the delivery receipt or the affidavit of messenger) or at such time as delivery is refused by the addressee upon presentation. Any notice or communication under this Agreement must be addressed, if to the Company, to: Origin Materials, Inc., 930 Riverside Parkway, Suite 10, West Sacramento, CA 95605, Attention: General Counsel, Email: jlee@originmaterials.com, and, if to any Holder, at such Holder's address, electronic mail address or facsimile number as set forth in the Company's books and records. Any party may change its address for notice at any time and from time to time by written notice to the other parties hereto, and such change of address shall become effective thirty (30) days after delivery of such notice as provided in this Section 5.1.

#### 5.2. <u>Assignment; No Third Party Beneficiaries</u>.

5.2.1. This Agreement and the rights, duties and obligations of the Company and the Holders of Registrable Securities, as the case may be, hereunder may not be assigned or delegated by the Company or the Holders of Registrable Securities, as the case may be, in whole or in part, except in connection with a Transfer of Registrable Securities by such Holder to a Permitted Transferee but only if such Permitted Transferee agrees to become bound by the restrictions set forth in this Agreement.

- 5.2.2. This Agreement and the provisions hereof shall be binding upon and shall inure to the benefit of each of the parties and its successors and the permitted assigns of the Holders, which shall include Permitted Transferees.
- 5.2.3. This Agreement shall not confer any rights or benefits on any persons that are not parties hereto, other than as expressly set forth in this Agreement and Section 5.2 hereof.
- 5.2.4. No assignment by any party hereto of such party's rights, duties and obligations hereunder shall be binding upon or obligate the Company unless and until the Company shall have received (i) written notice of such assignment as provided in <u>Section 5.1</u> hereof and (ii) the written agreement of the assignee, in a form reasonably satisfactory to the Company, to be bound by the terms and provisions of this Agreement (which may be accomplished by an addendum or certificate of joinder to this Agreement). Any transfer or assignment made other than as provided in this <u>Section 5.2</u> shall be null and void.
- 5.3. <u>Counterparts</u>. This Agreement may be executed in multiple counterparts (including facsimile or PDF counterparts), each of which shall be deemed an original, and all of which together shall constitute the same instrument, but only one of which need be produced.
- 5.4. Governing Law; Venue; Waiver of Trial by Jury. NOTWITHSTANDING THE PLACE WHERE THIS AGREEMENT MAY BE EXECUTED BY ANY OF THE PARTIES HERETO, THE PARTIES EXPRESSLY AGREE THAT THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED UNDER THE LAWS OF THE STATE OF NEW YORK AS APPLIED TO AGREEMENTS AMONG NEW YORK RESIDENTS ENTERED INTO AND TO BE PERFORMED ENTIRELY WITHIN NEW YORK, WITHOUT REGARD TO THE CONFLICT OF LAW PROVISIONS OF SUCH JURISDICTION. ANY LEGAL SUIT, ACTION OR PROCEEDING ARISING OUT OF OR BASED UPON THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED THEREBY MAY BE INSTITUTED IN THE FEDERAL COURTS OF THE UNITED STATES OR THE COURTS OF THE STATE OF NEW YORK IN EACH CASE LOCATED IN THE CITY OF NEW YORK, AND EACH PARTY IRREVOCABLY SUBMITS TO THE EXCLUSIVE JURISDICTION OF SUCH COURTS IN ANY SUCH SUIT, ACTION OR PROCEEDING.

EACH PARTY HERETO ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND, THEREFORE, EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT TO ANY ACTION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT.

5.5. <u>Amendments and Modifications</u>. Upon the written consent of the Company and the Holders of at least a majority in interest of the Registrable Securities at the time in question, compliance with any of the provisions, covenants and conditions set forth in this Agreement may be waived, or any of such provisions, covenants or conditions may be amended or modified; <u>provided, however</u>, that notwithstanding the foregoing, any amendment hereto or waiver hereof that

adversely affects one Holder or group of affiliated Holders, solely in its capacity as a holder of the shares of capital stock of the Company, in a manner that is materially different from the other Holders (in such capacity) shall require the consent of the Holder or group of affiliated Holders so affected. No course of dealing between any Holder or the Company and any other party hereto or any failure or delay on the part of a Holder or the Company in exercising any rights or remedies under this Agreement shall operate as a waiver of any rights or remedies of any Holder or the Company. No single or partial exercise of any rights or remedies under this Agreement by a party shall operate as a waiver or preclude the exercise of any other rights or remedies hereunder or thereunder by such party.

- 5.6. Other Registration Rights. The Company represents and warrants that no person, other than a Holder of Registrable Securities, has any right to require the Company to register any securities of the Company for sale or to include such securities of the Company in any Registration filed by the Company for the sale of securities for its own account or for the account of any other person. The parties hereby terminate the (i) Existing Registration Rights Agreement, (ii) the Second Amended and Restated Founders' Agreement, dated as of March 30, 2010, by and among Micromidas, Inc. and certain individuals as listed therein, and (iii) the Third Amended and Restated Investors' Rights Agreement, dated as of August 3, 2018, by and among Micromidas, Inc. and certain individuals as listed therein; in each case, which shall be of no further force and effect and are hereby superseded and replaced in their entirety by this Agreement. Further, the Company represents and warrants that this Agreement supersedes any other registration rights agreement or agreement with similar terms and conditions and in the event of a conflict between any such agreement or agreements and this Agreement, the terms of this Agreement shall prevail.
- 5.7. <u>Term.</u> This Agreement shall terminate upon the earlier of (i) the tenth anniversary of the date of this Agreement and (ii) with respect to any Holder, the date as of which such Holder no longer holds any Registrable Securities. The provisions of <u>Section 2.5</u> and <u>Article V</u> shall survive any termination.
- 5.8. <u>Severability</u>. If any provision of this Agreement shall be invalid, illegal or unenforceable, the validity, legality and enforceability of the remaining provisions shall not in any way be affected or impaired thereby.

[SIGNATURE PAGES FOLLOW]

**IN WITNESS WHEREOF**, the undersigned have caused this Agreement to be executed as of the date first written above.

#### **COMPANY:**

ORIGIN MATERIALS, INC., a Delaware corporation

By: /s/ John Bissell

Name: John Bissell

Title: President and Co-Chief Executive Officer

## THE SPONSOR:

ARTIUS ACQUISITION PARTNERS LLC, a Delaware limited liability company

By: /s/ Boon Sim

Name: Boon Sim

Title: Chief Executive Officer

[Signature Page to Amended and Restated Investor Rights Agreement]

#### **NEW HOLDERS:**

#### John Bissell

By: /s/ John Bissell

Name: John Bissell

### **Charles Drucker**

By: /s/ Charles Drucker

Name: Charles Drucker

## **Richard Riley**

By: /s/ Richard Riley

Name: Richard Riley

## **Riley Family Trust**

By: /s/ Richard Riley

Name: Richard Riley Title: Trustee

# **Riley Family Trust**

By: /s/ Richard Riley

Name: Richard Riley Title: Trustee

## Riley Investment Trust I

By: /s/ Richard Riley

Name: Richard Riley Title: Trustee

## **Riley Separate Property Trust**

By: /s/ Richard Riley

Name: Richard Riley Title: Trustee

# **Stephen and Jill Galowitz JTWROS**

By: /s/ Stephen Galowitz

Name: Stephen Galowitz

By: /s/ Jill Galowitz

Name: Jill Galowitz

[Signature Page to Amended and Restated Investor Rights Agreement]

# **Stephen Galowitz**

By: /s/ Stephen Galowitz

Name: Stephen Galowitz

# **Nate Whaley**

By: /s/ Nate Whaley

Name: Nate Whaley

## Joshua Lee

By: /s/ Joshua Lee

Name: Joshua Lee

# William Harvey

By: /s/ William Harvey

Name: William Harvey

#### INDEMNITY AGREEMENT

THIS INDEMNITY AGREEMENT (this "Agreement")	) dated as of [	_], is made by and between <b>ORIGIN MATERIALS, INC.,</b> a
Delaware corporation (the "Company") and	("Indemnitee").	

#### RECITALS

- A. The Company desires to attract and retain the services of highly qualified individuals as directors, officers, employees and agents.
- **B.** The Company's amended and restated bylaws (the "**Bylaws**") require that the Company indemnify its directors and officers, and empowers the Company to indemnify its employees and agents, as authorized by the Delaware General Corporation Law, as amended (the "**Code**"), under which the Company is organized and such Bylaws expressly provide that the indemnification provided therein is not exclusive and contemplates that the Company may enter into separate agreements with its directors, officers and other persons to set forth specific indemnification provisions.
- **C.** Indemnite does not regard the protection currently provided by applicable law, the Bylaws, the Company's other governing documents, and available insurance as adequate under the present circumstances, and the Company has determined that Indemnitee and other directors, officers, employees and agents of the Company may not be willing to serve or continue to serve in such capacities without additional protection.
- **D.** The Company desires and has requested Indemnitee to serve or continue to serve as a director, officer, employee or agent of the Company, as the case may be, and has proffered this Agreement to Indemnitee as an additional inducement to serve in such capacity.
- **E.** Indemnitee is willing to serve, or to continue to serve, as a director, officer, employee or agent of the Company, as the case may be, if Indemnitee is furnished the indemnity provided for herein by the Company.

#### AGREEMENT

**NOW THEREFORE**, in consideration of the mutual covenants and agreements set forth herein, the parties hereto, intending to be legally bound, hereby agree as follows:

#### 1. Definitions.

(a) Agent. For purposes of this Agreement, the term "Agent" of the Company means any person who: (i) is or was a director, officer, employee, agent, or other fiduciary of the Company or a subsidiary of the Company; or (ii) is or was serving at the request or for the convenience of, or representing the interests of, the Company or a subsidiary of the Company, as a director, officer, employee, agent, or other fiduciary of a foreign or domestic corporation, partnership, joint venture, trust or other enterprise.

- **(b)** Change in Control. For purposes of this Agreement, a "Change in Control" shall be deemed to have occurred if (i) any "person" (as such term is used in Sections 13(d) and 14(d) of the Securities Exchange Act of 1934, as amended), other than a trustee or other fiduciary holding securities under an employee benefit plan of the Company or a corporation owned directly or indirectly by the stockholders of the Company in substantially the same proportions as their ownership of stock of the Company, is or becomes the "beneficial owner" (as defined in Rule 13d-3 under said Act), directly or indirectly, of securities of the Company representing 20% or more of the total voting power represented by the Company's then outstanding Voting Securities, (ii) individuals who on the date of this Agreement are members of the Board (the "Incumbent Board") cease for any reason to constitute at least a majority of the members of the Board (provided, however, that if the appointment or election (or nomination for election) of any new Board member was approved or recommended by a majority vote of the members of the Incumbent Board then still in office, such new member shall be considered as a member of the Incumbent Board), or (iii) the stockholders of the Company approve a merger or consolidation of the Company with any other corporation, other than a merger or consolidation which would result in the Voting Securities of the Company outstanding immediately prior thereto continuing to represent (either by remaining outstanding or by being converted into Voting Securities of the surviving entity) at least 80% of the total voting power represented by the Voting Securities of the Company or such surviving entity outstanding immediately after such merger or consolidation, or the stockholders of the Company approve a plan of complete liquidation of the Company's assets.
- **(c) Expenses**. For purposes of this Agreement, the term "Expenses" shall be broadly construed and shall include, without limitation, all direct and indirect costs of any type or nature whatsoever (including, without limitation, all attorneys', witness, or other professional fees and related disbursements, and other out-of-pocket costs of whatever nature) actually and reasonably incurred by Indemnitee in connection with the investigation, defense or appeal of a proceeding or establishing or enforcing a right to indemnification under this Agreement, the Code or otherwise.
- **(d) Enterprise**. For purposes of this Agreement, the term "Enterprise" means any other corporation, limited liability company, partnership, joint venture, trust, employee benefit plan or other entity for which Indemnitee is or was serving at the request of the Company as a director, officer, employee, or Agent.
- **(e) Independent Counsel.** For purposes of this Agreement, the term "Independent Counsel" means a law firm, or a partner (or, if applicable, member) of such a law firm, that is experienced in matters of corporation law and neither presently is, nor in the past five (5) years has been, retained to represent: (i) the Company or Indemnitee in any matter material to either such party, or (ii) any other party to the proceeding giving rise to a claim for indemnification hereunder. Notwithstanding the foregoing, the term "Independent Counsel" shall not include any person who, under the applicable standards of professional conduct then prevailing, would have a conflict of interest in representing either the Company or Indemnitee in an action to determine Indemnitee's rights under this Agreement. The Company will pay the reasonable fees and expenses of the Independent Counsel referred to above and to fully indemnify such counsel against any and all expenses, claims, liabilities and damages arising out of or relating to this Agreement or its engagement pursuant hereto.

- **(f) Liabilities.** For purposes of this Agreement, the term "Liabilities" shall be broadly construed and shall include, without limitation, judgments, damages, deficiencies, liabilities, losses, penalties, excise taxes, fines, assessments and amounts paid in settlement, including any interest and any federal, state, local or foreign taxes imposed as a result of the actual or deemed receipt of any payment under this Agreement.
- **(g) Proceedings**. For purposes of this Agreement, the term "proceeding" shall be broadly construed and shall include, without limitation, any threatened, pending, or completed action, suit, claim, counterclaim, cross claim, arbitration, mediation, alternate dispute resolution mechanism, investigation, inquiry, administrative hearing, or any other actual, threatened or completed proceeding, whether brought in the right of the Company or otherwise and whether of a civil, criminal, administrative or investigative nature, and whether formal or informal in any case, in which Indemnitee was, is or will be involved as a party, potential party, non-party witness, or otherwise by reason of: (i) the fact that Indemnitee is or was a director or officer of the Company; (ii) the fact that any action taken by Indemnitee (or a failure to take action by Indemnitee) or of any action (or failure to act) on Indemnitee's part while acting as an Agent; or (iii) the fact that Indemnitee is or was serving at the request of the Company as a director, officer, employee or agent of another corporation, partnership, joint venture, trust, employee benefit plan, or other enterprise, and in any such case described above, whether or not serving in any such capacity at the time any liability or Expense is incurred for which indemnification, reimbursement, or advancement of Expenses may be provided under this Agreement. If the Indemnitee believes in good faith that a given situation may lead to or culminate in the institution of a proceeding, this shall be considered a proceeding under this paragraph.
- **(h) Subsidiary**. For purposes of this Agreement, the term "subsidiary" means any corporation, limited liability company, or other entity, of which more than 50% of the outstanding voting securities or equity interests are owned, directly or indirectly, by the Company and one or more of its subsidiaries, and any other corporation, limited liability company, partnership, joint venture, trust, employee benefit plan or other enterprise of which Indemnitee is or was serving at the request of the Company as an Agent.
- (i) Voting Securities. For purposes of this Agreement, "Voting Securities" shall mean any securities of the Company that vote generally in the election of directors.
- 2. Agreement to Serve. Indemnitee will serve, or continue to serve, as the case may be, as an Agent, faithfully and to the best of his or her ability, at the will of such entity designated by the Company and at the request of the Company (or under separate agreement, if such agreement exists), in the capacity Indemnitee currently serves such entity, so long as Indemnitee is duly appointed or elected and qualified in accordance with the applicable provisions of the governance documents of such entity, or until such time as Indemnitee tenders his or her resignation in writing; provided, however, that nothing contained in this Agreement is intended as an employment agreement between Indemnitee and the Company or any of its subsidiaries or to create any right to continued employment of Indemnitee with the Company or any of its subsidiaries in any capacity.

The Company acknowledges that it has entered into this Agreement and assumes the obligations imposed on it hereby, in addition to and separate from its obligations to Indemnitee under the Bylaws, to induce Indemnitee to serve, or continue to serve, as an Agent, and the Company acknowledges that Indemnitee is relying upon this Agreement in serving as an Agent.

#### 3. Indemnification.

- (a) Indemnification in Third Party Proceedings. Subject to Section 10 below, the Company shall indemnify Indemnitee to the fullest extent permitted by the Code, as the same may be amended from time to time (but, to the fullest extent of the law, only to the extent that such amendment permits Indemnitee to broader indemnification rights than the Code permitted prior to adoption of such amendment), if Indemnitee is a party to or threatened to be made a party to or otherwise involved in any proceeding, other than a proceeding by or in the right of the Company to procure a judgment in its favor, for any and all Expenses and Liabilities (including all interest, assessments and other charges paid or payable in connection with or in respect of such Expenses and Liabilities) incurred by Indemnitee in connection with the investigation, defense, settlement or appeal of such proceeding, if Indemnitee acted in good faith and in a manner Indemnitee reasonably believed to be in or not opposed to the best interests of the Company and, in the case of a criminal proceeding had no reasonable cause to believe that Indemnitee's conduct was unlawful. The parties hereto intend that this Agreement shall provide to the fullest extent permitted by law for indemnification in excess of that expressly permitted by statute, including, without limitation, any indemnification provided by the Certificate of Incorporation of the Company, the Bylaws, vote of its stockholders or disinterested directors, or applicable law.
- (b) Indemnification in Derivative Actions and Direct Actions by the Company. Subject to Section 10 below, the Company shall indemnify Indemnitee to the fullest extent permitted by the Code, as the same may be amended from time to time (but, fullest extent permitted by applicable law, only to the extent that such amendment permits Indemnitee to broader indemnification rights than the Code permitted prior to adoption of such amendment), if Indemnitee is a party to or threatened to be made a party to or otherwise involved in any proceeding by or in the right of the Company to procure a judgment in its favor, against any and all Expenses actually and reasonably incurred by Indemnitee in connection with the investigation, defense, settlement, or appeal of such proceedings, if Indemnitee acted in good faith and in a manner Indemnitee reasonably believed to be in or not opposed to the best interests of the Company. No indemnification for Expenses shall be made under this Section 3(b) in respect of any claim, issue or matter as to which Indemnitee shall have been finally adjudged by a court competent jurisdiction to be liable to the Company, unless and only to the extent that the Chancery Court of the State of Delaware or any court in which the proceeding was brought shall determine upon application that, despite the adjudication of liability but in view of all the circumstances of the case, Indemnitee is fairly and reasonably entitled to indemnification.
- 4. Indemnification of Expenses of Successful Party. To the fullest extent permitted by law, the Company shall indemnify Indemnitee against all Expenses in connection with a proceeding to the extent that Indemnitee has been successful on the merits or otherwise in defense of any proceeding or in defense of any claim, issue or matter therein, in whole or part, including the dismissal of any action without prejudice. If Indemnitee is not wholly successful in

such proceeding but is successful, on the merits or otherwise, as to one or more but less than all claims, issues or matters in such proceeding, the Company shall indemnify Indemnitee against all Expenses incurred by Indemnitee or on Indemnitee's behalf in connection with or related to each successfully resolved claim, issue or matter to the fullest extent permitted by law.

- 5. **Partial Indemnification; Witness Indemnification**. If Indemnitee is entitled under any provision of this Agreement to indemnification by the Company for some or a portion of any Expenses and Liabilities incurred by Indemnitee in the investigation, defense, settlement or appeal of a proceeding, but is precluded by applicable law or the specific terms of this Agreement to indemnification for the total amount thereof, the Company shall nevertheless indemnify Indemnitee for the portion thereof to which Indemnitee is entitled. Notwithstanding any other provision of this Agreement, to the fullest extent permitted by applicable law and to the extent that Indemnitee is, by reason of Indemnitee's acting as an Agent, a witness or otherwise asked to participate in any proceeding to which Indemnitee is not a party, Indemnitee shall be indemnified against all Expenses incurred by Indemnitee or on Indemnitee's behalf in connection therewith.
- 6. Advancement of Expenses. To the extent not prohibited by law, the Company shall advance the Expenses incurred by Indemnitee in connection with any proceeding, and such advancement shall be made within twenty (30) days after the receipt by the Company of a statement or statements requesting such advances (which shall include invoices received by Indemnitee in connection with such Expenses but, in the case of invoices in connection with legal services, any references to legal work performed or to expenditures made that would cause Indemnitee to waive any privilege accorded by applicable law shall not be included with the invoice). Advances shall be unsecured, interest free and without regard to Indemnitee's ability to repay the Expenses. Advances shall include any and all Expenses incurred by Indemnitee pursuing an action to enforce Indemnitee's right to indemnification under this Agreement or otherwise and this right of advancement, including expenses incurred preparing and forwarding statements to the Company to support the advances claimed. Indemnitee acknowledges that the execution and delivery of this Agreement shall constitute an undertaking providing that Indemnitee shall, to the fullest extent required by law, repay the advance (without interest) if and to the extent that it is ultimately determined by a court of competent jurisdiction in a final judgment, not subject to appeal, that Indemnitee is not entitled to be indemnified by the Company. The right to advances under this Section shall continue until final disposition of any proceeding, including any appeal therein. This Section 6 shall not apply to any claim made by Indemnitee for which indemnity is excluded pursuant to Section 10(b).

#### 7. Notice and Other Indemnification Procedures.

(a) Notification of Proceeding. Indemnitee will notify the Company in writing promptly upon being served with any summons, citation, subpoena, complaint, indictment, information or other document relating to any proceeding or matter which may be subject to indemnification or advancement of Expenses covered hereunder. The written notification to the Company shall include a description of the nature of the proceeding and the facts underlying the proceeding. The failure of Indemnitee to so notify the Company shall not relieve the Company of any obligation which it may have to Indemnitee under this Agreement or otherwise and any delay in so notifying the Company shall not constitute a waiver by Indemnitee of any rights under this Agreement. The Company will be entitled to participate in the proceeding at its own expense.

- **(b) Request for Indemnification Payments**. To obtain indemnification under this Agreement, Indemnitee shall submit to the Company a written request, including therein or therewith such documentation and information as is reasonably available to Indemnitee and is reasonably necessary to determine whether and to what extent Indemnitee is entitled to indemnification under the terms of this Agreement, and shall request payment thereof by the Company.
- (c) Determination of Right to Indemnification Payments. Upon written request by Indemnitee for indemnification pursuant to the Section 7(b) hereof, a determination with respect to Indemnitee's entitlement thereto shall be made in the specific case by one of the following four methods, which shall be at the election of the Board of Directors: (1) by a majority vote of the disinterested directors, even though less than a quorum, (2) by a committee of disinterested directors designated by a majority vote of the disinterested directors, even though less than a quorum, (3) if there are no disinterested directors or if the disinterested directors so direct, by Independent Counsel in a written opinion to the Board of Directors, a copy of which shall be delivered to the Indemnitee, or (4) if so directed by the Board of Directors, by the stockholders of the Company; provided, however, that if there has been a Change in Control, then such determination shall be made by Independent Counsel selected by Indemnitee and approved by the Company (which approval shall not be unreasonably withheld). For purposes hereof, disinterested directors are those members of the board of directors of the Company who are not parties to the action, suit or proceeding in respect of which indemnification is sought by Indemnitee. Indemnification payments requested by Indemnitee under Section 3 hereof shall be made by the Company within sixty (60) days after the later or (1) receipt of the written request of Indemnitee and (2) the final disposition of the Proceeding for which Indemnification is sought. Claims for advancement of Expenses shall be made under the provisions of Section 6 herein.
- (d) Application for Enforcement. In the event the Company fails to make timely payments as set forth in Sections 6 or 7(c) above, Indemnitee shall have the right to apply to the Chancery Court of the State of Delaware for the purpose of enforcing Indemnitee's right to indemnification or advancement of Expenses pursuant to this Agreement. In such an enforcement hearing or proceeding, the burden of proof shall be on the Company to prove that indemnification or advancement of Expenses to Indemnitee is not required under this Agreement or permitted by applicable law. Any determination by the Company (including its Board of Directors, a committee thereof, Independent Counsel) or stockholders, that Indemnitee is not entitled to indemnification hereunder, shall not be a defense by the Company to the action nor create any presumption that Indemnitee is not entitled to indemnification or advancement of Expenses hereunder.
- **(e) Indemnification of Certain Expenses**. The Company shall indemnify Indemnitee against all Expenses incurred in connection with any hearing or proceeding under this Section 7 unless the Company prevails in such hearing or proceeding on the merits in all material respects.

#### 8. Presumptions and Effect of Certain Proceedings.

- (a) In making a determination with respect to entitlement to indemnification hereunder, the person or persons or entity making such determination will, to the fullest extent not prohibited by law, presume Indemnitee is entitled to indemnification under this Agreement if Indemnitee has submitted a request for indemnification in accordance with Section 7 of this Agreement, and the Company will, to the fullest extent not prohibited by law, have the burden of proof to overcome that presumption. Neither the failure of the Company (including by its directors or Independent Counsel) to have made a determination prior to the commencement of any action pursuant to this Agreement that indemnification is proper in the circumstances because Indemnitee has met the applicable standard of conduct, nor an actual determination by the Company (including by its directors or Independent Counsel) that Indemnitee has not met such applicable standard of conduct, will be a defense to the action or create a presumption that Indemnitee has not met the applicable standard of conduct.
- **(b)** If the determination of the Indemnitee's entitlement to indemnification has not made pursuant to Section 7 within sixty (60) days after the later of (i) receipt by the Company of Indemnitee's request for indemnification pursuant to Section 7 and (ii) the final disposition of the Proceeding for which Indemnitee requested Indemnification (the "Determination Period"), the requisite determination of entitlement to indemnification will, to the fullest extent not prohibited by law, be deemed to have been made and Indemnitee will be entitled to such indemnification, absent (i) a misstatement by Indemnitee of a material fact, or an omission of a material fact necessary to make Indemnitee's statement not materially misleading, in connection with the request for indemnification, or (ii) a prohibition of such indemnification under applicable law. The Determination Period may be extended for a reasonable time, not to exceed an additional thirty (30) days, if the person, persons or entity making the determination with respect to entitlement to indemnification in good faith requires such additional time for the obtaining or evaluating of documentation and/or information relating thereto; and provided, further, the Determination Period may be extended an additional fifteen (15) days if the determination of entitlement to indemnification is to be made by the stockholders pursuant to Section 7(c) of this Agreement.
- (c) The termination of any Proceeding or of any claim, issue or matter therein, by judgment, order, settlement or conviction, or upon a plea of nolo contendere or its equivalent, will not (except as otherwise expressly provided in this Agreement) of itself adversely affect the right of Indemnitee to indemnification or create a presumption that Indemnitee did not act in good faith and in a manner which Indemnitee reasonably believed to be in or not opposed to the best interests of the Company or, with respect to any criminal Proceeding, that Indemnitee had reasonable cause to believe that Indemnitee's conduct was unlawful.
- (d) For purposes of any determination of good faith, Indemnitee will be deemed to have acted in good faith if Indemnitee acted based on the records or books of account of the Company, its subsidiaries, or an Enterprise, including financial statements, or on information supplied to Indemnitee by the directors or officers of the Company, its subsidiaries, or an Enterprise in the course of their duties, or on the advice of legal counsel for the Company, its subsidiaries, or an Enterprise or on information or records given or reports made to the Company or an Enterprise by an independent certified public accountant or by an appraiser,

financial advisor or other expert selected with reasonable care by or on behalf of the Company, its subsidiaries, or an Enterprise. Further, Indemnitee will be deemed to have acted in a manner "not opposed to the best interests of the Company," as referred to in this Agreement if Indemnitee acted in good faith and in a manner Indemnitee reasonably believed to be in the best interests of the participants and beneficiaries of an employee benefit plan. The provisions of this Section 8(d) is not exclusive and does not limit in any way the other circumstances in which the Indemnitee may be deemed to have met the applicable standard of conduct set forth in this Agreement.

- **(e)** The knowledge and/or actions, or failure to act, of any director, officer, trustee, partner, managing member, fiduciary, agent or employee of the Enterprise may not be imputed to Indemnitee for purposes of determining Indemnitee's right to indemnification under this Agreement.
- **9. Insurance**. To the extent that the Company maintains an insurance policy or policies providing liability insurance for Agents or for agents of any other Enterprise, Indemnitee shall be covered by such policy or policies in accordance with its or their terms to the maximum extent of the coverage available for any such Agent or agent under such policy or policies. If, at the time of the receipt of a notice of a claim pursuant to the terms hereof, the Company has director and officer liability insurance in effect or otherwise potentially available, the Company shall give prompt notice of the commencement of such proceeding to the insurers in accordance with the procedures set forth in the respective policies. The Company shall thereafter take all necessary or desirable action to cause such insurers to pay, on behalf of the Indemnitee, all amounts payable as a result of such proceeding in accordance with the terms of such policies.

## 10. Exceptions.

(a) Certain Matters. Any provision herein to the contrary notwithstanding, the Company shall not be obligated pursuant to the terms of this Agreement to indemnify Indemnitee on account of any proceeding with respect to: (i) remuneration paid to Indemnitee if it is determined by final judgment or other final adjudication that such remuneration was in violation of law (and, in this respect, both the Company and Indemnitee have been advised that the Securities and Exchange Commission believes that indemnification for liabilities arising under the federal securities laws is against public policy and is, therefore, unenforceable and that claims for indemnification should be submitted to appropriate courts for adjudication, as indicated in Section 10(d) below); (ii) a final judgment rendered against Indemnitee for an accounting, disgorgement or repayment of profits made from the purchase or sale by Indemnitee of securities of the Company against Indemnitee pursuant to the provisions of Section 16(b) of the Securities Exchange Act of 1934, as amended, or other provisions of any federal, state or local statute or rules and regulations thereunder; or (iii) a final judgment or other final adjudication that Indemnitee's conduct was in bad faith, knowingly fraudulent or deliberately dishonest or constituted willful misconduct (but only to the extent of such specific determination); or (iv) on account of conduct that is established by a final judgment as constituting a breach of Indemnitee's duty of loyalty to the Company or resulting in any personal profit or advantage to which Indemnitee is not legally entitled. For purposes of the foregoing sentence, a final judgment or other adjudication may be reached in either the underlying proceeding or action in connection with which indemnification is sought or a separate proceeding or action to establish rights and liabilities under this Agreement.

- **(b)** Claims Initiated by Indemnitee. Any provision herein to the contrary notwithstanding, the Company shall not be obligated to indemnify or advance Expenses to Indemnitee with respect to proceedings or claims initiated or brought by Indemnitee against the Company or its Agents and not by way of defense, except (i) with respect to proceedings brought to establish or enforce a right to indemnification or advancement under this Agreement or under any other agreement, provision in the Bylaws or Certificate of Incorporation or applicable law, or (ii) with respect to any other proceeding initiated by Indemnitee that is either approved by the Board of Directors or Indemnitee's participation is required by applicable law. However, indemnification or advancement of Expenses may be provided by the Company in specific cases if the Board of Directors determines it to be appropriate.
- (c) Unauthorized Settlements. Any provision herein to the contrary notwithstanding, the Company shall not be obligated pursuant to the terms of this Agreement to indemnify Indemnitee under this Agreement for any amounts paid in settlement of a proceeding effected without the Company's written consent. Neither the Company nor Indemnitee shall unreasonably withhold consent to any proposed settlement; provided, however, that the Company may in any event decline to consent to (or to otherwise admit or agree to any liability for indemnification hereunder in respect of) any proposed settlement if the Company is also a party in such proceeding and determines in good faith that such settlement is not in the best interests of the Company and its stockholders.
- (d) Securities Act Liabilities. Any provision herein to the contrary notwithstanding, the Company shall not be obligated pursuant to the terms of this Agreement to indemnify Indemnitee or otherwise act in violation of any undertaking appearing in and required by the rules and regulations promulgated under the Securities Act of 1933, as amended (the "Act"), or in any registration statement filed with the SEC under the Act. Indemnitee acknowledges that paragraph (h) of Item 512 of Regulation S-K currently generally requires the Company to undertake in connection with any registration statement filed under the Act to submit the issue of the enforceability of Indemnitee's rights under this Agreement in connection with any liability under the Act on public policy grounds to a court of appropriate jurisdiction and to be governed by any final adjudication of such issue. Indemnitee specifically agrees that any such undertaking shall supersede the provisions of this Agreement and to be bound by any such undertaking.
- **(e) Prior Payments.** The Company shall not be obligated pursuant to the terms of this Agreement to indemnify or advance Expenses to Indemnitee under this Agreement for which payment has actually been made to or on behalf of Indemnitee under any insurance policy or other indemnity provision, except to the extent made by [insert name] as provided in Section 13 and except with respect to any excess beyond the amount paid under any insurance policy or indemnity policy.
- 11. Nonexclusivity and Survival of Rights. The provisions for indemnification and advancement of Expenses set forth in this Agreement shall not be deemed exclusive of any other rights which Indemnitee may at any time be entitled under any provision of applicable law, the

Company's Certificate of Incorporation, Bylaws or other agreements, both as to action in Indemnitee's official capacity and Indemnitee's action as an Agent, in any court in which a proceeding is brought, and Indemnitee's rights hereunder shall continue after Indemnitee has ceased acting as an Agent and shall inure to the benefit of the heirs, executors, administrators and assigns of Indemnitee. The obligations and duties of the Company to Indemnitee under this Agreement shall be binding on the Company and its successors and assigns until terminated in accordance with its terms. The Company shall require any successor (whether direct or indirect, by purchase, merger, consolidation or otherwise) to all or substantially all of the business or assets of the Company, expressly to assume and agree to perform this Agreement in the same manner and to the same extent that the Company would be required to perform if no such succession had taken place.

No amendment, alteration or repeal of this Agreement or of any provision hereof shall limit or restrict any right of Indemnitee under this Agreement in respect of any action taken or omitted by such Indemnitee in his or her corporate status prior to such amendment, alteration or repeal. To the extent that a change in the Code, whether by statute or judicial decision, permits greater indemnification or advancement of Expenses than would be afforded currently under the Company's Certificate of Incorporation, Bylaws and this Agreement, it is the intent of the parties hereto that Indemnitee shall enjoy by this Agreement the greater benefits so afforded by such change. No right or remedy herein conferred is intended to be exclusive of any other right or remedy, and every other right and remedy shall be cumulative and in addition to every other right and remedy given hereunder or now or hereafter existing at law or in equity or otherwise. The assertion or employment of any right or remedy hereunder, or otherwise, by Indemnitee shall not prevent the concurrent assertion or employment of any other right or remedy by Indemnitee.

**12. Term**. This Agreement shall continue until and terminate upon the later of: (a) ten (10) years after the date that Indemnitee shall have ceased to serve as an Agent; or (b) one (1) year after the final termination of any proceeding, including any appeal then pending, in respect to which Indemnitee was granted rights of indemnification or advancement of Expenses hereunder.

#### 13. Other Rights to Indemnification or Advancement; Subrogation.

- (a) The Company hereby acknowledges that Indemnitee may have certain rights to indemnification, advancement of Expenses and/or insurance provided by one or more other Persons, other than an Enterprise, with whom or which Indemnitee may be associated (including, without limitation, [insert name]). The relationship between the Company and such other Persons with respect to the Indemnitee's rights to indemnification, advancement of Expenses, and insurance is described by this subsection, subject to the provisions of subsection (b) of this Section 13 with respect to a proceeding concerning Indemnitee's status with an Enterprise.
  - i. The Company hereby acknowledges and agrees:

- 1) the Company is the indemnitor of first resort with respect to any request for indemnification or advancement of Expenses made pursuant to this Agreement concerning any proceeding;
- 2) the Company is primarily liable for all indemnification and indemnification or advancement of Expenses obligations for any Proceeding, whether created by law, organizational or constituent documents, contract (including this Agreement) or otherwise;
- 3) any obligation of any other Persons with whom or which Indemnitee may be associated (including, without limitation, [insert name]) to indemnify Indemnitee and/or advance Expenses to Indemnitee in respect of any proceeding are secondary to the obligations of the Company's obligations:
- 4) the Company will indemnify Indemnitee and advance Expenses to Indemnitee hereunder to the fullest extent provided herein without regard to any rights Indemnitee may have against any other Person with whom or which Indemnitee may be associated (including, [insert name]) or insurer of any such Person; and
- ii. the Company irrevocably waives, relinquishes and releases (A) any other Person with whom or which Indemnitee may be associated (including, without limitation, [insert name]) from any claim of contribution, subrogation, reimbursement, exoneration or indemnification, or any other recovery of any kind in respect of amounts paid by the Company to Indemnitee pursuant to this Agreement and (B) any right to participate in any claim or remedy of Indemnitee against any Person (including, without limitation, [insert name]), whether or not such claim, remedy or right arises in equity or under contract, statute or common law, including, without limitation, the right to take or receive from any Person (including, without limitation, [insert name]), directly or indirectly, in cash or other property or by set-off or in any other manner, payment or security on account of such claim, remedy or right.
- iii. In the event any other Person with whom or which Indemnitee may be associated (including, without limitation, [insert name]) or their insurers advances or extinguishes any liability or loss for Indemnitee, the payor has a right of subrogation against the Company or its insurers for all amounts so paid which would otherwise be payable by the Company or its insurers under this Agreement. In no event will payment by any other Person with whom or which Indemnitee may be associated (including, without limitation, [insert name]) or their insurers affect the obligations of the Company hereunder or shift primary liability for the Company's obligation to indemnify or advance of Expenses to any other Person with whom or which Indemnitee may be associated (including, without limitation, [insert name]).
- iv. Any indemnification or advancement of Expenses provided by any other Person with whom or which Indemnitee may be associated (including, without limitation, [insert name]) is specifically in excess over the Company's obligation to indemnify and advance Expenses or any valid and collectible insurance (including but not limited to any malpractice insurance or professional errors and omissions insurance) provided by the Company.
- **(b)** The Company's obligation to indemnify or advance Expenses hereunder to Indemnitee for any proceeding concerning Indemnitee's status with an Enterprise will be

reduced by any amount Indemnitee has actually received as indemnification or advancement of Expenses from such Enterprise. The Company and Indemnitee intend that any such Enterprise (and its insurers) be the indemnitor of first resort with respect to indemnification and advancement of Expenses for any proceeding related to or arising from Indemnitee's status with such Enterprise. The Company's obligation to indemnify and advance Expenses to Indemnitee is secondary to the obligations the Enterprise or its insurers owe to Indemnitee. Indemnitee agrees to take all reasonably necessary and desirable action to obtain from an Enterprise indemnification and advancement of Expenses for any Proceeding related to or arising from Indemnitee's corporate status with such Enterprise.

- (c) In the event of payment under this Agreement, the Company shall be subrogated to the extent of such payment to all of the rights of recovery of Indemnitee from any insurance carrier or Enterprise. Indemnitee shall, at the request and expense of the Company, execute all papers required and shall do everything that may be reasonably necessary to secure such rights, including the execution of such documents necessary to enable the Company effectively to bring suit to enforce such rights.
- **14. Interpretation of Agreement**. It is understood that the parties hereto intend this Agreement to be interpreted and enforced so as to provide indemnification and advancement of Expenses to Indemnitee to the fullest extent now or hereafter permitted by law.
- 15. Severability. If any provision of this Agreement shall be held to be invalid, illegal or unenforceable for any reason whatsoever, (a) the validity, legality and enforceability of the remaining provisions of the Agreement (including without limitation, all portions of any paragraphs of this Agreement containing any such provision held to be invalid, illegal or unenforceable, that are not themselves invalid, illegal or unenforceable) shall not in any way be affected or impaired thereby; and (b) to the fullest extent possible, the provisions of this Agreement (including, without limitation, all portions of any paragraph of this Agreement containing any such provision held to be invalid, illegal or unenforceable, that are not themselves invalid, illegal or unenforceable) shall be construed so as to give effect to the intent manifested by the provision held invalid, illegal or unenforceable and to give effect to Section 14 hereof.
- **16. Amendment and Waiver**. No supplement, modification, amendment, or cancellation of this Agreement shall be binding unless executed in writing by the parties hereto. No waiver of any of the provisions of this Agreement shall be deemed or shall constitute a waiver of any other provision hereof (whether or not similar) nor shall such waiver constitute a continuing waiver.
- 17. Notice. Except as otherwise provided herein, any notice or demand which, by the provisions hereof, is required or which may be given to or served upon the parties hereto shall be in writing and, if by electronic transmission, shall be deemed to have been validly served, given or delivered when sent, if by overnight delivery, courier or personal delivery, shall be deemed to have been validly served, given or delivered upon actual delivery and, if mailed, shall be deemed to have been validly served, given or delivered three (3) business days after deposit in the United States mail, as registered or certified mail, with proper postage prepaid and addressed to the party or parties to be notified at the addresses set forth on the signature page of this Agreement (or such other address(es) as a party may designate for itself by like notice). If to the Company, notices and demands shall be delivered to the attention of the Secretary of the Company.

- **18. Governing Law**. This Agreement shall be governed exclusively by and construed according to the laws of the State of Delaware, as applied to contracts between Delaware residents entered into and to be performed entirely within Delaware.
- **19. Counterparts.** This Agreement may be executed in one or more counterparts, each of which shall for all purposes be deemed to be an original but all of which together shall constitute but one and the same Agreement. Only one such counterpart need be produced to evidence the existence of this Agreement.
- **20. Headings**. The headings of the sections of this Agreement are inserted for convenience only and shall not be deemed to constitute part of this Agreement or to affect the construction hereof.
- **21. Entire Agreement.** Subject to Section 11 hereof, this Agreement constitutes the entire agreement between the parties with respect to the subject matter hereof and supersedes all prior agreements, understandings and negotiations, written and oral, between the parties with respect to the subject matter of this Agreement; provided, however, that this Agreement is a supplement to and in furtherance of the Company's Certificate of Incorporation, Bylaws, the Code and any other applicable law, and shall not be deemed a substitute therefor, and does not diminish or abrogate any rights of Indemnitee thereunder.
- **22. Contribution**. To the fullest extent permissible under applicable law, if the indemnification provided for in this Agreement is unavailable to Indemnitee for any reason whatsoever, the Company, in lieu of indemnifying Indemnitee, shall contribute to the amount incurred by Indemnitee, whether for judgments, fines, penalties, excise taxes, amounts paid or to be paid in settlement and/or for Expenses, in connection with any claim relating to an indemnifiable event under this Agreement, in such proportion as is deemed fair and reasonable in light of all of the circumstances of such proceeding in order to reflect (i) the relative benefits received by the Company and Indemnitee as a result of the event(s) and/or transaction(s) giving cause to such proceeding; and/or (ii) the relative fault of the Company and Indemnitee in connection with such event(s) and/or transaction(s).
- 23. Consent to Jurisdiction. This Agreement and the legal relations among the parties shall be governed by, and construed and enforced in accordance with, the laws of the State of Delaware, without regard to its conflict of laws rules. The Company and Indemnitee hereby irrevocably and unconditionally (i) agree that any action or proceeding arising out of or in connection with this Agreement shall be brought only in the Chancery Court of the State of Delaware (the "Delaware Court"), and not in any other state or federal court in the United States of America or any court in any other country, (ii) consent to submit to the exclusive jurisdiction of the Delaware Court for purposes of any action or proceeding arising out of or in connection with this Agreement, (iii) agree to appoint, to the extent such party is not otherwise subject to service of process in the State of Delaware, an agent in the State of Delaware as such party's agent for acceptance of legal process in connection with any such action or proceeding against such party with the same legal force and validity as if served upon such party personally within

the State of Delaware, (iv) waive any objection to the laying of venue of any such action or proceeding in the Delaware Court, and (v) waive, and agree not to plead or to make, any claim that any such action or proceeding brought in the Delaware Court has been brought in an improper or inconvenient forum.

[Signature Page to Follow]

	ORIGIN MATERIALS, INC.
	By: [Name] [Title]
	INDEMNITEE
	Signature of Indemnitee
	Print or Type Name of Indemnitee
[Signature Page to Indemnity Agre	rement]

**IN WITNESS WHEREOF**, the parties hereto have entered into this Agreement effective as of the date first above written.

# ORIGIN MATERIALS, INC.

## NON-EMPLOYEE DIRECTOR COMPENSATION POLICY

Set forth below are the material terms of the Non-Employee Director Compensation Policy (the "*Policy*") of the Board of Directors (the "*Board*") of Origin Materials, Inc. (the "*Company*")

TERM		POLICY		
1.	Effective Date	• The Policy is effective as of the closing of the merger between Micromidas, Inc., a Delaware corporation doing business as Origin Materials, Inc. and Zero Carbon Merger Sub Inc., a Delaware corporation and a whollyowned subsidiary of Artius Acquisition Inc. (the "Closing")		
2.	Eligible Directors	• Directors who are not (i) employees or consultants to the Company or (ii) affiliated with Artius Acquisition Partners LLC		
		<ul> <li>Any director may elect to waive receipt of his or her compensation</li> </ul>		
3.	Cash Compensation	Annual cash retainer: \$50,000		
		Additional cash retainers:		
		<ul> <li>Service on committee: \$10,000 Audit Committee, \$5,000 Compensation Committee and \$5,000 Nominating &amp; Corporate Governance Committee</li> </ul>		
		<ul> <li>Service as chair of a committee: \$20,000 Audit Committee, \$15,000 Compensation Committee and \$15,000 Nominating &amp; Corporate Governance Committee</li> </ul>		
		<ul> <li>Cash retainers shall be paid quarterly in arrears in four equal installments, subject to continuing service on the Board, any committee, and any committee chair, as applicable, during each such quarter</li> </ul>		
4.	<b>Equity Compensation</b>	All equity awards will be made pursuant to the Company's 2021 Equity Incentive Plan		
		<ul> <li>All eligible directors - initial grant of restricted stock unit award: \$150,000 (grant date fair market value) (the "Initial Grant")</li> </ul>		
		<ul> <li>All eligible directors - annual grant of restricted stock unit award: \$130,000 (grant date fair market value) (the "Annual Grant")</li> </ul>		
		• Chairperson - additional annual grant of restricted stock unit award: \$75,000 (grant date fair market value)		
5.	Equity Compensation Terms	• The Initial Grant shall vest in three equal installments on the first, second and third anniversary of the grant date, subject to continuing service on the Board through the applicable vesting date		
		• The Annual Grant shall vest on the first anniversary of the grant date, subject to continuing service on the Board through the vesting date		
		• Each director shall be required to acquire and hold shares of common stock of the Company with a fair market value of at least \$250,000 by the later of the fifth anniversary of (i) the Closing, and (ii) such director's election to the Board		
		• The Initial Grant and first Annual Grant shall be made as soon as reasonably practicable following the Closing and the filing of a registration statement on Form S-8		

#### ORIGIN MATERIALS, INC. 2021 EQUITY INCENTIVE PLAN

#### ADOPTED BY THE BOARD OF DIRECTORS: MAY 25, 2021 APPROVED BY THE STOCKHOLDERS: JUNE 23, 2021

#### 1. GENERAL.

- **(a) Plan Purpose.** The Company, by means of the Plan, seeks to secure and retain the services of Employees, Directors and Consultants, to provide incentives for such persons to exert maximum efforts for the success of the Company and any Affiliate and to provide a means by which such persons may be given an opportunity to benefit from increases in value of the Common Stock through the granting of Awards.
- **(b) Available Awards.** The Plan provides for the grant of the following Awards: (i) Incentive Stock Options; (ii) Nonstatutory Stock Options; (iii) SARs; (iv) Restricted Stock Awards; (v) RSU Awards; (vi) Performance Awards; and (vii) Other Awards.
- **(c) Adoption Date; Effective Date.** The Plan will come into existence on the Adoption Date, but no Award may be granted prior to the Effective Date.

#### 2. SHARES SUBJECT TO THE PLAN.

- (a) Share Reserve. Subject to adjustment in accordance with Section 2(c) and any adjustments as necessary to implement any Capitalization Adjustments, the aggregate number of shares of Common Stock that may be issued pursuant to Awards will not exceed 18,467,109 shares. In addition, subject to any adjustments as necessary to implement any Capitalization Adjustments, such aggregate number of shares of Common Stock will automatically increase on January 1 of each year for a period of ten years commencing on January 1, 2022 and ending on (and including) January 1, 2031, in an amount equal to five percent (5%) of the Fully-Diluted Common Stock on December 31 of the preceding year (the "Evergreen Measurement Date"); provided, however, that (i) the Board may act prior to January 1st of a given year to provide that the increase for such year will be a lesser number of shares of Common Stock and (ii) the number of shares added to the Share Reserve on January 1st of a given year shall be reduced automatically to the extent necessary to avoid causing the Share Reserve to exceed a number of shares equal to fifteen percent (15%) of the Fully-Diluted Common Stock on the applicable Evergreen Measurement Date.
- **(b)** Aggregate Incentive Stock Option Limit. Notwithstanding anything to the contrary in Section 2(a) and subject to any adjustments as necessary to implement any Capitalization Adjustments, the aggregate maximum number of shares of Common Stock that may be issued pursuant to the exercise of Incentive Stock Options is 55,401,327 shares.

## (c) Share Reserve Operation.

(i) Limit Applies to Common Stock Issued Pursuant to Awards. For clarity, the Share Reserve is a limit on the number of shares of Common Stock that may be issued pursuant to Awards and does not limit the granting of Awards, except that the Company will keep available at all times the number of shares of Common Stock reasonably required to satisfy its obligations to issue shares pursuant to such Awards. Shares may be issued in connection with a merger or acquisition as permitted by, as applicable, Nasdaq Listing Rule 5635(c), NYSE Listed Company Manual Section 303A.08, NYSE American Company Guide Section 711 or other applicable rule, and such issuance will not reduce the number of shares available for issuance under the Plan.

- (ii) Actions that Do Not Constitute Issuance of Common Stock and Do Not Reduce Share Reserve. The following actions do not result in an issuance of shares under the Plan and accordingly do not reduce the number of shares subject to the Share Reserve and available for issuance under the Plan: (1) the expiration or termination of any portion of an Award without the shares covered by such portion of the Award having been issued, (2) the settlement of any portion of an Award in cash (i.e., the Participant receives cash rather than Common Stock), (3) the withholding of shares that would otherwise be issued by the Company to satisfy the exercise, strike or purchase price of an Award; or (4) the withholding of shares that would otherwise be issued by the Company to satisfy a tax withholding obligation in connection with an Award.
- (iii) Reversion of Previously Issued Shares of Common Stock to Share Reserve. The following shares of Common Stock previously issued pursuant to an Award and accordingly initially deducted from the Share Reserve will be added back to the Share Reserve and again become available for issuance under the Plan: (1) any shares that are forfeited back to or repurchased by the Company because of a failure to meet a contingency or condition required for the vesting of such shares; (2) any shares that are reacquired by the Company to satisfy the exercise, strike or purchase price of an Award; and (3) any shares that are reacquired by the Company to satisfy a tax withholding obligation in connection with an Award.

## 3. ELIGIBILITY AND LIMITATIONS.

- (a) Eligible Award Recipients. Subject to the terms of the Plan, Employees, Directors and Consultants are eligible to receive Awards.
- (b) Specific Award Limitations.
- (i) **Limitations on Incentive Stock Option Recipients.** Incentive Stock Options may be granted only to Employees of the Company or a "parent corporation" or "subsidiary corporation" thereof (as such terms are defined in Sections 424(e) and (f) of the Code).
- (ii) Incentive Stock Option \$100,000 Limitation. To the extent that the aggregate Fair Market Value (determined at the time of grant) of Common Stock with respect to which Incentive Stock Options are exercisable for the first time by any Optionholder during any calendar year (under all plans of the Company and any Affiliates) exceeds \$100,000 (or such other limit established in the Code) or otherwise does not comply with the rules governing Incentive Stock Options, the Options or portions thereof that exceed such limit (according to the order in which they were granted) or otherwise do not comply with such rules will be treated as Nonstatutory Stock Options, notwithstanding any contrary provision of the applicable Option Agreement(s).
- (iii) Limitations on Incentive Stock Options Granted to Ten Percent Stockholders. A Ten Percent Stockholder may not be granted an Incentive Stock Option unless (i) the exercise price of such Option is at least 110% of the Fair Market Value on the date of grant of such Option and (ii) the Option is not exercisable after the expiration of five years from the date of grant of such Option.
- (iv) Limitations on Nonstatutory Stock Options and SARs. Nonstatutory Stock Options and SARs may not be granted to Employees, Directors and Consultants unless the stock underlying such Awards is treated as "service recipient stock" under Section 409A or unless such Awards otherwise comply with the requirements of Section 409A.

- **(c) Aggregate Incentive Stock Option Limit.** The aggregate maximum number of shares of Common Stock that may be issued pursuant to the exercise of Incentive Stock Options is the number of shares specified in Section 2(b).
- (d) Non-Employee Director Compensation Limit. The aggregate value of all compensation granted (with respect to equity-based awards, measured based on grant date value) or paid (with respect to cash-based awards), as applicable, to any individual for service as a Non-Employee Director with respect to any calendar year, including Awards granted and cash fees paid by the Company to such Non-Employee Director, will not exceed (i) \$750,000 in total value or (ii) in the event such Non-Employee Director is first appointed or elected to the Board during such calendar year, \$1,000,000 in total value, in each case, calculating the value of any equity awards based on the grant date fair value of such equity awards for financial reporting purposes. The limitations in this Section 3(d) shall apply commencing with the first calendar year that begins following the Effective Date.

## 4. OPTIONS AND STOCK APPRECIATION RIGHTS.

Each Option and SAR will have such terms and conditions as determined by the Board. Each Option will be designated in writing as an Incentive Stock Option or Nonstatutory Stock Option at the time of grant; provided, however, that if an Option is not so designated or if an Option designated as an Incentive Stock Option fails to qualify as an Incentive Stock Option, then such Option will be a Nonstatutory Stock Option, and the shares purchased upon exercise of each type of Option will be separately accounted for. Each SAR will be denominated in shares of Common Stock equivalents. The terms and conditions of separate Options and SARs need not be identical; provided, however, that each Option Agreement and SAR Agreement will conform (through incorporation of provisions hereof by reference in the Award Agreement or otherwise) to the substance of each of the following provisions:

- **(a) Term.** Subject to Section 3(b) regarding Ten Percent Stockholders, no Option or SAR will be exercisable after the expiration of ten years from the date of grant of such Award or such shorter period specified in the Award Agreement.
- **(b)** Exercise or Strike Price. Subject to Section 3(b) regarding Ten Percent Stockholders, the exercise or strike price of each Option or SAR will not be less than 100% of the Fair Market Value on the date of grant of such Award. Notwithstanding the foregoing, an Option or SAR may be granted with an exercise or strike price lower than 100% of the Fair Market Value on the date of grant of such Award if such Award is granted pursuant to an assumption of or substitution for another option or stock appreciation right pursuant to a Corporate Transaction and in a manner consistent with the provisions of Sections 409A and, if applicable, 424(a) of the Code.
- **(c)** Exercise Procedure and Payment of Exercise Price for Options. In order to exercise an Option, the Participant must provide notice of exercise to the Plan Administrator in accordance with the procedures specified in the Option Agreement or otherwise provided by the Company. The Board has the authority to grant Options that do not permit all of the following methods of payment (or otherwise restrict the ability to use certain methods) and to grant Options that require the consent of the Company to utilize a particular method of payment. The exercise price of an Option may be paid, to the extent permitted by Applicable Law and as determined by the Board, by one or more of the following methods of payment to the extent set forth in the Option Agreement:
  - (i) by cash or check, bank draft or money order payable to the Company;

- (ii) pursuant to a "cashless exercise" program developed under Regulation T as promulgated by the Federal Reserve Board that, prior to the issuance of the Common Stock subject to the Option, results in either the receipt of cash (or check) by the Company or the receipt of irrevocable instructions to pay the exercise price to the Company from the sales proceeds;
- (iii) by delivery to the Company (either by actual delivery or attestation) of shares of Common Stock that are already owned by the Participant free and clear of any liens, claims, encumbrances or security interests, with a Fair Market Value on the date of exercise that does not exceed the exercise price, provided that (1) at the time of exercise the Common Stock is publicly traded, (2) any remaining balance of the exercise price not satisfied by such delivery is paid by the Participant in cash or other permitted form of payment, (3) such delivery would not violate any Applicable Law or agreement restricting the redemption of the Common Stock, (4) any certificated shares are endorsed or accompanied by an executed assignment separate from certificate, and (5) such shares have been held by the Participant for any minimum period necessary to avoid adverse accounting treatment as a result of such delivery;
- (iv) if the Option is a Nonstatutory Stock Option, by a "net exercise" arrangement pursuant to which the Company will reduce the number of shares of Common Stock issuable upon exercise by the largest whole number of shares with a Fair Market Value on the date of exercise that does not exceed the exercise price, provided that (1) such shares used to pay the exercise price will not be exercisable thereafter and (2) any remaining balance of the exercise price not satisfied by such net exercise is paid by the Participant in cash or other permitted form of payment; or
  - (v) in any other form of consideration that may be acceptable to the Board and permissible under Applicable Law.
- (d) Exercise Procedure and Payment of Appreciation Distribution for SARs. In order to exercise any SAR, the Participant must provide notice of exercise to the Plan Administrator in accordance with the SAR Agreement. The appreciation distribution payable to a Participant upon the exercise of a SAR will not be greater than an amount equal to the excess of (i) the aggregate Fair Market Value on the date of exercise of a number of shares of Common Stock equal to the number of Common Stock equivalents that are vested and being exercised under such SAR, over (ii) the strike price of such SAR. Such appreciation distribution may be paid to the Participant in the form of Common Stock or cash (or any combination of Common Stock and cash) or in any other form of payment, as determined by the Board and specified in the SAR Agreement.
- **(e) Transferability.** Options and SARs may not be transferred to third party financial institutions for value. The Board may impose such additional limitations on the transferability of an Option or SAR as it determines. In the absence of any such determination by the Board, the following restrictions on the transferability of Options and SARs will apply, provided that except as explicitly provided herein, neither an Option nor a SAR may be transferred for consideration and provided, further, that if an Option is an Incentive Stock Option, such Option may be deemed to be a Nonstatutory Stock Option as a result of such transfer:
- (i) Restrictions on Transfer. An Option or SAR will not be transferable, except by will or by the laws of descent and distribution, and will be exercisable during the lifetime of the Participant only by the Participant; provided, however, that the Board may permit transfer of an Option or SAR in a manner that is not prohibited by applicable tax and securities laws upon the Participant's request, including to a trust if the Participant is considered to be the sole beneficial owner of such trust (as determined under Section 671 of the Code and applicable state law) while such Option or SAR is held in such trust, provided that the Participant and the trustee enter into a transfer and other agreements required by the Company.

- (ii) **Domestic Relations Orders.** Notwithstanding the foregoing, subject to the execution of transfer documentation in a format acceptable to the Company and subject to the approval of the Board or a duly authorized Officer, an Option or SAR may be transferred pursuant to a domestic relations order.
- **(f) Vesting.** The Board may impose such restrictions on or conditions to the vesting and/or exercisability of an Option or SAR as determined by the Board. Except as otherwise provided in the Award Agreement or other written agreement between a Participant and the Company or an Affiliate, vesting of Options and SARs will cease upon termination of the Participant's Continuous Service.
- **(g) Termination of Continuous Service for Cause.** Except as explicitly otherwise provided in the Award Agreement or other written agreement between a Participant and the Company or an Affiliate, if a Participant's Continuous Service is terminated for Cause, the Participant's Options and SARs will terminate and be forfeited immediately upon such termination of Continuous Service, and the Participant will be prohibited from exercising any portion (including any vested portion) of such Awards on and after the date of such termination of Continuous Service and the Participant will have no further right, title or interest in such forfeited Award, the shares of Common Stock subject to the forfeited Award, or any consideration in respect of the forfeited Award.
- (h) Post-Termination Exercise Period Following Termination of Continuous Service for Reasons Other than Cause. Subject to Section 4(i), if a Participant's Continuous Service terminates for any reason other than for Cause, the Participant may exercise his or her Option or SAR to the extent vested, but only within the following period of time or, if applicable, such other period of time provided in the Award Agreement or other written agreement between a Participant and the Company or an Affiliate; provided, however, that in no event may such Award be exercised after the expiration of its maximum term (as set forth in Section 4(a)):
- (i) three months following the date of such termination if such termination is a termination without Cause (other than any termination due to the Participant's Disability or death);
  - (ii) 12 months following the date of such termination if such termination is due to the Participant's Disability;
  - (iii) 18 months following the date of such termination if such termination is due to the Participant's death; or
- (iv) 18 months following the date of the Participant's death if such death occurs following the date of such termination but during the period such Award is otherwise exercisable (as provided in (i) or (ii) above).

Following the date of such termination, to the extent the Participant does not exercise such Award within the applicable Post-Termination Exercise Period (or, if earlier, prior to the expiration of the maximum term of such Award), such unexercised portion of the Award will terminate, and the Participant will have no further right, title or interest in the terminated Award, the shares of Common Stock subject to the terminated Award, or any consideration in respect of the terminated Award.

(i) Restrictions on Exercise; Extension of Exercisability. A Participant may not exercise an Option or SAR at any time that the issuance of shares of Common Stock upon such exercise would

violate Applicable Law. Except as otherwise provided in the Award Agreement or other written agreement between a Participant and the Company or an Affiliate, if a Participant's Continuous Service terminates for any reason other than for Cause and, at any time during the last thirty days of the applicable Post-Termination Exercise Period: (i) the exercise of the Participant's Option or SAR would be prohibited solely because the issuance of shares of Common Stock upon such exercise would violate Applicable Law, or (ii) the immediate sale of any shares of Common Stock issued upon such exercise would violate the Company's Trading Policy, then the applicable Post-Termination Exercise Period will be extended to the last day of the calendar month that commences following the date the Award would otherwise expire, with an additional extension of the exercise period to the last day of the next calendar month to apply if any of the foregoing restrictions apply at any time during such extended exercise period, generally without limitation as to the maximum permitted number of extensions); provided, however, that in no event may such Award be exercised after the expiration of its maximum term (as set forth in Section 4(a)).

- (j) Non-Exempt Employees. No Option or SAR, whether or not vested, granted to an Employee who is a non-exempt employee for purposes of the Fair Labor Standards Act of 1938, as amended, will be first exercisable for any shares of Common Stock until at least six months following the date of grant of such Award. Notwithstanding the foregoing, in accordance with the provisions of the Worker Economic Opportunity Act, any vested portion of such Award may be exercised earlier than six months following the date of grant of such Award in the event of (i) such Participant's death or Disability, (ii) a Corporate Transaction in which such Award is not assumed, continued or substituted, (iii) a Change in Control, or (iv) such Participant's retirement (as such term may be defined in the Award Agreement or another applicable agreement or, in the absence of any such definition, in accordance with the Company's then current employment policies and guidelines). This Section 4(j) is intended to operate so that any income derived by a non-exempt employee in connection with the exercise or vesting of an Option or SAR will be exempt from his or her regular rate of pay.
  - (k) Whole Shares. Options and SARs may be exercised only with respect to whole shares of Common Stock or their equivalents.

#### 5. AWARDS OTHER THAN OPTIONS AND STOCK APPRECIATION RIGHTS.

**(a) Restricted Stock Awards and RSU Awards.** Each Restricted Stock Award and RSU Award will have such terms and conditions as determined by the Board; provided, however, that each Restricted Stock Award Agreement and RSU Award Agreement will conform (through incorporation of the provisions hereof by reference in the Award Agreement or otherwise) to the substance of each of the following provisions:

## (i) Form of Award.

- (1) Restricted Stock Awards: To the extent consistent with the Company's Bylaws, at the Board's election, shares of Common Stock subject to a Restricted Stock Award may be (i) held in book entry form subject to the Company's instructions until such shares become vested or any other restrictions lapse, or (ii) evidenced by a certificate, which certificate will be held in such form and manner as determined by the Board. Unless otherwise determined by the Board, a Participant will have voting and other rights as a stockholder of the Company with respect to any shares subject to a Restricted Stock Award.
- (2) RSU Awards: An RSU Award represents a Participant's right to be issued on a future date the number of shares of Common Stock that is equal to the number of restricted stock units subject to the RSU Award. As a holder of an RSU Award, a Participant is an unsecured

creditor of the Company with respect to the Company's unfunded obligation, if any, to issue shares of Common Stock in settlement of such Award and nothing contained in the Plan or any RSU Agreement, and no action taken pursuant to its provisions, will create or be construed to create a trust of any kind or a fiduciary relationship between a Participant and the Company or an Affiliate or any other person. A Participant will not have voting or any other rights as a stockholder of the Company with respect to any RSU Award (unless and until shares are actually issued in settlement of a vested RSU Award).

## (ii) Consideration.

- (1) Restricted Stock Awards: A Restricted Stock Award may be granted in consideration for (A) cash or check, bank draft or money order payable to the Company, (B) services to the Company or an Affiliate, or (C) any other form of consideration as the Board may determine and permissible under Applicable Law.
- (2) RSU Awards: Unless otherwise determined by the Board at the time of grant, an RSU Award will be granted in consideration for the Participant's services to the Company or an Affiliate, such that the Participant will not be required to make any payment to the Company (other than such services) with respect to the grant or vesting of the RSU Award, or the issuance of any shares of Common Stock pursuant to the RSU Award. If, at the time of grant, the Board determines that any consideration must be paid by the Participant (in a form other than the Participant's services to the Company or an Affiliate) upon the issuance of any shares of Common Stock in settlement of the RSU Award, such consideration may be paid in any form of consideration as the Board may determine and permissible under Applicable Law.
- (iii) **Vesting.** The Board may impose such restrictions on or conditions to the vesting of a Restricted Stock Award or RSU Award as determined by the Board. Except as otherwise provided in the Award Agreement or other written agreement between a Participant and the Company or an Affiliate, vesting of Restricted Stock Awards and RSU Awards will cease upon termination of the Participant's Continuous Service.
- (iv) Termination of Continuous Service. Except as otherwise provided in the Award Agreement or other written agreement between a Participant and the Company or an Affiliate, if a Participant's Continuous Service terminates for any reason, (i) the Company may receive through a forfeiture condition or a repurchase right any or all of the shares of Common Stock held by the Participant under his or her Restricted Stock Award that have not vested as of the date of such termination as set forth in the Restricted Stock Award Agreement and the Participant will have no further right, title or interest in the Restricted Stock Award, the shares of Common Stock subject to the Restricted Stock Award, or any consideration in respect of the Restricted Stock Award and (ii) any portion of his or her RSU Award that has not vested will be forfeited upon such termination and the Participant will have no further right, title or interest in the RSU Award, the shares of Common Stock issuable pursuant to the RSU Award, or any consideration in respect of the RSU Award.
- (v) Dividends and Dividend Equivalents. Dividends or dividend equivalents may be paid or credited, as applicable, with respect to any shares of Common Stock subject to a Restricted Stock Award or RSU Award, as determined by the Board and specified in the Award Agreement.
- (vi) Settlement of RSU Awards. An RSU Award may be settled by the issuance of shares of Common Stock or cash (or any combination thereof) or in any other form of payment, as determined by the Board and specified in the RSU Award Agreement. At the time of grant, the Board may determine to impose such restrictions or conditions that delay such delivery to a date following the vesting of the RSU Award.

- **(b) Performance Awards.** With respect to any Performance Award, the length of any Performance Period, the Performance Goals to be achieved during the Performance Period, the other terms and conditions of such Award, and the measure of whether and to what degree such Performance Goals have been attained will be determined by the Board.
- (c) Other Awards. Other forms of Awards valued in whole or in part by reference to, or otherwise based on, Common Stock, including the appreciation in value thereof, may be granted either alone or in addition to Awards provided for under Section 4 and the preceding provisions of this Section 5. Subject to the provisions of the Plan, the Board will have sole and complete discretion to determine the persons to whom and the time or times at which such Other Awards will be granted, the number of shares of Common Stock (or the cash equivalent thereof) to be granted pursuant to such Other Awards and all other terms and conditions of such Other Awards.

## 6. ADJUSTMENTS UPON CHANGES IN COMMON STOCK; OTHER CORPORATE EVENTS.

- (a) Capitalization Adjustments. In the event of a Capitalization Adjustment, the Board shall appropriately and proportionately adjust: (i) the class(es) and maximum number of shares of Common Stock subject to the Plan and the maximum number of shares by which the Share Reserve may annually increase pursuant to Section 2(a), (ii) the class(es) and maximum number of shares that may be issued pursuant to the exercise of Incentive Stock Options pursuant to Section 2(b), and (iii) the class(es) and number of securities and exercise price, strike price or purchase price of Common Stock subject to outstanding Awards. The Board shall make such adjustments, and its determination shall be final, binding and conclusive.

  Notwithstanding the foregoing, no fractional shares or rights for fractional shares of Common Stock shall be created in order to implement any Capitalization Adjustment. The Board shall determine an appropriate equivalent benefit, if any, for any fractional shares or rights to fractional shares that might be created by the adjustments referred to in the preceding provisions of this Section.
- **(b) Dissolution or Liquidation**. Except as otherwise provided in the Award Agreement, in the event of a dissolution or liquidation of the Company, all outstanding Awards (other than Awards consisting of vested and outstanding shares of Common Stock not subject to a forfeiture condition or the Company's right of repurchase) will terminate immediately prior to the completion of such dissolution or liquidation, and the shares of Common Stock subject to the Company's repurchase rights or subject to a forfeiture condition may be repurchased or reacquired by the Company notwithstanding the fact that the holder of such Award is providing Continuous Service; provided, however, that the Board may determine to cause some or all Awards to become fully vested, exercisable and/or no longer subject to repurchase or forfeiture (to the extent such Awards have not previously expired or terminated) before the dissolution or liquidation is completed but contingent on its completion.
- **(c) Corporate Transaction.** The following provisions will apply to Awards in the event of a Corporate Transaction unless otherwise provided in the instrument evidencing the Award or any other written agreement between the Company or any Affiliate and the Participant or unless otherwise expressly provided by the Board at the time of grant of an Award.
- (i) Awards May Be Assumed. In the event of a Corporate Transaction, any surviving corporation or acquiring corporation (or the surviving or acquiring corporation's parent company) may assume or continue any or all Awards outstanding under the Plan or may substitute similar awards for Awards outstanding under the Plan (including but not limited to, awards to acquire the same consideration paid to the stockholders of the Company pursuant to the Corporate Transaction), and any reacquisition or repurchase rights held by the Company in respect of Common Stock issued pursuant to Awards may be assigned by the Company to the successor of the Company (or the successor's parent company, if any), in connection with such Corporate Transaction. A surviving corporation or acquiring

corporation (or its parent) may choose to assume or continue only a portion of an Award or substitute a similar award for only a portion of an Award, or may choose to assume or continue the Awards held by some, but not all Participants. The terms of any assumption, continuation or substitution will be set by the Board.

- (ii) Awards Held by Current Participants. In the event of a Corporate Transaction in which the surviving corporation or acquiring corporation (or its parent company) does not assume or continue such outstanding Awards or substitute similar awards for such outstanding Awards, then with respect to Awards that have not been assumed, continued or substituted and that are held by Participants whose Continuous Service has not terminated prior to the effective time of the Corporate Transaction (referred to as the "Current Participants"), the vesting of such Awards (and, with respect to Options and Stock Appreciation Rights, the time when such Awards may be exercised) will be accelerated in full to a date prior to the effective time of such Corporate Transaction (contingent upon the effectiveness of the Corporate Transaction) as the Board determines (or, if the Board does not determine such a date, to the date that is five days prior to the effective time of the Corporate Transaction), and such Awards will terminate if not exercised (if applicable) at or prior to the effective time of the Corporate Transaction, and any reacquisition or repurchase rights held by the Company with respect to such Awards will lapse (contingent upon the effectiveness of the Corporate Transaction). With respect to the vesting of Performance Awards that will accelerate upon the occurrence of a Corporate Transaction pursuant to this subsection (ii) and that have multiple vesting levels depending on the level of performance, unless otherwise provided in the Award Agreement, the vesting of such Performance Awards will accelerate at 100% of the target level upon the occurrence of the Corporate Transaction in which the Awards are not assumed in accordance with Section 6(c)(i). With respect to the vesting of Awards that will accelerate upon the occurrence of a Corporate Transaction pursuant to this subsection (ii) and are settled in the form of a cash payment, such cash payment will be made no later than 30 days following the occurrence of the Corp
- (iii) Awards Held by Persons other than Current Participants. In the event of a Corporate Transaction in which the surviving corporation or acquiring corporation (or its parent company) does not assume or continue such outstanding Awards or substitute similar awards for such outstanding Awards, then with respect to Awards that have not been assumed, continued or substituted and that are held by persons other than Current Participants, such Awards will terminate if not exercised (if applicable) prior to the occurrence of the Corporate Transaction; provided, however, that any reacquisition or repurchase rights held by the Company with respect to such Awards will not terminate and may continue to be exercised notwithstanding the Corporate Transaction.
- **(iv) Payment for Awards in Lieu of Exercise.** Notwithstanding the foregoing, in the event an Award will terminate if not exercised prior to the effective time of a Corporate Transaction, the Board may provide, in its sole discretion, that the holder of such Award may not exercise such Award but will receive a payment, in such form as may be determined by the Board, equal in value, at the effective time, to the excess, if any, of (1) the value of the property the Participant would have received upon the exercise of the Award (including, at the discretion of the Board, any unvested portion of such Award), over (2) any exercise price payable by such holder in connection with such exercise.
- **(d)** Appointment of Stockholder Representative. As a condition to the receipt of an Award under this Plan, a Participant will be deemed to have agreed that the Award will be subject to the terms of any agreement governing a Corporate Transaction involving the Company, including, without limitation, a provision for the appointment of a stockholder representative that is authorized to act on the Participant's behalf with respect to any escrow, indemnities and any contingent consideration.

(e) No Restriction on Right to Undertake Transactions. The grant of any Award under the Plan and the issuance of shares pursuant to any Award does not affect or restrict in any way the right or power of the Company or the stockholders of the Company to make or authorize any adjustment, recapitalization, reorganization or other change in the Company's capital structure or its business, any merger or consolidation of the Company, any issue of stock or of options, rights or options to purchase stock or of bonds, debentures, preferred or prior preference stocks whose rights are superior to or affect the Common Stock or the rights thereof or which are convertible into or exchangeable for Common Stock, or the dissolution or liquidation of the Company, or any sale or transfer of all or any part of its assets or business, or any other corporate act or proceeding, whether of a similar character or otherwise.

## 7. ADMINISTRATION.

- (a) Administration by Board. The Board will administer the Plan unless and until the Board delegates administration of the Plan to a Committee or Committees, as provided in subsection (c) below.
  - **(b) Powers of Board.** The Board will have the power, subject to, and within the limitations of, the express provisions of the Plan:
- (i) To determine from time to time: (1) which of the persons eligible under the Plan will be granted Awards; (2) when and how each Award will be granted; (3) what type or combination of types of Award will be granted; (4) the provisions of each Award granted (which need not be identical), including the time or times when a person will be permitted to receive an issuance of Common Stock or other payment pursuant to an Award; (5) the number of shares of Common Stock or cash equivalent with respect to which an Award will be granted to each such person; (6) the Fair Market Value applicable to an Award; and (7) the terms of any Performance Award that is not valued in whole or in part by reference to, or otherwise based on, the Common Stock, including the amount of cash payment or other property that may be earned and the timing of payment.
- (ii) To construe and interpret the Plan and Awards granted under it, and to establish, amend and revoke rules and regulations for its administration. The Board, in the exercise of this power, may correct any defect, omission or inconsistency in the Plan or in any Award Agreement, in a manner and to the extent it deems necessary or expedient to make the Plan or Award fully effective.
  - (iii) To settle all controversies regarding the Plan and Awards granted under it.
- (iv) To accelerate the time at which an Award may first be exercised or the time during which an Award or any part thereof will vest, notwithstanding the provisions in the Award Agreement stating the time at which it may first be exercised or the time during which it will vest.
- (v) To prohibit the exercise of any Option, SAR or other exercisable Award during a period of up to 30 days prior to the consummation of any pending stock dividend, stock split, combination or exchange of shares, merger, consolidation or other distribution (other than normal cash dividends) of Company assets to stockholders, or any other change affecting the shares of Common Stock or the share price of the Common Stock including any Corporate Transaction, for reasons of administrative convenience.
- (vi) To suspend or terminate the Plan at any time. Suspension or termination of the Plan will not Materially Impair rights and obligations under any Award granted while the Plan is in effect except with the written consent of the affected Participant.

- (vii) To amend the Plan in any respect the Board deems necessary or advisable; provided, however, that stockholder approval will be required for any amendment to the extent required by Applicable Law. Except as provided above, rights under any Award granted before amendment of the Plan will not be Materially Impaired by any amendment of the Plan unless (1) the Company requests the consent of the affected Participant, and (2) such Participant consents in writing.
  - (viii) To submit any amendment to the Plan for stockholder approval.
- (ix) To approve forms of Award Agreements for use under the Plan and to amend the terms of any one or more Awards, including, but not limited to, amendments to provide terms more favorable to the Participant than previously provided in the Award Agreement, subject to any specified limits in the Plan that are not subject to Board discretion; provided however, that, a Participant's rights under any Award will not be Materially Impaired by any such amendment unless (1) the Company requests the consent of the affected Participant, and (2) such Participant consents in writing.
- (x) Generally, to exercise such powers and to perform such acts as the Board deems necessary or expedient to promote the best interests of the Company and that are not in conflict with the provisions of the Plan or Awards.
- (xi) To adopt such procedures and sub-plans as are necessary or appropriate to permit and facilitate participation in the Plan by, or take advantage of specific tax treatment for Awards granted to, Employees, Directors or Consultants who are foreign nationals or employed outside the United States (provided that Board approval will not be necessary for immaterial modifications to the Plan or any Award Agreement to ensure or facilitate compliance with the laws of the relevant foreign jurisdiction).
- (xii) To effect, at any time and from time to time, subject to the consent of any Participant whose Award is Materially Impaired by such action, (1) the reduction of the exercise price (or strike price) of any outstanding Option or SAR; (2) the cancellation of any outstanding Option or SAR and the grant in substitution therefor of (A) a new Option, SAR, Restricted Stock Award, RSU Award or Other Award, under the Plan or another equity plan of the Company, covering the same or a different number of shares of Common Stock, (B) cash and/or (C) other valuable consideration (as determined by the Board); or (3) any other action that is treated as a repricing under generally accepted accounting principles.

# (c) Delegation to Committee.

(i) General. The Board may delegate some or all of the administration of the Plan to a Committee or Committees. If administration of the Plan is delegated to a Committee, the Committee will have, in connection with the administration of the Plan, the powers theretofore possessed by the Board that have been delegated to the Committee, including the power to delegate to another Committee or a subcommittee of the Committee any of the administrative powers the Committee is authorized to exercise (and references in this Plan to the Board will thereafter be to the Committee or subcommittee), subject, however, to such resolutions, not inconsistent with the provisions of the Plan, as may be adopted from time to time by the Board. Each Committee may retain the authority to concurrently administer the Plan with Committee or subcommittee to which it has delegated its authority hereunder and may, at any time, revest in such Committee some or all of the powers previously delegated. The Board may retain the authority to concurrently administer the Plan with any Committee and may, at any time, revest in the Board some or all of the powers previously delegated.

- (ii) Rule 16b-3 Compliance. To the extent an Award is intended to qualify for the exemption from Section 16(b) of the Exchange Act that is available under Rule 16b-3 of the Exchange Act, the Award will be granted by the Board or a Committee that consists solely of two or more Non-Employee Directors, as determined under Rule 16b-3(b)(3) of the Exchange Act and thereafter any action establishing or modifying the terms of the Award will be approved by the Board or a Committee meeting such requirements to the extent necessary for such exemption to remain available.
- **(d) Effect of Board's Decision.** All determinations, interpretations and constructions made by the Board or any Committee in good faith will not be subject to review by any person and will be final, binding and conclusive on all persons.
- **(e) Delegation to an Officer.** The Board or any Committee may delegate to one or more Officers the authority to do one or both of the following (i) designate Employees who are not Officers to be recipients of Options and SARs (and, to the extent permitted by Applicable Law, other types of Awards) and, to the extent permitted by Applicable Law, the terms thereof, and (ii) determine the number of shares of Common Stock to be subject to such Awards granted to such Employees; provided, however, that the resolutions or charter adopted by the Board or any Committee evidencing such delegation will specify the total number of shares of Common Stock that may be subject to the Awards granted by such Officer and that such Officer may not grant an Award to himself or herself. Any such Awards will be granted on the applicable form of Award Agreement most recently approved for use by the Board or the Committee, unless otherwise provided in the resolutions approving the delegation authority. Notwithstanding anything to the contrary herein, neither the Board nor any Committee may delegate to an Officer who is acting solely in the capacity of an Officer (and not also as a Director) the authority to determine the Fair Market Value.

## 8. TAX WITHHOLDING

- (a) Withholding Authorization. As a condition to acceptance of any Award under the Plan, a Participant authorizes withholding from payroll and any other amounts payable to such Participant, and otherwise agrees to make adequate provision for (including), any sums required to satisfy any U.S. federal, state, local and/or foreign tax or social insurance contribution withholding obligations of the Company or an Affiliate, if any, which arise in connection with the exercise, vesting or settlement of such Award, as applicable. Accordingly, a Participant may not be able to exercise an Award even though the Award is vested, and the Company shall have no obligation to issue shares of Common Stock subject to an Award, unless and until such obligations are satisfied.
- **(b)** Satisfaction of Withholding Obligation. To the extent permitted by the terms of an Award Agreement, the Company may, in its sole discretion, satisfy any U.S. federal, state, local and/or foreign tax or social insurance withholding obligation relating to an Award by any of the following means or by a combination of such means: (i) causing the Participant to tender a cash payment; (ii) withholding shares of Common Stock from the shares of Common Stock issued or otherwise issuable to the Participant in connection with the Award; (iii) withholding cash from an Award settled in cash; (iv) withholding payment from any amounts otherwise payable to the Participant; (v) by allowing a Participant to effectuate a "cashless exercise" pursuant to a program developed under Regulation T as promulgated by the Federal Reserve Board, or (vi) by such other method as may be set forth in the Award Agreement.
- (c) No Obligation to Notify or Minimize Taxes; No Liability to Claims. Except as required by Applicable Law the Company has no duty or obligation to any Participant to advise such holder as to the time or manner of exercising such Award. Furthermore, the Company has no duty or obligation to warn or otherwise advise such holder of a pending termination or expiration of an Award or

a possible period in which the Award may not be exercised. The Company has no duty or obligation to minimize the tax consequences of an Award to the holder of such Award and will not be liable to any holder of an Award for any adverse tax consequences to such holder in connection with an Award. As a condition to accepting an Award under the Plan, each Participant (i) agrees to not make any claim against the Company, or any of its Officers, Directors, Employees or Affiliates related to tax liabilities arising from such Award or other Company compensation and (ii) acknowledges that such Participant was advised to consult with his or her own personal tax, financial and other legal advisors regarding the tax consequences of the Award and has either done so or knowingly and voluntarily declined to do so. Additionally, each Participant acknowledges any Option or SAR granted under the Plan is exempt from Section 409A only if the exercise or strike price is at least equal to the "fair market value" of the Common Stock on the date of grant as determined by the Internal Revenue Service and there is no other impermissible deferral of compensation associated with the Award. Additionally, as a condition to accepting an Option or SAR granted under the Plan, each Participant agrees not make any claim against the Company, or any of its Officers, Directors, Employees or Affiliates in the event that the Internal Revenue Service asserts that such exercise price or strike price is less than the "fair market value" of the Common Stock on the date of grant as subsequently determined by the Internal Revenue Service.

(d) Withholding Indemnification. As a condition to accepting an Award under the Plan, in the event that the amount of the Company's and/or its Affiliate's withholding obligation in connection with such Award was greater than the amount actually withheld by the Company and/or its Affiliates, each Participant agrees to indemnify and hold the Company and/or its Affiliates harmless from any failure by the Company and/or its Affiliates to withhold the proper amount.

#### 9. MISCELLANEOUS.

- (a) Source of Shares. The stock issuable under the Plan will be shares of authorized but unissued or reacquired Common Stock, including shares repurchased by the Company on the open market or otherwise.
- **(b) Use of Proceeds from Sales of Common Stock.** Proceeds from the sale of shares of Common Stock pursuant to Awards will constitute general funds of the Company.
- (c) Corporate Action Constituting Grant of Awards. Corporate action constituting a grant by the Company of an Award to any Participant will be deemed completed as of the date of such corporate action, unless otherwise determined by the Board, regardless of when the instrument, certificate, or letter evidencing the Award is communicated to, or actually received or accepted by, the Participant. In the event that the corporate records (e.g., Board consents, resolutions or minutes) documenting the corporate action approving the grant contain terms (e.g., exercise price, vesting schedule or number of shares) that are inconsistent with those in the Award Agreement or related grant documents, the corporate records will control and the Participant will have no legally binding right to the incorrect term in the Award Agreement or related grant documents.
- (d) Stockholder Rights. No Participant will be deemed to be the holder of, or to have any of the rights of a holder with respect to, any shares of Common Stock subject to such Award unless and until (i) such Participant has satisfied all requirements for exercise of the Award pursuant to its terms, if applicable, and (ii) the issuance of the Common Stock subject to such Award is reflected in the records of the Company.
- **(e) No Employment or Other Service Rights.** Nothing in the Plan, any Award Agreement or any other instrument executed thereunder or in connection with any Award granted pursuant thereto

will confer upon any Participant any right to continue to serve the Company or an Affiliate in the capacity in effect at the time the Award was granted or affect the right of the Company or an Affiliate to terminate at will and without regard to any future vesting opportunity that a Participant may have with respect to any Award (i) the employment of an Employee with or without notice and with or without cause, (ii) the service of a Consultant pursuant to the terms of such Consultant's agreement with the Company or an Affiliate, or (iii) the service of a Director pursuant to the Bylaws of the Company or an Affiliate, and any applicable provisions of the corporate law of the state or foreign jurisdiction in which the Company or the Affiliate is incorporated, as the case may be. Further, nothing in the Plan, any Award Agreement or any other instrument executed thereunder or in connection with any Award will constitute any promise or commitment by the Company or an Affiliate regarding the fact or nature of future positions, future work assignments, future compensation or any other term or condition of employment or service or confer any right or benefit under the Award or the Plan unless such right or benefit has specifically accrued under the terms of the Award Agreement and/or Plan.

- (f) Change in Time Commitment. In the event a Participant's regular level of time commitment in the performance of his or her services for the Company and any Affiliates is reduced (for example, and without limitation, if the Participant is an Employee of the Company and the Employee has a change in status from a full-time Employee to a part-time Employee or takes an extended leave of absence) after the date of grant of any Award to the Participant, the Board may determine, to the extent permitted by Applicable Law, to (i) make a corresponding reduction in the number of shares or cash amount subject to any portion of such Award that is scheduled to vest or become payable after the date of such change in time commitment, and (ii) in lieu of or in combination with such a reduction, extend the vesting or payment schedule applicable to such Award. In the event of any such reduction, the Participant will have no right with respect to any portion of the Award that is so reduced or extended.
- **(g) Execution of Additional Documents.** As a condition to accepting an Award under the Plan, the Participant agrees to execute any additional documents or instruments necessary or desirable, as determined in the Plan Administrator's sole discretion, to carry out the purposes or intent of the Award, or facilitate compliance with securities and/or other regulatory requirements, in each case at the Plan Administrator's request.
- **(h)** Electronic Delivery and Participation. Any reference herein or in an Award Agreement to a "written" agreement or document will include any agreement or document delivered electronically, filed publicly at www.sec.gov (or any successor website thereto) or posted on the Company's intranet (or other shared electronic medium controlled by the Company to which the Participant has access). By accepting any Award the Participant consents to receive documents by electronic delivery and to participate in the Plan through any on-line electronic system established and maintained by the Plan Administrator or another third party selected by the Plan Administrator. The form of delivery of any Common Stock (e.g., a stock certificate or electronic entry evidencing such shares) shall be determined by the Company.
- (i) Clawback/Recovery. All Awards granted under the Plan will be subject to recoupment in accordance with any clawback policy that the Company is required to adopt pursuant to the listing standards of any national securities exchange or association on which the Company's securities are listed or as is otherwise required by the Dodd-Frank Wall Street Reform and Consumer Protection Act or other Applicable Law and any clawback policy that the Company otherwise adopts, to the extent applicable and permissible under Applicable Law. In addition, the Board may impose such other clawback, recovery or recoupment provisions in an Award Agreement as the Board determines necessary or appropriate, including but not limited to a reacquisition right in respect of previously acquired shares of Common Stock or other cash or property upon the occurrence of Cause. No recovery of compensation under such a clawback policy will be an event giving rise to a Participant's right to voluntary terminate employment upon a "resignation for good reason," or for a "constructive termination" or any similar term under any plan of or agreement with the Company.

- **(j) Securities Law Compliance.** A Participant will not be issued any shares in respect of an Award unless either (i) the shares are registered under the Securities Act; or (ii) the Company has determined that such issuance would be exempt from the registration requirements of the Securities Act. Each Award also must comply with other Applicable Law governing the Award, and a Participant will not receive such shares if the Company determines that such receipt would not be in material compliance with Applicable Law.
- **(k) Transfer or Assignment of Awards; Issued Shares.** Except as expressly provided in the Plan or the form of Award Agreement, Awards granted under the Plan may not be transferred or assigned by the Participant. After the vested shares subject to an Award have been issued, or in the case of Restricted Stock and similar awards, after the issued shares have vested, the holder of such shares is free to assign, hypothecate, donate, encumber or otherwise dispose of any interest in such shares provided that any such actions are in compliance with the provisions herein, the terms of the Trading Policy and Applicable Law.
- (I) Effect on Other Employee Benefit Plans. The value of any Award granted under the Plan, as determined upon grant, vesting or settlement, shall not be included as compensation, earnings, salaries, or other similar terms used when calculating any Participant's benefits under any employee benefit plan sponsored by the Company or any Affiliate, except as such plan otherwise expressly provides. The Company expressly reserves its rights to amend, modify, or terminate any of the Company's or any Affiliate's employee benefit plans.
- (m) Deferrals. To the extent permitted by Applicable Law, the Board, in its sole discretion, may determine that the delivery of Common Stock or the payment of cash, upon the exercise, vesting or settlement of all or a portion of any Award may be deferred and may also establish programs and procedures for deferral elections to be made by Participants. Deferrals will be made in accordance with the requirements of Section 409A.
- (n) Section 409A. Unless otherwise expressly provided for in an Award Agreement, the Plan and Award Agreements will be interpreted to the greatest extent possible in a manner that makes the Plan and the Awards granted hereunder exempt from Section 409A, and, to the extent not so exempt, in compliance with the requirements of Section 409A. If the Board determines that any Award granted hereunder is not exempt from and is therefore subject to Section 409A, the Award Agreement evidencing such Award will incorporate the terms and conditions necessary to avoid the consequences specified in Section 409A(a)(1) of the Code, and to the extent an Award Agreement is silent on terms necessary for compliance, such terms are hereby incorporated by reference into the Award Agreement. Notwithstanding anything to the contrary in this Plan (and unless the Award Agreement specifically provides otherwise), if the shares of Common Stock are publicly traded, and if a Participant holding an Award that constitutes "deferred compensation" under Section 409A is a "specified employee" for purposes of Section 409A, no distribution or payment of any amount that is due because of a "separation from service" (as defined in Section 409A without regard to alternative definitions thereunder) will be issued or paid before the date that is six months and one day following the date of such Participant's "separation from service" or, if earlier, the date of the Participant's death, unless such distribution or payment can be made in a manner that complies with Section 409A, and any amounts so deferred will be paid in a lump sum on the day after such six month period elapses, with the balance paid thereafter on the original schedule.
- (o) Choice of Law. This Plan and any controversy arising out of or relating to this Plan shall be governed by, and construed in accordance with, the internal laws of the State of Delaware, without regard to conflict of law principles that would result in any application of any law other than the law of the State of Delaware.

#### 10. COVENANTS OF THE COMPANY.

(a) Compliance with Law. The Company will seek to obtain from each regulatory commission or agency, as may be deemed to be necessary, having jurisdiction over the Plan such authority as may be required to grant Awards and to issue and sell shares of Common Stock upon exercise or vesting of the Awards; provided, however, that this undertaking will not require the Company to register under the Securities Act the Plan, any Award or any Common Stock issued or issuable pursuant to any such Award. If, after reasonable efforts and at a reasonable cost, the Company is unable to obtain from any such regulatory commission or agency the authority that counsel for the Company deems necessary or advisable for the lawful issuance and sale of Common Stock under the Plan, the Company will be relieved from any liability for failure to issue and sell Common Stock upon exercise or vesting of such Awards unless and until such authority is obtained. A Participant is not eligible for the grant of an Award or the subsequent issuance of Common Stock pursuant to the Award if such grant or issuance would be in violation of any Applicable Law.

## 11. ADDITIONAL RULES FOR AWARDS SUBJECT TO SECTION 409A.

- **(a) Application.** Unless the provisions of this Section of the Plan are expressly superseded by the provisions in the form of Award Agreement, the provisions of this Section shall apply and shall supersede anything to the contrary set forth in the Award Agreement for a Non-Exempt Award.
- **(b) Non-Exempt Awards Subject to Non-Exempt Severance Arrangements.** To the extent a Non-Exempt Award is subject to Section 409A due to application of a Non-Exempt Severance Arrangement, the following provisions of this subsection (b) apply.
- (i) If the Non-Exempt Award vests in the ordinary course during the Participant's Continuous Service in accordance with the vesting schedule set forth in the Award Agreement, and does not accelerate vesting under the terms of a Non-Exempt Severance Arrangement, in no event will the shares be issued in respect of such Non-Exempt Award any later than the later of: (i) December 31st of the calendar year that includes the applicable vesting date, or (ii) the 60th day that follows the applicable vesting date.
- (ii) If vesting of the Non-Exempt Award accelerates under the terms of a Non-Exempt Severance Arrangement in connection with the Participant's Separation from Service, and such vesting acceleration provisions were in effect as of the date of grant of the Non-Exempt Award and, therefore, are part of the terms of such Non-Exempt Award as of the date of grant, then the shares will be earlier issued in settlement of such Non-Exempt Award upon the Participant's Separation from Service in accordance with the terms of the Non-Exempt Severance Arrangement, but in no event later than the 60<sup>th</sup> day that follows the date of the Participant's Separation from Service. However, if at the time the shares would otherwise be issued the Participant is subject to the distribution limitations contained in Section 409A applicable to "specified employees," as defined in Section 409A(a)(2)(B)(i) of the Code, such shares shall not be issued before the date that is six months following the date of such Participant's Separation from Service, or, if earlier, the date of the Participant's death that occurs within such six month period.
- (iii) If vesting of a Non-Exempt Award accelerates under the terms of a Non-Exempt Severance Arrangement in connection with a Participant's Separation from Service, and such vesting acceleration provisions were not in effect as of the date of grant of the Non-Exempt Award and,

therefore, are not a part of the terms of such Non-Exempt Award on the date of grant, then such acceleration of vesting of the Non-Exempt Award shall not accelerate the issuance date of the shares, but the shares shall instead be issued on the same schedule as set forth in the Grant Notice as if they had vested in the ordinary course during the Participant's Continuous Service, notwithstanding the vesting acceleration of the Non-Exempt Award. Such issuance schedule is intended to satisfy the requirements of payment on a specified date or pursuant to a fixed schedule, as provided under Treasury Regulations Section 1.409A-3(a)(4).

- (c) Treatment of Non-Exempt Awards Upon a Corporate Transaction for Employees and Consultants. The provisions of this subsection (c) shall apply and shall supersede anything to the contrary set forth in the Plan with respect to the permitted treatment of any Non-Exempt Award in connection with a Corporate Transaction if the Participant was either an Employee or Consultant upon the applicable date of grant of the Non-Exempt Award.
- (i) **Vested Non-Exempt Awards.** The following provisions shall apply to any Vested Non-Exempt Award in connection with a Corporate Transaction:
- (1) If the Corporate Transaction is also a Section 409A Change in Control then the Acquiring Entity may not assume, continue or substitute the Vested Non-Exempt Award. Upon the Section 409A Change in Control the settlement of the Vested Non-Exempt Award will automatically be accelerated and the shares will be immediately issued in respect of the Vested Non-Exempt Award. Alternatively, the Company may instead provide that the Participant will receive a cash settlement equal to the Fair Market Value of the shares that would otherwise be issued to the Participant upon the Section 409A Change in Control.
- (2) If the Corporate Transaction is not also a Section 409A Change in Control, then the Acquiring Entity must either assume, continue or substitute each Vested Non-Exempt Award. The shares to be issued in respect of the Vested Non-Exempt Award shall be issued to the Participant by the Acquiring Entity on the same schedule that the shares would have been issued to the Participant if the Corporate Transaction had not occurred. In the Acquiring Entity's discretion, in lieu of an issuance of shares, the Acquiring Entity may instead substitute a cash payment on each applicable issuance date, equal to the Fair Market Value of the shares that would otherwise be issued to the Participant on such issuance dates, with the determination of the Fair Market Value of the shares made on the date of the Corporate Transaction.
- (ii) Unvested Non-Exempt Awards. The following provisions shall apply to any Unvested Non-Exempt Award unless otherwise determined by the Board pursuant to subsection (e) of this Section.
- (1) In the event of a Corporate Transaction, the Acquiring Entity shall assume, continue or substitute any Unvested Non-Exempt Award. Unless otherwise determined by the Board, any Unvested Non-Exempt Award will remain subject to the same vesting and forfeiture restrictions that were applicable to the Award prior to the Corporate Transaction. The shares to be issued in respect of any Unvested Non-Exempt Award shall be issued to the Participant by the Acquiring Entity on the same schedule that the shares would have been issued to the Participant if the Corporate Transaction had not occurred. In the Acquiring Entity's discretion, in lieu of an issuance of shares, the Acquiring Entity may instead substitute a cash payment on each applicable issuance date, equal to the Fair Market Value of the shares that would otherwise be issued to the Participant on such issuance dates, with the determination of Fair Market Value of the shares made on the date of the Corporate Transaction.

- (2) If the Acquiring Entity will not assume, substitute or continue any Unvested Non-Exempt Award in connection with a Corporate Transaction, then such Award shall automatically terminate and be forfeited upon the Corporate Transaction with no consideration payable to any Participant in respect of such forfeited Unvested Non-Exempt Award. Notwithstanding the foregoing, to the extent permitted and in compliance with the requirements of Section 409A, the Board may in its discretion determine to elect to accelerate the vesting and settlement of the Unvested Non-Exempt Award upon the Corporate Transaction, or instead substitute a cash payment equal to the Fair Market Value of such shares that would otherwise be issued to the Participant, as further provided in subsection (e)(ii) below. In the absence of such discretionary election by the Board, any Unvested Non-Exempt Award shall be forfeited without payment of any consideration to the affected Participants if the Acquiring Entity will not assume, substitute or continue the Unvested Non-Exempt Awards in connection with the Corporate Transaction.
- (3) The foregoing treatment shall apply with respect to all Unvested Non-Exempt Awards upon any Corporate Transaction, and regardless of whether or not such Corporate Transaction is also a Section 409A Change in Control.
- **(d) Treatment of Non-Exempt Awards Upon a Corporate Transaction for Non-Employee Directors.** The following provisions of this subsection (d) shall apply and shall supersede anything to the contrary that may be set forth in the Plan with respect to the permitted treatment of a Non-Exempt Director Award in connection with a Corporate Transaction.
- (i) If the Corporate Transaction is also a Section 409A Change in Control then the Acquiring Entity may not assume, continue or substitute the Non-Exempt Director Award. Upon the Section 409A Change in Control the vesting and settlement of any Non-Exempt Director Award will automatically be accelerated and the shares will be immediately issued to the Participant in respect of the Non-Exempt Director Award. Alternatively, the Company may provide that the Participant will instead receive a cash settlement equal to the Fair Market Value of the shares that would otherwise be issued to the Participant upon the Section 409A Change in Control pursuant to the preceding provision.
- (ii) If the Corporate Transaction is not also a Section 409A Change in Control, then the Acquiring Entity must either assume, continue or substitute the Non-Exempt Director Award. Unless otherwise determined by the Board, the Non-Exempt Director Award will remain subject to the same vesting and forfeiture restrictions that were applicable to the Award prior to the Corporate Transaction. The shares to be issued in respect of the Non-Exempt Director Award shall be issued to the Participant by the Acquiring Entity on the same schedule that the shares would have been issued to the Participant if the Corporate Transaction had not occurred. In the Acquiring Entity's discretion, in lieu of an issuance of shares, the Acquiring Entity may instead substitute a cash payment on each applicable issuance date, equal to the Fair Market Value of the shares that would otherwise be issued to the Participant on such issuance dates, with the determination of Fair Market Value made on the date of the Corporate Transaction.
- **(e)** If the RSU Award is a Non-Exempt Award, then the provisions in this Section 11(e) shall apply and supersede anything to the contrary that may be set forth in the Plan or the Award Agreement with respect to the permitted treatment of such Non-Exempt Award:
- (i) Any exercise by the Board of discretion to accelerate the vesting of a Non-Exempt Award shall not result in any acceleration of the scheduled issuance dates for the shares in respect of the Non-Exempt Award unless earlier issuance of the shares upon the applicable vesting dates would be in compliance with the requirements of Section 409A.

- (ii) The Company explicitly reserves the right to earlier settle any Non-Exempt Award to the extent permitted and in compliance with the requirements of Section 409A, including pursuant to any of the exemptions available in Treasury Regulations Section 1.409A-3(j)(4)(ix).
- (iii) To the extent the terms of any Non-Exempt Award provide that it will be settled upon a Change in Control or Corporate Transaction, to the extent it is required for compliance with the requirements of Section 409A, the Change in Control or Corporate Transaction event triggering settlement must also constitute a Section 409A Change in Control. To the extent the terms of a Non-Exempt Award provides that it will be settled upon a termination of employment or termination of Continuous Service, to the extent it is required for compliance with the requirements of Section 409A, the termination event triggering settlement must also constitute a Separation From Service. However, if at the time the shares would otherwise be issued to a Participant in connection with a "separation from service" such Participant is subject to the distribution limitations contained in Section 409A applicable to "specified employees," as defined in Section 409A(a)(2)(B)(i) of the Code, such shares shall not be issued before the date that is six months following the date of the Participant's Separation From Service, or, if earlier, the date of the Participant's death that occurs within such six month period.
- (iv) The provisions in this subsection (e) for delivery of the shares in respect of the settlement of an RSU Award that is a Non-Exempt Award are intended to comply with the requirements of Section 409A so that the delivery of the shares to the Participant in respect of such Non-Exempt Award will not trigger the additional tax imposed under Section 409A, and any ambiguities herein will be so interpreted.

## 12. SEVERABILITY.

If all or any part of the Plan or any Award Agreement is declared by any court or governmental authority to be unlawful or invalid, such unlawfulness or invalidity shall not invalidate any portion of the Plan or such Award Agreement not declared to be unlawful or invalid. Any Section of the Plan or any Award Agreement (or part of such a Section) so declared to be unlawful or invalid shall, if possible, be construed in a manner which will give effect to the terms of such Section or part of a Section to the fullest extent possible while remaining lawful and valid.

#### 13. TERMINATION OF THE PLAN.

The Board may suspend or terminate the Plan at any time. No Incentive Stock Options may be granted after the tenth anniversary of the earlier of: (i) the Adoption Date, or (ii) the date the Plan is approved by the Company's stockholders. No Awards may be granted under the Plan while the Plan is suspended or after it is terminated.

## 14. DEFINITIONS.

As used in the Plan, the following definitions apply to the capitalized terms indicated below:

- (a) "Acquiring Entity" means the surviving or acquiring corporation (or its parent company) in connection with a Corporate Transaction.
- (b) "Adoption Date" means the date the Plan is first approved by the Board or Compensation Committee.
- **(c)** "Affiliate" means, at the time of determination, any "parent" or "subsidiary" of the Company as such terms are defined in Rule 405 promulgated under the Securities Act. The Board may determine the time or times at which "parent" or "subsidiary" status is determined within the foregoing definition.
- (d) "Applicable Law" means any applicable securities, federal, state, foreign, material local or municipal or other law, statute, constitution, principle of common law, resolution, ordinance, code, edict, decree, rule, listing rule, regulation, judicial decision, ruling or requirement issued, enacted, adopted, promulgated, implemented or otherwise put into effect by or under the authority of any Governmental Body (including under the authority of any applicable self-regulating organization such as the Nasdaq Stock Market, New York Stock Exchange, or the Financial Industry Regulatory Authority).
- **(e)** "*Award*" means any right to receive Common Stock, cash or other property granted under the Plan (including an Incentive Stock Option, a Nonstatutory Stock Option, a Restricted Stock Award, an RSU Award, a SAR, a Performance Award or any Other Award).
- **(f)** "Award Agreement" means a written or electronic agreement between the Company and a Participant evidencing the terms and conditions of an Award. The Award Agreement generally consists of the Grant Notice and the agreement containing the written summary of the general terms and conditions applicable to the Award and which is provided, including through electronic means, to a Participant along with the Grant Notice.
- **(g)** "Board" means the Board of Directors of the Company (or its designee). Any decision or determination made by the Board shall be a decision or determination that is made in the sole discretion of the Board (or its designee), and such decision or determination shall be final and binding on all Participants
- **(h)** "Capitalization Adjustment" means any change that is made in, or other events that occur with respect to, the Common Stock subject to the Plan or subject to any Award after the date the Plan is adopted by the Board without the receipt of consideration by the Company through merger, consolidation, reorganization, recapitalization, reincorporation, stock dividend, dividend in property other than cash, large nonrecurring cash dividend, stock split, reverse stock split, liquidating dividend, combination of shares, exchange of shares, change in corporate structure or any similar equity restructuring transaction, as that term is used in Statement of Financial Accounting Standards Board Accounting Standards Codification Topic 718 (or any successor thereto). Notwithstanding the foregoing, the conversion of any convertible securities of the Company will not be treated as a Capitalization Adjustment.
- (i) "Cause" has the meaning ascribed to such term in any written agreement between a Participant and the Company defining such term and, in the absence of such agreement, such term means, with respect to a Participant, the occurrence of any of the following events: (i) the Participant's dishonest

statements or acts with respect to the Company or any Affiliate of the Company, or any current or prospective customers, suppliers, vendors or other third parties with which such entity does business; (ii) the Participant's commission of (A) a felony or (B) any misdemeanor involving moral turpitude, deceit, dishonesty or fraud; (iii) the Participant's failure to perform the Participant's assigned duties and responsibilities to the reasonable satisfaction of the Company which failure continues, in the reasonable judgment of the Company, after written notice given to the Participant by the Company; (iv) the Participant's gross negligence, willful misconduct or insubordination with respect to the Company or any Affiliate of the Company; or (v) the Participant's material violation of any provision of any agreement(s) between the Participant and the Company relating to noncompetition, nonsolicitation, nondisclosure and/or assignment of inventions. The determination that a termination of the Participant's Continuous Service is either for Cause or without Cause will be made by the Board with respect to Participants who are executive officers of the Company and by the Company's Chief Executive Officer with respect to Participants who are not executive officers of the Company. Any determination by the Company that the Continuous Service of a Participant was terminated with or without Cause for the purposes of outstanding Awards held by such Participant will have no effect upon any determination of the rights or obligations of the Company or such Participant for any other purpose.

- **(j)** "Change in Control" or "Change of Control" means the occurrence, in a single transaction or in a series of related transactions, of any one or more of the following events:
- (i) any Exchange Act Person becomes the Owner, directly or indirectly, of securities of the Company representing more than 50% of the combined voting power of the Company's then outstanding securities other than by virtue of a merger, consolidation or similar transaction. Notwithstanding the foregoing, a Change in Control shall not be deemed to occur (A) on account of the acquisition of securities of the Company directly from the Company, (B) on account of the acquisition of securities of the Company by an investor, any affiliate thereof or any other Exchange Act Person that acquires the Company's securities in a transaction or series of related transactions the primary purpose of which is to obtain financing for the Company through the issuance of equity securities, or (C) solely because the level of Ownership held by any Exchange Act Person (the "Subject Person") exceeds the designated percentage threshold of the outstanding voting securities as a result of a repurchase or other acquisition of voting securities by the Company reducing the number of shares outstanding, provided that if a Change in Control would occur (but for the operation of this sentence) as a result of the acquisition of voting securities by the Company, and after such share acquisition, the Subject Person becomes the Owner of any additional voting securities that, assuming the repurchase or other acquisition had not occurred, increases the percentage of the then outstanding voting securities Owned by the Subject Person over the designated percentage threshold, then a Change in Control shall be deemed to occur;
- (ii) there is consummated a merger, consolidation or similar transaction involving (directly or indirectly) the Company and, immediately after the consummation of such merger, consolidation or similar transaction, the stockholders of the Company immediately prior thereto do not Own, directly or indirectly, either (A) outstanding voting securities representing more than 50% of the combined outstanding voting power of the Acquiring Entity in such merger, consolidation or similar transaction or (B) more than 50% of the combined outstanding voting power of the parent of the Acquiring Entity in such merger, consolidation or similar transaction, in each case in substantially the same proportions as their Ownership of the outstanding voting securities of the Company immediately prior to such transaction;
- (iii) there is consummated a sale, lease, exclusive license or other disposition of all or substantially all of the consolidated assets of the Company and its Subsidiaries, other than a sale, lease, license or other disposition of all or substantially all of the consolidated assets of the Company and

its Subsidiaries to an Entity, more than 50% of the combined voting power of the voting securities of which are Owned by stockholders of the Company in substantially the same proportions as their Ownership of the outstanding voting securities of the Company immediately prior to such sale, lease, license or other disposition; or

(iv) individuals who, on the date the Plan is adopted by the Board, are members of the Board (the "*Incumbent Board*") cease for any reason to constitute at least a majority of the members of the Board; provided, however, that if the appointment or election (or nomination for election) of any new Board member was approved or recommended by a majority vote of the members of the Incumbent Board then still in office, such new member shall, for purposes of this Plan, be considered as a member of the Incumbent Board.

Notwithstanding the foregoing or any other provision of this Plan, (A) the term Change in Control shall not include a sale of assets, merger or other transaction effected exclusively for the purpose of changing the domicile of the Company, (B) the definition of Change in Control (or any analogous term) in an individual written agreement between the Company or any Affiliate and the Participant shall supersede the foregoing definition with respect to Awards subject to such agreement; provided, however, that if no definition of Change in Control or any analogous term is set forth in such an individual written agreement, the foregoing definition shall apply, and (C) with respect to any nonqualified deferred compensation that becomes payable on account of the Change in Control, the transaction or event described in clause (i), (ii), (iii), or (iv) also constitutes a Section 409A Change in Control if required in order for the payment not to violate Section 409A of the Code.

- (k) "Code" means the Internal Revenue Code of 1986, as amended, including any applicable regulations and guidance thereunder.
- (I) "Committee" means the Compensation Committee and any other committee of one or more Directors to whom authority has been delegated by the Board or Compensation Committee in accordance with the Plan.
  - (m) "Common Stock" means the Class A common stock of the Company.
  - (n) "Company" means Origin Materials, Inc., a Delaware corporation.
  - (o) "Compensation Committee" means the Compensation Committee of the Board.
- **(p)** "Consultant" means any person, including an advisor, who is (i) engaged by the Company or an Affiliate to render consulting or advisory services and is compensated for such services, or (ii) serving as a member of the board of directors of an Affiliate and is compensated for such services. However, service solely as a Director, or payment of a fee for such service, will not cause a Director to be considered a "Consultant" for purposes of the Plan. Notwithstanding the foregoing, a person is treated as a Consultant under this Plan only if a Form S-8 Registration Statement under the Securities Act is available to register either the offer or the sale of the Company's securities to such person.
- **(q)** "Continuous Service" means that the Participant's service with the Company or an Affiliate, whether as an Employee, Director or Consultant, is not interrupted or terminated. A change in the capacity in which the Participant renders service to the Company or an Affiliate as an Employee, Director or Consultant or a change in the Entity for which the Participant renders such service, provided that there is no interruption or termination of the Participant's service with the Company or an Affiliate, will not terminate a Participant's Continuous Service; provided, however, that if the Entity for which a Participant is rendering services ceases to qualify as an Affiliate, as determined by the Board, such

Participant's Continuous Service will be considered to have terminated on the date such Entity ceases to qualify as an Affiliate. For example, a change in status from an Employee of the Company to a Consultant of an Affiliate or to a Director will not constitute an interruption of Continuous Service. To the extent permitted by law, the Board or the chief executive officer of the Company, in that party's sole discretion, may determine whether Continuous Service will be considered interrupted in the case of (i) any leave of absence approved by the Board or chief executive officer, including sick leave, military leave or any other personal leave, or (ii) transfers between the Company, an Affiliate, or their successors. Notwithstanding the foregoing, a leave of absence will be treated as Continuous Service for purposes of vesting in an Award only to such extent as may be provided in the Company's leave of absence policy, in the written terms of any leave of absence agreement or policy applicable to the Participant, or as otherwise required by law. In addition, to the extent required for exemption from or compliance with Section 409A, the determination of whether there has been a termination of Continuous Service will be made, and such term will be construed, in a manner that is consistent with the definition of "separation from service" as defined under Treasury Regulation Section 1.409A-1(h) (without regard to any alternative definition thereunder).

- **(r)** "Corporate Transaction" means the consummation, in a single transaction or in a series of related transactions, of any one or more of the following events:
- (i) a sale or other disposition of all or substantially all, as determined by the Board, of the consolidated assets of the Company and its Subsidiaries;
  - (ii) a sale or other disposition of at least 50% of the outstanding securities of the Company;
  - (iii) a merger, consolidation or similar transaction following which the Company is not the surviving corporation; or
- (iv) a merger, consolidation or similar transaction following which the Company is the surviving corporation but the shares of Common Stock outstanding immediately preceding the merger, consolidation or similar transaction are converted or exchanged by virtue of the merger, consolidation or similar transaction into other property, whether in the form of securities, cash or otherwise.

Notwithstanding the foregoing or any other provision of this Plan, (A) the term Corporate Transaction shall not include a sale of assets, merger or other transaction effected exclusively for the purpose of changing the domicile of the Company, (B) the definition of Corporate Transaction (or any analogous term) in an individual written agreement between the Company or any Affiliate and the Participant shall supersede the foregoing definition with respect to Awards subject to such agreement; provided, however, that if no definition of Corporate Transaction or any analogous term is set forth in such an individual written agreement, the foregoing definition shall apply, and (C) with respect to any nonqualified deferred compensation that becomes payable on account of the Corporate Transaction, the transaction or event described in clause (i), (ii), (iii), or (iv) also constitutes a Section 409A Change in Control if required in order for the payment not to violate Section 409A of the Code.

- (s) "Director" means a member of the Board.
- (t) "determine" or "determined" means as determined by the Board or the Committee (or its designee) in its sole discretion.

- (u) "Disability" means, with respect to a Participant, such Participant is unable to engage in any substantial gainful activity by reason of any medically determinable physical or mental impairment which can be expected to result in death or which has lasted or can be expected to last for a continuous period of not less than 12 months, as provided in Section 22(e)(3) of the Code, and will be determined by the Board on the basis of such medical evidence as the Board deems warranted under the circumstances.
- **(v)** "*Effective Date*" means the effective date of this Plan, which is the date of the closing of the transactions contemplated by the Agreement and Plan of Merger by and among Artius Acquisition Inc., Zero Carbon Merger Sub Inc. and Micromidas, Inc., dated as of February 16, 2021, provided that this Plan is approved by the Company's stockholders prior to such date.
- **(w)** "*Employee*" means any person employed by the Company or an Affiliate. However, service solely as a Director, or payment of a fee for such services, will not cause a Director to be considered an "Employee" for purposes of the Plan.
  - (x) "*Employer*" means the Company or the Affiliate of the Company that employs the Participant.
  - (y) "Entity" means a corporation, partnership, limited liability company or other entity.
  - (z) "Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.
- (aa) "Exchange Act Person" means any natural person, Entity or "group" (within the meaning of Section 13(d) or 14(d) of the Exchange Act), except that "Exchange Act Person" will not include (i) the Company or any Subsidiary of the Company, (ii) any employee benefit plan of the Company or any Subsidiary of the Company or any trustee or other fiduciary holding securities under an employee benefit plan of the Company or any Subsidiary of the Company, (iii) an underwriter temporarily holding securities pursuant to a registered public offering of such securities, (iv) an Entity Owned, directly or indirectly, by the stockholders of the Company in substantially the same proportions as their Ownership of stock of the Company, or (v) any natural person, Entity or "group" (within the meaning of Section 13(d) or 14(d) of the Exchange Act) that, as of the Effective Date, is the Owner, directly or indirectly, of securities of the Company representing more than 50% of the combined voting power of the Company's then outstanding securities.
- **(bb)** "Fair Market Value" means, as of any date, unless otherwise determined by the Board, the value of the Common Stock (as determined on a per share or aggregate basis, as applicable) determined as follows:
- (i) If the Common Stock is listed on any established stock exchange or traded on any established market, the Fair Market Value will be the closing sales price for such stock as quoted on such exchange or market (or the exchange or market with the greatest volume of trading in the Common Stock) on the date of determination, as reported in a source the Board deems reliable.
- (ii) If there is no closing sales price for the Common Stock on the date of determination, then the Fair Market Value will be the closing selling price on the last preceding date for which such quotation exists.
- (iii) In the absence of such markets for the Common Stock, or if otherwise determined by the Board, the Fair Market Value will be determined by the Board in good faith and in a manner that complies with Sections 409A and 422 of the Code.

- (cc) "Fully-Diluted Common Stock" means, as of any date, the aggregate number of (i) shares of Common Stock issued and outstanding and (ii) securities convertible into or exercisable for shares of Common Stock (whether vested or unvested).
- (dd) "Governmental Body" means any: (a) nation, state, commonwealth, province, territory, county, municipality, district or other jurisdiction of any nature; (b) federal, state, local, municipal, foreign or other government; (c) governmental or regulatory body, or quasi-governmental body of any nature (including any governmental division, department, administrative agency or bureau, commission, authority, instrumentality, official, ministry, fund, foundation, center, organization, unit, body or Entity and any court or other tribunal, and for the avoidance of doubt, any Tax authority) or other body exercising similar powers or authority; or (d) self-regulatory organization (including the Nasdaq Stock Market, New York Stock Exchange, and the Financial Industry Regulatory Authority).
- **(ee)** "*Grant Notice*" means the notice provided to a Participant that he or she has been granted an Award under the Plan and which includes the name of the Participant, the type of Award, the date of grant of the Award, number of shares of Common Stock subject to the Award or potential cash payment right, (if any), the vesting schedule for the Award (if any) and other key terms applicable to the Award.
- **(ff)** "*Incentive Stock Option*" means an option granted pursuant to Section 4 of the Plan that is intended to be, and qualifies as, an "incentive stock option" within the meaning of Section 422 of the Code.
- (gg) "Materially Impair" means any amendment to the terms of the Award that materially adversely affects the Participant's rights under the Award. A Participant's rights under an Award will not be deemed to have been Materially Impaired by any such amendment if the Board, in its sole discretion, determines that the amendment, taken as a whole, does not materially impair the Participant's rights. For example, the following types of amendments to the terms of an Award do not Materially Impair the Participant's rights under the Award: (i) imposition of reasonable restrictions on the minimum number of shares subject to an Option or SAR that may be exercised, (ii) to maintain the qualified status of the Award as an Incentive Stock Option under Section 422 of the Code, (iii) to change the terms of an Incentive Stock Option in a manner that disqualifies, impairs or otherwise affects the qualified status of the Award as an Incentive Stock Option under Section 422 of the Code, (iv) to clarify the manner of exemption from, or to bring the Award into compliance with or qualify it for an exemption from, Section 409A, or (v) to comply with other Applicable Laws.
- **(hh)** "Non-Employee Director" means a Director who either (i) is not a current employee or officer of the Company or an Affiliate, does not receive compensation, either directly or indirectly, from the Company or an Affiliate for services rendered as a consultant or in any capacity other than as a Director (except for an amount as to which disclosure would not be required under Item 404(a) of Regulation S-K promulgated pursuant to the Securities Act ("**Regulation S-K**")), does not possess an interest in any other transaction for which disclosure would be required under Item 404(a) of Regulation S-K, and is not engaged in a business relationship for which disclosure would be required pursuant to Item 404(b) of Regulation S-K, or (ii) is otherwise considered a "non-employee director" for purposes of Rule 16b-3.
- (ii) "Non-Exempt Award" means any Award that is subject to, and not exempt from, Section 409A, including as the result of (i) a deferral of the issuance of the shares subject to the Award which is elected by the Participant or imposed by the Company, or (ii) the terms of any Non-Exempt Severance Agreement.

- (jj) "Non-Exempt Director Award" means a Non-Exempt Award granted to a Participant who was a Director but not an Employee on the applicable grant date.
- **(kk)** "Non-Exempt Severance Arrangement" means a severance arrangement or other agreement between the Participant and the Company that provides for acceleration of vesting of an Award and issuance of the shares in respect of such Award upon the Participant's termination of employment or separation from service (as such term is defined in Section 409A(a)(2)(A)(i) of the Code (and without regard to any alternative definition thereunder) ("Separation from Service") and such severance benefit does not satisfy the requirements for an exemption from application of Section 409A provided under Treasury Regulations Section 1.409A-1(b)(4), 1.409A-1(b)(9) or otherwise.
  - (II) "Nonstatutory Stock Option" means any option granted pursuant to Section 4 of the Plan that does not qualify as an Incentive Stock Option.
  - (mm) "Officer" means a person who is an officer of the Company within the meaning of Section 16 of the Exchange Act.
- (nn) "Option" means an Incentive Stock Option or a Nonstatutory Stock Option to purchase shares of Common Stock granted pursuant to the Plan.
- (oo) "Option Agreement" means a written or electronic agreement between the Company and the Optionholder evidencing the terms and conditions of the Option grant. The Option Agreement includes the Grant Notice for the Option and the agreement containing the written summary of the general terms and conditions applicable to the Option and which is provided, including through electronic means, to a Participant along with the Grant Notice. Each Option Agreement will be subject to the terms and conditions of the Plan.
- **(pp)** "*Optionholder*" means a person to whom an Option is granted pursuant to the Plan or, if applicable, such other person who holds an outstanding Option.
- (qq) "Other Award" means an award valued in whole or in part by reference to, or otherwise based on, Common Stock, including the appreciation in value thereof (e.g., options or stock rights with an exercise price or strike price less than 100% of the Fair Market Value at the time of grant) that is not an Incentive Stock Option, Nonstatutory Stock Option, SAR, Restricted Stock Award, RSU Award or Performance Award.
- **(rr)** "Other Award Agreement" means a written or electronic agreement between the Company and a holder of an Other Award evidencing the terms and conditions of an Other Award grant. Each Other Award Agreement will be subject to the terms and conditions of the Plan.
- (ss) "Own," "Owner," "Owner," "Ownership" means that a person or Entity will be deemed to "Own," to have "Owned," to be the "Owner" of, or to have acquired "Ownership" of securities if such person or Entity, directly or indirectly, through any contract, arrangement, understanding, relationship or otherwise, has or shares voting power, which includes the power to vote or to direct the voting, with respect to such securities.
- **(tt)** "Participant" means an Employee, Director or Consultant to whom an Award is granted pursuant to the Plan or, if applicable, such other person who holds an outstanding Award.
- **(uu)** "Performance Award" means an Award that may vest or may be exercised or a cash award that may vest or become earned and paid contingent upon the attainment during a Performance

Period of certain Performance Goals and which is granted under the terms and conditions of Section 5(b) pursuant to such terms as are approved by the Board. In addition, to the extent permitted by Applicable Law and set forth in the applicable Award Agreement, the Board may determine that cash or other property may be used in payment of Performance Awards. Performance Awards that are settled in cash or other property are not required to be valued in whole or in part by reference to, or otherwise based on, the Common Stock.

"Performance Criteria" means the one or more criteria that the Board will select for purposes of establishing the Performance Goals for a Performance Period. The Performance Criteria that will be used to establish such Performance Goals may be based on any one of, or combination of, the following as determined by the Board: earnings (including earnings per share and net earnings); earnings before interest, taxes and depreciation; earnings before interest, taxes, depreciation and amortization; total stockholder return; return on equity or average stockholder's equity; return on assets, investment, or capital employed; stock price; margin (including gross margin); income (before or after taxes); operating income; operating income after taxes; pre-tax profit; operating cash flow; sales or revenue targets; increases in revenue or product revenue; expenses and cost reduction goals; improvement in or attainment of working capital levels; economic value added (or an equivalent metric); market share; cash flow; cash flow per share; share price performance; debt reduction; customer satisfaction; stockholders' equity; capital expenditures; debt levels; operating profit or net operating profit; workforce diversity; growth of net income or operating income; billings; financing; regulatory milestones; stockholder liquidity; corporate governance and compliance; intellectual property; personnel matters; progress of internal research; progress of partnered programs; partner satisfaction; budget management; partner or collaborator achievements; internal controls, including those related to the Sarbanes-Oxley Act of 2002; investor relations, analysts and communication; implementation or completion of projects or processes; employee retention; number of users, including unique users; strategic partnerships or transactions (including in-licensing and out-licensing of intellectual property); establishing relationships with respect to the marketing, distribution and sale of the Company's products; supply chain achievements; co-development, co-marketing, profit sharing, joint venture or other similar arrangements; individual performance goals; corporate development and planning goals; and other measures of performance selected by the Board or Committee whether or not listed herein.

(ww) "Performance Goals" means, for a Performance Period, the one or more goals established by the Board for the Performance Period based upon the Performance Criteria. Performance Goals may be based on a Company-wide basis, with respect to one or more business units, divisions, Affiliates, or business segments, and in either absolute terms or relative to the performance of one or more comparable companies or the performance of one or more relevant indices. Unless specified otherwise by the Board (i) in the Award Agreement at the time the Award is granted or (ii) in such other document setting forth the Performance Goals at the time the Performance Goals are established, the Board will appropriately make adjustments in the method of calculating the attainment of Performance Goals for a Performance Period as follows: (1) to exclude restructuring and/or other nonrecurring charges; (2) to exclude exchange rate effects; (3) to exclude the effects of changes to generally accepted accounting principles; (4) to exclude the effects of any statutory adjustments to corporate tax rates; (5) to exclude the effects of items that are "unusual" in nature or occur "infrequently" as determined under generally accepted accounting principles; (6) to exclude the dilutive effects of acquisitions or joint ventures; (7) to assume that any business divested by the Company achieved performance objectives at targeted levels during the balance of a Performance Period following such divestiture; (8) to exclude the effect of any change in the outstanding shares of common stock of the Company by reason of any stock dividend or split, stock repurchase, reorganization, recapitalization, merger, consolidation, spin-off, combination or exchange of shares or other similar corporate change, or any distributions to common stockholders other than regular cash dividends; (9) to exclude the effects of stock based compensation and the award of bonuses under the Company's bonus plans; (10) to exclude costs incurred in connection with

potential acquisitions or divestitures that are required to be expensed under generally accepted accounting principles; and (11) to exclude the goodwill and intangible asset impairment charges that are required to be recorded under generally accepted accounting principles. In addition, the Board may establish or provide for other adjustment items in the Award Agreement at the time the Award is granted or in such other document setting forth the Performance Goals at the time the Performance Goals are established. In addition, the Board retains the discretion to reduce or eliminate the compensation or economic benefit due upon attainment of Performance Goals and to define the manner of calculating the Performance Criteria it selects to use for such Performance Period. Partial achievement of the specified criteria may result in the payment or vesting corresponding to the degree of achievement as specified in the Award Agreement or the written terms of a Performance Cash Award.

- (xx) "Performance Period" means the period of time selected by the Board over which the attainment of one or more Performance Goals will be measured for the purpose of determining a Participant's right to vesting or exercise of an Award. Performance Periods may be of varying and overlapping duration, at the sole discretion of the Board.
  - (yy) "Plan" means this Origin Materials, Inc. 2021 Equity Incentive Plan.
- (zz) "Plan Administrator" means the person, persons, and/or third-party administrator designated by the Company to administer the day to day operations of the Plan and the Company's other equity incentive programs.
- **(aaa)** "*Post-Termination Exercise Period*" means the period following termination of a Participant's Continuous Service within which an Option or SAR is exercisable, as specified in Section 4(h).
- **(bbb)** "Restricted Stock Award" or "RSA" means an Award of shares of Common Stock which is granted pursuant to the terms and conditions of Section 5(a).
- (ccc) "Restricted Stock Award Agreement" means a written or electronic agreement between the Company and a holder of a Restricted Stock Award evidencing the terms and conditions of a Restricted Stock Award grant. The Restricted Stock Award Agreement includes the Grant Notice for the Restricted Stock Award and the agreement containing the written summary of the general terms and conditions applicable to the Restricted Stock Award and which is provided, including by electronic means, to a Participant along with the Grant Notice. Each Restricted Stock Award Agreement will be subject to the terms and conditions of the Plan.
- **(ddd)** "*RSU Award*" or "*RSU*" means an Award of restricted stock units representing the right to receive an issuance of shares of Common Stock which is granted pursuant to the terms and conditions of Section 5(a).
- **(eee)** "RSU Award Agreement" means a written or electronic agreement between the Company and a holder of an RSU Award evidencing the terms and conditions of an RSU Award. The RSU Award Agreement includes the Grant Notice for the RSU Award and the agreement containing the written summary of the general terms and conditions applicable to the RSU Award and which is provided, including by electronic means, to a Participant along with the Grant Notice. Each RSU Award Agreement will be subject to the terms and conditions of the Plan.
  - (fff) "Rule 16b-3" means Rule 16b-3 promulgated under the Exchange Act or any successor to Rule 16b-3, as in effect from time to time.

- (ggg) "Rule 405" means Rule 405 promulgated under the Securities Act.
- **(hhh)** "Section 409A" means Section 409A of the Code and the regulations and other guidance thereunder.
- (iii) "Section 409A Change in Control" means a change in the ownership or effective control of the Company, or in the ownership of a substantial portion of the Company's assets, as provided in Section 409A(a)(2)(A)(v) of the Code and Treasury Regulations Section 1.409A-3(i)(5) (without regard to any alternative definition thereunder).
  - (jjj) "Securities Act" means the Securities Act of 1933, as amended.
  - (kkk) "Share Reserve" means the number of shares available for issuance under the Plan as set forth in Section 2(a).
- (III) "Stock Appreciation Right" or "SAR" means a right to receive the appreciation on Common Stock that is granted pursuant to the terms and conditions of Section 4.
- (mmm) "SAR Agreement" means a written or electronic agreement between the Company and a holder of a SAR evidencing the terms and conditions of a SAR grant. The SAR Agreement includes the Grant Notice for the SAR and the agreement containing the written summary of the general terms and conditions applicable to the SAR and which is provided, including by electronic means, to a Participant along with the Grant Notice. Each SAR Agreement will be subject to the terms and conditions of the Plan.
- **(nnn)** "Subsidiary" means, with respect to the Company, (i) any corporation of which more than 50% of the outstanding capital stock having ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether, at the time, stock of any other class or classes of such corporation will have or might have voting power by reason of the happening of any contingency) is at the time, directly or indirectly, Owned by the Company, and (ii) any partnership, limited liability company or other entity in which the Company has a direct or indirect interest (whether in the form of voting or participation in profits or capital contribution) of more than 50%.
- **(000)** "*Ten Percent Stockholder*" means a person who Owns (or is deemed to Own pursuant to Section 424(d) of the Code) stock possessing more than 10% of the total combined voting power of all classes of stock of the Company or any Affiliate.
- **(ppp)** "*Trading Policy*" means the Company's policy permitting certain individuals to sell Company shares only during certain "window" periods and/or otherwise restricts the ability of certain individuals to transfer or encumber Company shares, as in effect from time to time.
- **(qqq)** "Unvested Non-Exempt Award" means the portion of any Non-Exempt Award that had not vested in accordance with its terms upon or prior to the date of any Corporate Transaction.
- **(rrr)** "Vested Non-Exempt Award" means the portion of any Non-Exempt Award that had vested in accordance with its terms upon or prior to the date of a Corporate Transaction.

# ORIGIN MATERIALS, INC. RSU AWARD GRANT NOTICE (2021 EQUITY INCENTIVE PLAN)

Origin Materials, Inc. (the "*Company*") has awarded to you (the "*Participant*") the number of restricted stock units specified and on the terms set forth below in consideration of your services (the "*RSU Award*"). Your RSU Award is subject to all of the terms and conditions as set forth herein and in the Origin Materials, Inc. 2021 Equity Incentive Plan (the "*Plan*") and the Award Agreement (the "*Agreement*"), which are attached hereto and incorporated herein in their entirety. Capitalized terms not explicitly defined herein but defined in the Plan or the Agreement shall have the meanings set forth in the Plan or the Agreement.

Date of Grant.	
Vesting Commenceme	ent Date:
Number of Restricted	Stock Units:
Vesting Schedule:	[].  Notwithstanding the foregoing, vesting shall terminate upon the Participant's termination of Continuous Service.
Issuance Schedule:	One share of Common Stock will be issued at the time set forth in Section 5 of the Agreement for each restricted stock unit which vests.

Participant:

**Participant Acknowledgements:** By your signature below or by electronic acceptance or authentication in a form authorized by the Company, you understand and agree that:

- The RSU Award is governed by this RSU Award Grant Notice (the "*Grant Notice*"), and the provisions of the Plan and the Agreement, all of which are made a part of this document. Unless otherwise provided in the Plan, this Grant Notice and the Agreement (together, the "*RSU Award Agreement*") may not be modified, amended or revised except in a writing signed by you and a duly authorized officer of the Company.
- You have read and are familiar with the provisions of the Plan, the RSU Award Agreement and the Prospectus. In the event of any conflict between the provisions in the RSU Award Agreement, or the Prospectus and the terms of the Plan, the terms of the Plan shall control.
- The RSU Award Agreement sets forth the entire understanding between you and the Company regarding the acquisition of Common Stock and supersedes all prior oral and written agreements, promises and/or representations on that subject with the exception of: (i) other equity awards previously granted to you, and (ii) any written employment agreement, offer letter, severance agreement, written severance plan or policy, or other written agreement between the Company and you in each case that specifies the terms that should govern this RSU Award.

ORIGIN MATERIALS, INC.:	PARTICIPANT:
Ву:	
Signature	Signature
Title:	Date:
Date:	
ATTACHMENTS: Award Agreement, 2021 Equity Incentive Plan	

#### ATTACHMENT I

## ORIGIN MATERIALS, INC. AWARD AGREEMENT (2021 EQUITY INCENTIVE PLAN)

As reflected by your RSU Award Grant Notice ("Grant Notice"), Origin Materials, Inc. (the "Company") has granted you a RSU Award under the Origin Materials, Inc. 2021 Equity Incentive Plan (the "Plan") for the number of restricted stock units as indicated in your Grant Notice (the "RSU Award"). The terms of your RSU Award as specified in this Award Agreement for your RSU Award (this "Agreement") and the Grant Notice constitute your "RSU Award Agreement". Defined terms not explicitly defined in this Agreement but defined in the Grant Notice or the Plan shall have the same definitions as in the Grant Notice or Plan, as applicable.

The general terms applicable to your RSU Award are as follows:

- 1. GOVERNING PLAN DOCUMENT. Your RSU Award is subject to all the provisions of the Plan, including but not limited to the provisions in:
- (a) Section 6 of the Plan regarding the impact of a Capitalization Adjustment, dissolution, liquidation, or Corporate Transaction on your RSU Award;
- **(b)** Section 9(e) of the Plan regarding the Company's retained rights to terminate your Continuous Service notwithstanding the grant of the RSU Award; and
  - (c) Section 8 of the Plan regarding the tax consequences of your RSU Award.

Your RSU Award is further subject to all interpretations, amendments, rules and regulations, which may from time to time be promulgated and adopted pursuant to the Plan. In the event of any conflict between the RSU Award Agreement and the provisions of the Plan, the provisions of the Plan shall control.

- 2. GRANT OF THE RSU AWARD. This RSU Award represents your right to be issued on a future date the number of shares of the Company's Common Stock that is equal to the number of restricted stock units indicated in the Grant Notice as modified to reflect any Capitalization Adjustment and subject to your satisfaction of the vesting conditions set forth therein (the "Restricted Stock Units"). Any additional Restricted Stock Units that become subject to the RSU Award pursuant to Capitalization Adjustments as set forth in the Plan and the provisions of Section 3 below, if any, shall be subject, in a manner determined by the Board, to the same forfeiture restrictions, restrictions on transferability, and time and manner of delivery as applicable to the other Restricted Stock Units covered by your RSU Award.
- **3. DIVIDENDS.** You shall receive no benefit or adjustment to this RSU Award with respect to any cash dividend, stock dividend or other distribution that does not result from a Capitalization Adjustment; provided, however, that this sentence will not apply with respect to any shares of Common Stock that are delivered to you in connection with your RSU Award after such shares have been delivered to you.
- **4. WITHHOLDING OBLIGATIONS.** As further provided in Section 8 of the Plan, you hereby authorize withholding from payroll and any other amounts payable to you, and otherwise agree to make adequate provision for, any sums required to satisfy the federal, state, local and foreign tax withholding obligations, if any, which arise in connection with your RSU Award (the "*Withholding Obligation*") in accordance with the withholding procedures established by the Company. Unless the Withholding Obligation is satisfied, the Company shall have no obligation to deliver to you any Common Stock in respect of the RSU Award. In the event the Withholding Obligation of the Company arises prior to the delivery to you of Common Stock or it is determined after the delivery of Common Stock to you that the amount of the Withholding Obligation was greater than the amount withheld by the Company, you agree to indemnify and hold the Company harmless from any failure by the Company to withhold the proper amount.

## 5. DATE OF ISSUANCE.

- (a) The issuance of shares in respect of the Restricted Stock Units is intended to comply with Treasury Regulations Section 1.409A-1(b)(4) and will be construed and administered in such a manner. Subject to the satisfaction of the Withholding Obligation, if any, in the event one or more Restricted Stock Units vests, the Company shall issue to you one (1) share of Common Stock for each Restricted Stock Unit that vests on the applicable vesting date(s) (subject to any adjustment under Section 3 above, and subject to any different provisions in the Grant Notice). Each issuance date determined by this paragraph is referred to as an "*Original Issuance Date*."
- **(b)** If the Original Issuance Date falls on a date that is not a business day, delivery shall instead occur on the next following business day. In addition, <u>if</u>:
- (i) the Original Issuance Date does not occur (1) during an "open window period" applicable to you, as determined by the Company in accordance with the Company's then-effective policy on trading in Company securities, or (2) on a date when you are otherwise permitted to sell shares of Common Stock on an established stock exchange or stock market (including but not limited to under a previously established written trading plan that meets the requirements of Rule 10b5-1 under the Exchange Act and was entered into in compliance with the Company's policies (a "10b5-1 Arrangement)), and
- (ii) either (1) a Withholding Obligation does not apply, or (2) the Company decides, prior to the Original Issuance Date, (A) not to satisfy the Withholding Obligation by withholding shares of Common Stock from the shares otherwise due, on the Original Issuance Date, to you under this Award, and (B) not to permit you to enter into a "same day sale" commitment with a broker-dealer (including but not limited to a commitment under a 10b5-1 Arrangement) and (C) not to permit you to pay your Withholding Obligation in cash,

then the shares that would otherwise be issued to you on the Original Issuance Date will not be delivered on such Original Issuance Date and will instead be delivered on the first business day when you are not prohibited from selling shares of the Company's Common Stock in the open public market, but in no event later than December 31 of the calendar year in which the Original Issuance Date occurs (that is, the last day of your taxable year in which the Original Issuance Date occurs), or, if and only if permitted in a manner that complies with Treasury Regulations Section 1.409A-1(b)(4), no later than the date that is the 15th day of the third calendar month of the applicable year following the year in which the shares of Common Stock under this Award are no longer subject to a "substantial risk of forfeiture" within the meaning of Treasury Regulations Section 1.409A-1(d).

- (c) To the extent the RSU Award is a Non-Exempt RSU Award, the provisions of Section 11 of the Plan shall apply.
- **6. TRANSFERABILITY.** Except as otherwise provided in the Plan, your RSU Award is not transferable, except by will or by the applicable laws of descent and distribution.
- 7. **CORPORATE TRANSACTION.** Your RSU Award is subject to the terms of any agreement governing a Corporate Transaction involving the Company, including, without limitation, a provision for the appointment of a stockholder representative that is authorized to act on your behalf with respect to any escrow, indemnities and any contingent consideration.
- **8. NO LIABILITY FOR TAXES.** As a condition to accepting the RSU Award, you hereby (a) agree to not make any claim against the Company, or any of its Officers, Directors, Employees or Affiliates related to tax liabilities arising from the RSU Award or other Company compensation and (b) acknowledge that you were advised to consult with your own personal tax, financial and other legal advisors regarding the tax consequences of the RSU Award and have either done so or knowingly and voluntarily declined to do so.

- **9. SEVERABILITY**. If any part of this Agreement or the Plan is declared by any court or governmental authority to be unlawful or invalid, such unlawfulness or invalidity will not invalidate any portion of this Agreement or the Plan not declared to be unlawful or invalid. Any Section of this Agreement (or part of such a Section) so declared to be unlawful or invalid will, if possible, be construed in a manner which will give effect to the terms of such Section or part of a Section to the fullest extent possible while remaining lawful and valid.
- **10. OTHER DOCUMENTS.** You hereby acknowledge receipt of or the right to receive a document providing the information required by Rule 428(b)(1) promulgated under the Securities Act, which includes the Prospectus. In addition, you acknowledge receipt of the Company's Trading Policy.
- 11. QUESTIONS. If you have questions regarding these or any other terms and conditions applicable to your RSU Award, including a summary of the applicable federal income tax consequences please see the Prospectus.

Attachment II

2021 EQUITY INCENTIVE PLAN

1.

## ORIGIN MATERIALS, INC. STOCK OPTION GRANT NOTICE (2021 EQUITY INCENTIVE PLAN)

Origin Materials, Inc. (the "*Company*"), pursuant to the Company's 2021 Equity Incentive Plan (the "*Plan*"), has granted to you ("*Optionholder*") an option to purchase the number of shares of the Common Stock set forth below (the "*Option*"). Your Option is subject to all of the terms and conditions as set forth herein and in the Plan, and the Stock Option Agreement and the Notice of Exercise, all of which are attached hereto and incorporated herein in their entirety. Capitalized terms not explicitly defined herein but defined in the Plan or the Stock Option Agreement shall have the meanings set forth in the Plan or the Stock Option Agreement, as applicable.

Optionholder:

	Date of Grant:	
	Vesting Commencement Date:	
	Number of Shares of Common Stock Subject to Option:	
	Exercise Price (Per Share):	
	Total Exercise Price:	
	Expiration Date:	
Type of Grant:	[Incentive Stock Option] OR [Nonstatutory Stock Option]	
Exercise and Vesting Schedul	e: Subject to the Optionholder's Continuous Service through each applicable	le vesting date, the Option will vest as follows:
	[	

**Optionholder Acknowledgements:** By your signature below or by electronic acceptance or authentication in a form authorized by the Company, you understand and agree that:

- The Option is governed by this Stock Option Grant Notice (this "*Grant Notice*"), and the provisions of the Plan and the Stock Option Agreement and the Notice of Exercise, all of which are made a part of this document. Unless otherwise provided in the Plan, this Grant Notice and the Stock Option Agreement (together, the "*Option Agreement*") may not be modified, amended or revised except in a writing signed by you and a duly authorized officer of the Company.
- [If the Option is an Incentive Stock Option, it (plus other outstanding Incentive Stock Options granted to you) cannot be first *exercisable* for more than \$100,000 in value (measured by exercise price) in any calendar year. Any excess over \$100,000 is a Nonstatutory Stock Option.]

- You consent to receive this Grant Notice, the Stock Option Agreement, the Plan, the Prospectus and any other Plan-related documents by
  electronic delivery and to participate in the Plan through an on-line or electronic system established and maintained by the Company or
  another third party designated by the Company.
- You have read and are familiar with the provisions of the Plan, the Stock Option Agreement, the Notice of Exercise and the Prospectus. In the event of any conflict between the provisions in this Grant Notice, the Option Agreement, the Notice of Exercise, or the Prospectus and the terms of the Plan, the terms of the Plan shall control.
- The Option Agreement sets forth the entire understanding between you and the Company regarding the acquisition of Common Stock and supersedes all prior oral and written agreements, promises and/or representations on that subject with the exception of other equity awards previously granted to you and any written employment agreement, offer letter, severance agreement, written severance plan or policy, or other written agreement between the Company and you in each case that specifies the terms that should govern this Option.
- Counterparts may be delivered via facsimile, electronic mail (including pdf or any electronic signature complying with the U.S. federal ESIGN Act of 2000, Uniform Electronic Transactions Act or other applicable law) or other transmission method and any counterpart so delivered will be deemed to have been duly and validly delivered and be valid and effective for all purposes.

ORIGIN MATERIALS, INC.		OPTIONHOLDER:	
By:			
•	Signature	Signature	
Title:		Date:	
Date:		_	

ATTACHMENTS: Stock Option Agreement, 2021 Equity Incentive Plan, Notice of Exercise

### ATTACHMENT I

## ORIGIN MATERIALS, INC. STOCK OPTION AGREEMENT (2021 EQUITY INCENTIVE PLAN)

As reflected by your Stock Option Grant Notice ("*Grant Notice*"), Origin Materials, Inc. (the "*Company*") has granted you an option under the Company's 2021 Equity Incentive Plan (the "*Plan*") to purchase a number of shares of Common Stock at the exercise price indicated in your Grant Notice (the "*Option*"). Capitalized terms not explicitly defined in this Agreement but defined in the Grant Notice or the Plan shall have the meanings set forth in the Grant Notice or Plan, as applicable. The terms of your Option as specified in the Grant Notice and this Stock Option Agreement constitute your Option Agreement.

The general terms and conditions applicable to your Option are as follows:

- 1. GOVERNING PLAN DOCUMENT. Your Option is subject to all the provisions of the Plan, including but not limited to the provisions in:
  - (a) Section 6 regarding the impact of a Capitalization Adjustment, dissolution, liquidation, or Corporate Transaction on your Option;
  - (b) Section 9(e) regarding the Company's retained rights to terminate your Continuous Service notwithstanding the grant of the Option;
  - (c) Section 8 regarding the tax consequences of your Option.

Your Option is further subject to all interpretations, amendments, rules and regulations, which may from time to time be promulgated and adopted pursuant to the Plan. In the event of any conflict between the Option Agreement and the provisions of the Plan, the provisions of the Plan shall control.

## 2. EXERCISE.

and

- (a) You may generally exercise the vested portion of your Option for whole shares of Common Stock at any time during its term by delivery of payment of the exercise price and applicable withholding taxes and other required documentation to the Plan Administrator in accordance with the exercise procedures established by the Plan Administrator, which may include an electronic submission. Please review Sections 4(i), 4(j) and 7(b)(v) of the Plan, which may restrict or prohibit your ability to exercise your Option during certain periods.
  - (b) To the extent permitted by Applicable Law, you may pay your Option exercise price as follows:
    - (i) cash, check, bank draft or money order;

- (ii) subject to Company and/or Committee consent at the time of exercise, pursuant to a "cashless exercise" program as further described in Section 4(c)(ii) of the Plan if at the time of exercise the Common Stock is publicly traded;
- (iii) subject to Company and/or Committee consent at the time of exercise, by delivery of previously owned shares of Common Stock as further described in Section 4(c)(iii) of the Plan; or
- (iv) subject to Company and/or Committee consent at the time of exercise, if the Option is a Nonstatutory Stock Option, by a "net exercise" arrangement as further described in Section 4(c)(iv) of the Plan.
- (c) By accepting your Option, you agree that you will not sell, dispose of, transfer, make any short sale of, grant any option for the purchase of, or enter into any hedging or similar transaction with the same economic effect as a sale with respect to any shares of Common Stock or other securities of the Company held by you, for a period of one hundred eighty (180) days following the effective date of a registration statement of the Company filed under the Securities Act or such longer period as the underwriters or the Company will request to facilitate compliance with FINRA Rule 2241 or any successor or similar rules or regulation (the "Lock-Up Period"); provided, however, that nothing contained in this section will prevent the exercise of a repurchase option, if any, in favor of the Company during the Lock-Up Period. You further agree to execute and deliver such other agreements as may be reasonably requested by the Company or the underwriters that are consistent with the foregoing or that are necessary to give further effect thereto. In order to enforce the foregoing covenant, the Company may impose stop-transfer instructions with respect to your shares of Common Stock until the end of such period. You also agree that any transferee of any shares of Common Stock (or other securities) of the Company held by you will be bound by this Section 2(c). The underwriters of the Company's stock are intended third party beneficiaries of this Section 2(c) and will have the right, power and authority to enforce the provisions hereof as though they were a party hereto.
- **3. TERM.** You may not exercise your Option before the commencement of its term or after its term expires. The term of your Option commences on the Date of Grant and expires upon the earliest of the following:
  - (a) immediately upon the termination of your Continuous Service for Cause;
  - (b) three months after the termination of your Continuous Service for any reason other than Cause, Disability or death;
  - (c) 12 months after the termination of your Continuous Service due to your Disability;
  - (d) 18 months after your death if you die during your Continuous Service;
- (e) immediately upon a Corporate Transaction if the Board has determined that the Option will terminate in connection with a Corporate Transaction,
  - (f) the Expiration Date indicated in your Grant Notice; or
  - (g) the day before the 10th anniversary of the Date of Grant.

Notwithstanding the foregoing, if you die during the period provided in Section 3(b) or 3(c) above, the term of your Option shall not expire until the earlier of (i) 18 months after your death, (ii) upon any termination of the Option in connection with a Corporate Transaction, (iii) the Expiration Date indicated in your Grant Notice, or (iv) the day before the tenth anniversary of the Date of Grant. Additionally, the Post-Termination Exercise Period of your Option may be extended as provided in Section 4(i) of the Plan.

To obtain the federal income tax advantages associated with an Incentive Stock Option, the Code requires that at all times beginning on the date of grant of your Option and ending on the day three months before the date of your Option's exercise, you must be an employee of the Company or an Affiliate, except in the event of your death or Disability. If the Company provides for the extended exercisability of your Option under certain circumstances for your benefit, your Option will not necessarily be treated as an Incentive Stock Option if you exercise your Option more than three months after the date your employment terminates.

- **4. WITHHOLDING OBLIGATIONS.** As further provided in Section 8 of the Plan: (a) you may not exercise your Option unless the applicable tax withholding obligations are satisfied, and (b) at the time you exercise your Option, in whole or in part, or at any time thereafter as requested by the Company, you hereby authorize withholding from payroll and any other amounts payable to you, and otherwise agree to make adequate provision for (including by means of a "cashless exercise" pursuant to a program developed under Regulation T as promulgated by the Federal Reserve Board to the extent permitted by the Company), any sums required to satisfy the federal, state, local and foreign tax withholding obligations, if any, which arise in connection with the exercise of your Option in accordance with the withholding procedures established by the Company. Accordingly, you may not be able to exercise your Option even though the Option is vested, and the Company shall have no obligation to issue shares of Common Stock subject to your Option, unless and until such obligations are satisfied. In the event that the amount of the Company's withholding obligation in connection with your Option was greater than the amount actually withheld by the Company, you agree to indemnify and hold the Company harmless from any failure by the Company to withhold the proper amount.
- 5. INCENTIVE STOCK OPTION DISPOSITION REQUIREMENT. If your Option is an Incentive Stock Option, you must notify the Company in writing within 15 days after the date of any disposition of any of the shares of the Common Stock issued upon exercise of your Option that occurs within two years after the date of your Option grant or within one year after such shares of Common Stock are transferred upon exercise of your Option.
- **6. TRANSFERABILITY.** Except as otherwise provided in Section 4(e) of the Plan, your Option is not transferable, except by will or by the applicable laws of descent and distribution, and is exercisable during your life only by you.
- 7. **CORPORATE TRANSACTION.** Your Option is subject to the terms of any agreement governing a Corporate Transaction involving the Company, including, without limitation, a provision for the appointment of a stockholder representative that is authorized to act on your behalf with respect to any escrow, indemnities and any contingent consideration.

- **8. NO LIABILITY FOR TAXES.** As a condition to accepting the Option, you hereby (a) agree to not make any claim against the Company, or any of its Officers, Directors, Employees or Affiliates related to tax liabilities arising from the Option or other Company compensation and (b) acknowledge that you were advised to consult with your own personal tax, financial and other legal advisors regarding the tax consequences of the Option and have either done so or knowingly and voluntarily declined to do so. Additionally, you acknowledge that the Option is exempt from Section 409A only if the exercise price is at least equal to the "fair market value" of the Common Stock on the date of grant as determined by the Internal Revenue Service and there is no other impermissible deferral of compensation associated with the Option. Additionally, as a condition to accepting the Option, you agree not make any claim against the Company, or any of its Officers, Directors, Employees or Affiliates in the event that the Internal Revenue Service asserts that such exercise is less than the "fair market value" of the Common Stock on the date of grant as subsequently determined by the Internal Revenue Service.
- **9. SEVERABILITY.** If any part of this Option Agreement or the Plan is declared by any court or governmental authority to be unlawful or invalid, such unlawfulness or invalidity will not invalidate any portion of this Option Agreement or the Plan not declared to be unlawful or invalid. Any Section of this Option Agreement (or part of such a Section) so declared to be unlawful or invalid will, if possible, be construed in a manner which will give effect to the terms of such Section or part of a Section to the fullest extent possible while remaining lawful and valid
- **10. OTHER DOCUMENTS.** You hereby acknowledge receipt of or the right to receive a document providing the information required by Rule 428(b)(1) promulgated under the Securities Act, which includes the Prospectus. In addition, you acknowledge receipt of the Company's Trading Policy.
- 11. QUESTIONS. If you have questions regarding these or any other terms and conditions applicable to your Option, including a summary of the applicable federal income tax consequences please see the Prospectus.

\* \* \* \*

## ATTACHMENT II

2021 EQUITY INCENTIVE PLAN

### ATTACHMENT III

## ORIGIN MATERIALS, INC. NOTICE OF EXERCISE (2021 EQUITY INCENTIVE PLAN)

ORIGIN MATERIALS, INC. 930 RIVERSIDE PARKWAY SUITE 10 WEST SACRAMENTO, CA 95605

Date of Exercise:		

This constitutes notice to Origin Materials, Inc. (the "*Company*") that I elect to purchase the below number of shares of Common Stock of the Company (the "*Shares*") by exercising my Option for the price set forth below. Capitalized terms not explicitly defined in this Notice of Exercise but defined in the Stock Option Grant Notice, Stock Option Agreement or 2021 Equity Incentive Plan (the "*Plan*") shall have the meanings set forth in the Stock Option Grant Notice, Stock Option Agreement or Plan, as applicable. Use of certain payment methods is subject to Company and/or Committee consent and certain additional requirements set forth in the Stock Option Agreement and the Plan.

Type of option (check one):	Incentive $\square$	Nonstatutory L
Date of Grant:		
Number of Shares as to which Option is exercised:		
Certificates to be issued in name of:		
Total exercise price:	\$	
Cash, check, bank draft or money order delivered herewith:	\$	
Value of Shares delivered herewith:	\$	
Regulation T Program (cashless exercise)	\$	
Value of Shares pursuant to net exercise:	\$	

By this exercise, I agree (i) to provide such additional documents as you may require pursuant to the terms of the Plan, (ii) to satisfy the tax withholding obligations, if any, relating to the exercise of this Option as set forth in the Stock Option Agreement, and (iii) if this exercise relates to an incentive stock option, to notify you in writing within 15 days after the date of any disposition of any of the Shares issued upon exercise of this Option that occurs within two years after the Date of Grant or within one year after such Shares are issued upon exercise of this Option.

I further agree that, if required by the Company (or a representative of the underwriters) in connection with the first underwritten registration of the offering of any securities of the Company under the Securities Act, I will not sell, dispose of, transfer, make any short sale of, grant any option for the purchase of, or enter into any hedging or similar transaction with the same economic effect as a sale with respect to any shares of Common Stock or other securities of the Company for a period of one hundred eighty (180) days following the effective date of a registration statement of the Company filed under the Securities Act (or such longer period as the underwriters or the Company shall request to facilitate compliance with FINRA Rule 2241 or any successor or similar rule or regulation) (the "Lock-Up Period"). I further agree to execute and deliver such other agreements as may be reasonably requested by the Company or the underwriters that are consistent with the foregoing or that are necessary to give further effect thereto. In order to enforce the foregoing covenant, the Company may impose stop-transfer instructions with respect to securities subject to the foregoing restrictions until the end of such period.

te the foregoing covenant, the Company may impose stop-transfer instructions with respect to securities subject to the foregoing restrictions until	
d of such period.	
Very truly yours,	

## ORIGIN MATERIALS, INC. 2021 EMPLOYEE STOCK PURCHASE PLAN

ADOPTED BY THE BOARD OF DIRECTORS: MAY 25, 2021 APPROVED BY THE STOCKHOLDERS: JUNE 23, 2021

## 1. GENERAL; PURPOSE.

- (a) The Plan provides a means by which Eligible Employees of the Company and certain Designated Companies may be given an opportunity to purchase shares of Common Stock. The Plan permits the Company to grant a series of Purchase Rights to Eligible Employees under an Employee Stock Purchase Plan. In addition, the Plan permits the Company to grant a series of Purchase Rights to Eligible Employees that do not meet the requirements of an Employee Stock Purchase Plan.
- **(b)** The Plan includes two components: a 423 Component and a Non-423 Component. The Company intends (but makes no undertaking or representation to maintain) the 423 Component to qualify as an Employee Stock Purchase Plan. The provisions of the 423 Component, accordingly, will be construed in a manner that is consistent with the requirements of Section 423 of the Code. Except as otherwise provided in the Plan or determined by the Board, the Non-423 Component will operate and be administered in the same manner as the 423 Component.
- **(c)** The Company, by means of the Plan, seeks to retain the services of Eligible Employees, to secure and retain the services of new Employees and to provide incentives for such persons to exert maximum efforts for the success of the Company and its Related Corporations.

#### 2. ADMINISTRATION.

- (a) The Board or the Committee will administer the Plan. References herein to the Board shall be deemed to refer to the Committee except where context dictates otherwise.
  - **(b)** The Board will have the power, subject to, and within the limitations of, the express provisions of the Plan:
    - (i) To determine how and when Purchase Rights will be granted and the provisions of each Offering (which need not be identical).
- (ii) To designate from time to time (A) which Related Corporations will be eligible to participate in the Plan as Designated 423 Corporations, (B) which Related Corporations or Affiliates will be eligible to participate in the Plan as Designated Non-423 Corporations, and (C) which Designated Companies will participate in each separate Offering (to the extent that the Company makes separate Offerings).
- (iii) To construe and interpret the Plan and Purchase Rights, and to establish, amend and revoke rules and regulations for its administration. The Board, in the exercise of this power, may correct any defect, omission or inconsistency in the Plan, in a manner and to the extent it deems necessary or expedient to make the Plan fully effective.
  - (iv) To settle all controversies regarding the Plan and Purchase Rights granted under the Plan.
  - (v) To suspend or terminate the Plan at any time as provided in Section 12.

- (vi) To amend the Plan at any time as provided in Section 12.
- (vii) Generally, to exercise such powers and to perform such acts as it deems necessary or expedient to promote the best interests of the Company and its Related Corporations and to carry out the intent that the Plan be treated as an Employee Stock Purchase Plan with respect to the 423 Component.
- (viii) To adopt such rules, procedures and sub-plans as are necessary or appropriate to permit or facilitate participation in the Plan by Employees who are foreign nationals or employed or located outside the United States. Without limiting the generality of, and consistent with, the foregoing, the Board specifically is authorized to adopt rules, procedures, and sub-plans regarding, without limitation, eligibility to participate in the Plan, the definition of eligible "earnings," handling and making of Contributions, establishment of bank or trust accounts to hold Contributions, payment of interest, conversion of local currency, obligations to pay payroll tax, determination of beneficiary designation requirements, withholding procedures and handling of share issuances, any of which may vary according to applicable requirements, and which, if applicable to a Designated Non-423 Corporation, do not have to comply with the requirements of Section 423 of the Code.
- (c) The Board may delegate some or all of the administration of the Plan to a Committee or Committees. If administration is delegated to a Committee, the Committee will have, in connection with the administration of the Plan, the powers theretofore possessed by the Board that have been delegated to the Committee, including the power to delegate to a subcommittee any of the administrative powers the Committee is authorized to exercise (and references in this Plan to the Board will thereafter be to the Committee or subcommittee), subject, however, to such resolutions, not inconsistent with the provisions of the Plan, as may be adopted from time to time by the Board. Further, to the extent not prohibited by Applicable Law, the Board or Committee may, from time to time, delegate some or all of its authority under the Plan to one or more officers of the Company or other persons or groups of persons as it deems necessary, appropriate or advisable under conditions or limitations that it may set at or after the time of the delegation. The Board may retain the authority to concurrently administer the Plan with the Committee and may, at any time, revest in the Board some or all of the powers previously delegated. Whether or not the Board has delegated administration of the Plan to a Committee, the Board will have the final power to determine all questions of policy and expediency that may arise in the administration of the Plan.
- **(d)** All determinations, interpretations and constructions made by the Board in good faith will not be subject to review by any person and will be final, binding and conclusive on all persons.

### 3. SHARES OF COMMON STOCK SUBJECT TO THE PLAN.

(a) Subject to the provisions of Section 11(a) relating to Capitalization Adjustments, the maximum number of shares of Common Stock that may be issued under the Plan will not exceed 1,846,710 shares of Common Stock, plus the number of shares of Common Stock that are automatically added on January 1st of each year for a period of ten years commencing on January 1, 2022 and ending on (and including) January 1, 2031, in an amount equal to the lesser of (i) one percent (1%) of the Fully-Diluted Common Stock on December 31st of the preceding calendar year, and (ii) 3,693,420 shares of Common Stock. Notwithstanding the foregoing, the Board may act prior to the first day of any calendar year to provide that there will be no January 1st increase in the share reserve for such calendar year or that the increase in the share reserve for such calendar year will be a lesser number of shares of Common Stock than would otherwise occur pursuant to the preceding sentence. For the avoidance of doubt, up to the maximum number of shares of Common Stock reserved under this Section 3(a) may be used to satisfy purchases of Common Stock under the A23 Component and any remaining portion of such maximum number of shares may be used to satisfy purchases of Common Stock under the Non-423 Component.

- **(b)** If any Purchase Right granted under the Plan terminates without having been exercised in full, the shares of Common Stock not purchased under such Purchase Right will again become available for issuance under the Plan.
- **(c)** The stock purchasable under the Plan will be shares of authorized but unissued or reacquired Common Stock, including shares repurchased by the Company on the open market.

## 4. GRANT OF PURCHASE RIGHTS; OFFERING.

- (a) The Board may from time to time grant or provide for the grant of Purchase Rights to Eligible Employees under an Offering (consisting of one or more Purchase Periods) on an Offering Date or Offering Dates selected by the Board. Each Offering will be in such form and will contain such terms and conditions as the Board will deem appropriate, and, with respect to the 423 Component, will comply with the requirement of Section 423(b) (5) of the Code that all Employees granted Purchase Rights will have the same rights and privileges. The terms and conditions of an Offering shall be incorporated by reference into the Plan and treated as part of the Plan. The provisions of separate Offerings need not be identical, but each Offering will include (through incorporation of the provisions of this Plan by reference in the document comprising the Offering or otherwise) the period during which the Offering will be effective, which period will not exceed 27 months beginning with the Offering Date, and the substance of the provisions contained in Sections 5 through 8, inclusive.
- **(b)** If a Participant has more than one Purchase Right outstanding under the Plan, unless he or she otherwise indicates in forms delivered to the Company or a third party designated by the Company (each, a "*Company Designee*"): (i) each form will apply to all of his or her Purchase Rights under the Plan, and (ii) a Purchase Right with a lower exercise price (or an earlier-granted Purchase Right, if different Purchase Rights have identical exercise prices) will be exercised to the fullest possible extent before a Purchase Right with a higher exercise price (or a later-granted Purchase Right if different Purchase Rights have identical exercise prices) will be exercised.
- (c) The Board will have the discretion to structure an Offering so that if the Fair Market Value of a share of Common Stock on the first Trading Day of a new Purchase Period within that Offering is less than or equal to the Fair Market Value of a share of Common Stock on the Offering Date for that Offering, then (i) that Offering will terminate immediately as of that first Trading Day, and (ii) the Participants in such terminated Offering will be automatically enrolled in a new Offering beginning on the first Trading Day of such new Purchase Period.

### 5. ELIGIBILITY.

(a) Purchase Rights may be granted only to Employees of the Company or, as the Board may designate in accordance with Section 2(b), to Employees of a Related Corporation or an Affiliate. Except as provided in Section 5(b) or as required by Applicable Law, an Employee will not be eligible to be granted Purchase Rights unless, on the Offering Date, the Employee has been in the employ of the Company, a Related Corporation or an Affiliate, as the case may be, for such continuous period preceding such Offering Date as the Board may require, but in no event will the required period of continuous employment be equal to or greater than two years. In addition, the Board may (unless prohibited by Applicable Law) provide that no Employee will be eligible to be granted Purchase Rights under the Plan unless, on the Offering Date, such Employee's customary employment with the Company, the Related Corporation, or the Affiliate is more than 20 hours per week and more than five months per calendar year

or such other criteria as the Board may determine consistent with Section 423 of the Code with respect to the 423 Component. The Board may also exclude from participation in the Plan or any Offering Employees who are "highly compensated employees" (within the meaning of Section 423(b)(4) (D) of the Code) of the Company or a Related Corporation or a subset of such highly compensated employees.

- **(b)** The Board may provide that each person who, during the course of an Offering, first becomes an Eligible Employee will, on a date or dates specified in the Offering which coincides with the day on which such person becomes an Eligible Employee or which occurs thereafter, receive a Purchase Right under that Offering, which Purchase Right will thereafter be deemed to be a part of that Offering. Such Purchase Right will have the same characteristics as any Purchase Rights originally granted under that Offering, as described herein, except that:
- (i) the date on which such Purchase Right is granted will be the "Offering Date" of such Purchase Right for all purposes, including determination of the exercise price of such Purchase Right;
- (ii) the period of the Offering with respect to such Purchase Right will begin on its Offering Date and end coincident with the end of such Offering; and
- (iii) the Board may provide that if such person first becomes an Eligible Employee within a specified period of time before the end of the Offering, he or she will not receive any Purchase Right under that Offering.
- (c) No Employee will be eligible for the grant of any Purchase Rights under the 423 Component if, immediately after any such Purchase Rights are granted, such Employee owns stock possessing five percent or more of the total combined voting power or value of all classes of stock of the Company or of any Related Corporation. For purposes of this Section 5(c), the rules of Section 424(d) of the Code will apply in determining the stock ownership of any Employee, and stock which such Employee may purchase under all outstanding Purchase Rights and options will be treated as stock owned by such Employee.
- (d) As specified by Section 423(b)(8) of the Code, an Eligible Employee may be granted Purchase Rights under the 423 Component only if such Purchase Rights, together with any other rights granted under all Employee Stock Purchase Plans of the Company and any Related Corporations, do not permit such Eligible Employee's rights to purchase stock of the Company or any Related Corporation to accrue at a rate which, when aggregated, exceeds US \$25,000 of Fair Market Value of such stock (determined at the time such rights are granted, and which, with respect to the Plan, will be determined as of their respective Offering Dates) for each calendar year in which such rights are outstanding at any time.
- **(e)** Officers of the Company and any Designated Company, if they are otherwise Eligible Employees, will be eligible to participate in Offerings under the Plan. Notwithstanding the foregoing, the Board may (unless prohibited by Applicable Law) provide in an Offering that Employees who are highly compensated Employees within the meaning of Section 423(b)(4)(D) of the Code will not be eligible to participate.
- **(f)** Notwithstanding anything in this Section 5 to the contrary, in the case of an Offering under the Non-423 Component, an Eligible Employee (or group of Eligible Employees) may be excluded from participation in the Plan or an Offering if the Board has determined, in its sole discretion, that participation of such Eligible Employee(s) is not advisable or practical for any reason.

## 6. PURCHASE RIGHTS; PURCHASE PRICE.

- (a) On each Offering Date, each Eligible Employee, pursuant to an Offering made under the Plan, will be granted a Purchase Right to purchase up to that number of shares of Common Stock purchasable either with a percentage or with a maximum dollar amount, as designated by the Board, during the period that begins on the Offering Date (or such later date as the Board determines for a particular Offering) and ends on the date stated in the Offering, which date will be no later than the end of the Offering.
- **(b)** The Board will establish one or more Purchase Dates during an Offering on which Purchase Rights granted for that Offering will be exercised and shares of Common Stock will be purchased in accordance with such Offering.
- (c) In connection with each Offering made under the Plan, the Board may specify (i) a maximum number of shares of Common Stock that may be purchased by any Participant on any Purchase Date during such Offering, (ii) a maximum aggregate number of shares of Common Stock that may be purchased by all Participants pursuant to such Offering and/or (iii) a maximum aggregate number of shares of Common Stock that may be purchased by all Participants on any Purchase Date under the Offering. If the aggregate purchase of shares of Common Stock issuable upon exercise of Purchase Rights granted under the Offering would exceed any such maximum aggregate number, then, in the absence of any Board action otherwise, a pro rata (based on each Participant's accumulated Contributions) allocation of the shares of Common Stock (rounded down to the nearest whole share) available will be made in as nearly a uniform manner as will be practicable and equitable.
- **(d)** The purchase price of shares of Common Stock acquired pursuant to Purchase Rights will be specified by Board prior to the commencement of an Offering and will not be less than the lesser of:
  - (i) an amount equal to 85% of the Fair Market Value of the shares of Common Stock on the Offering Date; or
  - (ii) an amount equal to 85% of the Fair Market Value of the shares of Common Stock on the applicable Purchase Date.

### 7. PARTICIPATION; WITHDRAWAL; TERMINATION.

(a) An Eligible Employee may elect to participate in an Offering and authorize payroll deductions as the means of making Contributions by completing and delivering to the Company or a Company Designee, within the time specified for the Offering, an enrollment form provided by the Company or Company Designee. The enrollment form will specify the amount of Contributions not to exceed the maximum amount specified by the Board. Each Participant's Contributions will be credited to a bookkeeping account for such Participant under the Plan and will be deposited with the general funds of the Company except where Applicable Law requires that Contributions be deposited with a third party. If permitted in the Offering, a Participant may begin such Contributions with the first payroll occurring on or after the Offering Date (or, in the case of a payroll date that occurs after the end of the prior Offering but before the Offering Date of the next new Offering, Contributions from such payroll will be included in the new Offering). If permitted in the Offering, a Participant may thereafter reduce (including to zero) or increase his or her Contributions. If required under Applicable Law or if specifically provided in the Offering and to extent permitted by Section 423 of the Code with respect to the 423 Component, in addition to or instead of making Contributions by payroll deductions, a Participant may make Contributions through payment by cash, check or wire transfer prior to a Purchase Date.

- **(b)** During an Offering, a Participant may cease making Contributions and withdraw from the Offering by delivering to the Company or a Company Designee a withdrawal form provided by the Company. The Company may impose a deadline before a Purchase Date for withdrawing. Upon such withdrawal, such Participant's Purchase Right in that Offering will immediately terminate and the Company will distribute as soon as practicable to such Participant all of his or her accumulated but unused Contributions and such Participant's Purchase Right in that Offering shall thereupon terminate. A Participant's withdrawal from that Offering will have no effect upon his or her eligibility to participate in any other Offerings under the Plan, but such Participant will be required to deliver a new enrollment form to participate in subsequent Offerings.
- (c) Unless otherwise required by Applicable Law, Purchase Rights granted pursuant to any Offering under the Plan will terminate immediately if the Participant either (i) is no longer an Employee for any reason or for no reason (subject to any post-employment participation period required by Applicable Law) or (ii) is otherwise no longer eligible to participate. The Company will distribute as soon as practicable to such individual all of his or her accumulated but unused Contributions.
- (d) Unless otherwise determined by the Board, a Participant whose employment transfers or whose employment terminates with an immediate rehire (with no break in service) by or between the Company and a Designated Company or between Designated Companies will not be treated as having terminated employment for purposes of participating in the Plan or an Offering; however, if a Participant transfers from an Offering under the 423 Component to an Offering under the Non-423 Component, the exercise of the Participant transfers from an Offering under the Non-423 Component to an Offering under the 423 Component, the exercise of the Purchase Right will remain non-qualified under the Non-423 Component. The Board may establish different and additional rules governing transfers between separate Offerings within the 423 Component and between Offerings under the 423 Component and Offerings under the Non-423 Component.
- **(e)** During a Participant's lifetime, Purchase Rights will be exercisable only by such Participant. Purchase Rights are not transferable by a Participant, except by will, by the laws of descent and distribution, or, if permitted by the Company, by a beneficiary designation as described in Section 10.
- **(f)** Unless otherwise specified in the Offering or as required by Applicable Law, the Company will have no obligation to pay interest on Contributions.

#### 8. EXERCISE OF PURCHASE RIGHTS.

- (a) On each Purchase Date, each Participant's accumulated Contributions will be applied to the purchase of shares of Common Stock, up to the maximum number of shares of Common Stock permitted by the Plan and the applicable Offering, at the purchase price specified in the Offering. No fractional shares will be issued unless specifically provided for in the Offering.
- **(b)** Unless otherwise provided in the Offering, if any amount of accumulated Contributions remains in a Participant's account after the purchase of shares of Common Stock on the final Purchase Date of an Offering, then such remaining amount will not roll over to the next Offering and will instead be distributed in full to such Participant after the final Purchase Date of such Offering without interest (unless otherwise required by Applicable Law).
- (c) No Purchase Rights may be exercised to any extent unless the shares of Common Stock to be issued upon such exercise under the Plan are covered by an effective registration statement pursuant to the Securities Act and the Plan is in material compliance with all applicable U.S. federal and state,

foreign and other securities, exchange control and other laws applicable to the Plan. If on a Purchase Date the shares of Common Stock are not so registered or the Plan is not in such compliance, no Purchase Rights will be exercised on such Purchase Date, and, subject to Section 423 of the Code with respect to the 423 Component, the Purchase Date will be delayed until the shares of Common Stock are subject to such an effective registration statement and the Plan is in material compliance, except that the Purchase Date will in no event be more than 27 months from the Offering Date. If, on the Purchase Date, as delayed to the maximum extent permissible, the shares of Common Stock are not registered and the Plan is not in material compliance with all Applicable Laws, as determined by the Company in its sole discretion, no Purchase Rights will be exercised and all accumulated but unused Contributions will be distributed to the Participants without interest (unless the payment of interest is otherwise required by Applicable Law).

### 9. COVENANTS OF THE COMPANY.

The Company will seek to obtain from each U.S. federal or state, foreign or other regulatory commission, agency or other Governmental Body having jurisdiction over the Plan such authority as may be required to grant Purchase Rights and issue and sell shares of Common Stock thereunder unless the Company determines, in its sole discretion, that doing so is not practical or would cause the Company to incur costs that are unreasonable. If, after commercially reasonable efforts, the Company is unable to obtain the authority that counsel for the Company deems necessary for the grant of Purchase Rights or the lawful issuance and sale of Common Stock under the Plan, and at a commercially reasonable cost, the Company will be relieved from any liability for failure to grant Purchase Rights and/or to issue and sell Common Stock upon exercise of such Purchase Rights.

## 10. DESIGNATION OF BENEFICIARY.

- (a) The Company may, but is not obligated to, permit a Participant to submit a form designating a beneficiary who will receive any shares of Common Stock and/or Contributions from the Participant's account under the Plan if the Participant dies before such shares and/or Contributions are delivered to the Participant. The Company may, but is not obligated to, permit the Participant to change such designation of beneficiary. Any such designation and/or change must be on a form approved by the Company.
- **(b)** If a Participant dies, and in the absence of a valid beneficiary designation, the Company will deliver any shares of Common Stock and/or Contributions to the executor or administrator of the estate of the Participant. If no executor or administrator has been appointed (to the knowledge of the Company), the Company, in its sole discretion, may deliver such shares of Common Stock and/or Contributions, without interest (unless the payment of interest is otherwise required by Applicable Law), to the Participant's spouse, dependents or relatives, or if no spouse, dependent or relative is known to the Company, then to such other person as the Company may designate.

## 11. ADJUSTMENTS UPON CHANGES IN COMMON STOCK; CORPORATE TRANSACTIONS.

(a) In the event of a Capitalization Adjustment, the Board will appropriately and proportionately adjust: (i) the class(es) and maximum number of securities subject to the Plan pursuant to Section 3(a), (ii) the class(es) and maximum number of securities by which the share reserve is to increase automatically each year pursuant to Section 3(a), (iii) the class(es) and number of securities subject to, and the purchase price applicable to outstanding Offerings and Purchase Rights, and (iv) the class(es) and number of securities that are the subject of the purchase limits under each ongoing Offering. The Board will make these adjustments, and its determination will be final, binding and conclusive.

**(b)** In the event of a Corporate Transaction, then: (i) any surviving corporation or acquiring corporation (or the surviving or acquiring corporation's parent company) may assume or continue outstanding Purchase Rights or may substitute similar rights (including a right to acquire the same consideration paid to the stockholders in the Corporate Transaction) for outstanding Purchase Rights, or (ii) if any surviving or acquiring corporation (or its parent company) does not assume or continue such Purchase Rights or does not substitute similar rights for such Purchase Rights, then the Participants' accumulated Contributions will be used to purchase shares of Common Stock (rounded down to the nearest whole share) within ten business days (or such other period specified by the Board) prior to the Corporate Transaction under the outstanding Purchase Rights, and the Purchase Rights will terminate immediately after such purchase.

### 12. AMENDMENT, TERMINATION OR SUSPENSION OF THE PLAN.

- (a) The Board may amend the Plan at any time in any respect the Board deems necessary or advisable. However, except as provided in Section 11(a) relating to Capitalization Adjustments, stockholder approval will be required for any amendment of the Plan for which stockholder approval is required by Applicable Law.
- **(b)** The Board may suspend or terminate the Plan at any time. No Purchase Rights may be granted under the Plan while the Plan is suspended or after it is terminated.

Any benefits, privileges, entitlements and obligations under any outstanding Purchase Rights granted before an amendment, suspension or termination of the Plan will not be materially impaired by any such amendment, suspension or termination except (i) with the consent of the person to whom such Purchase Rights were granted, (ii) as necessary to facilitate compliance with any laws, listing requirements, or governmental regulations (including, without limitation, the provisions of Section 423 of the Code and the regulations and other interpretive guidance issued thereunder relating to Employee Stock Purchase Plans) including without limitation any such regulations or other guidance that may be issued or amended after the date the Plan is adopted by the Board, or (iii) as necessary to obtain or maintain favorable tax, listing, or regulatory treatment. To be clear, the Board may amend outstanding Purchase Rights without a Participant's consent if such amendment is necessary to ensure that the Purchase Right and/or the Plan complies with the requirements of Section 423 of the Code with respect to the 423 Component or with respect to other Applicable Laws, Notwithstanding anything in the Plan or any Offering Document to the contrary, the Board will be entitled to: (i) establish the exchange ratio applicable to amounts withheld in a currency other than U.S. dollars; (ii) permit Contributions in excess of the amount designated by a Participant in order to adjust for mistakes in the Company's processing of properly completed Contribution elections; (iii) establish reasonable waiting and adjustment periods and/or accounting and crediting procedures to ensure that amounts applied toward the purchase of Common Stock for each Participant properly correspond with amounts withheld from the Participant's Contributions; (iv) amend any outstanding Purchase Rights or clarify any ambiguities regarding the terms of any Offering to enable the Purchase Rights to qualify under and/or comply with Section 423 of the Code with respect to the 423 Component; and (v) establish other limitations or procedures as the Board determines in its sole discretion advisable that are consistent with the Plan. The actions of the Board pursuant to this paragraph will not be considered to alter or impair any Purchase Rights granted under an Offering as they are part of the initial terms of each Offering and the Purchase Rights granted under each Offering.

## 13. TAX QUALIFICATION; TAX WITHHOLDING.

(a) Although the Company may endeavor to (i) qualify a Purchase Right for special tax treatment under the laws of the United States or jurisdictions outside of the United States or (ii) avoid

adverse tax treatment, the Company makes no representation to that effect and expressly disavows any covenant to maintain special or to avoid unfavorable tax treatment, notwithstanding anything to the contrary in this Plan. The Company will be unconstrained in its corporate activities without regard to the potential negative tax impact on Participants.

- **(b)** Each Participant will make arrangements, satisfactory to the Company and any applicable Related Corporation, to enable the Company or the Related Corporation to fulfill any withholding obligation for Tax-Related Items. Without limitation to the foregoing, in the Company's sole discretion and subject to Applicable Law, such withholding obligation may be satisfied in whole or in part by (i) withholding from the Participant's salary or any other cash payment due to the Participant from the Company or a Related Corporation; (ii) withholding from the proceeds of the sale of shares of Common Stock acquired under the Plan, either through a voluntary sale or a mandatory sale arranged by the Company; or (iii) any other method deemed acceptable by the Board. The Company shall not be required to issue any shares of Common Stock under the Plan until such obligations are satisfied.
- (c) The 423 Component is exempt from the application of Section 409A of the Code, and any ambiguities herein shall be interpreted to so be exempt from Section 409A of the Code. The Non-423 Component is intended to be exempt from the application of Section 409A of the Code under the short-term deferral exception and any ambiguities shall be construed and interpreted in accordance with such intent. In furtherance of the foregoing and notwithstanding any provision in the Plan to the contrary, if the Committee determines that an option granted under the Plan may be subject to Section 409A of the Code or that any provision in the Plan would cause an option under the Plan to be subject to Section 409A, the Committee may amend the terms of the Plan and/or of an outstanding option granted under the Plan, or take such other action the Committee determines is necessary or appropriate, in each case, without the participant's consent, to exempt any outstanding option or future option that may be granted under the Plan from or to allow any such options to comply with Section 409A of the Code, but only to the extent any such amendments or action by the Committee would not violate Section 409A of the Code. Notwithstanding the foregoing, the Company shall have no liability to a participant or any other party if the option under the Plan that is intended to be exempt from or compliant with Section 409A of the Code is not so exempt or compliant or for any action taken by the Committee with respect thereto.

### 14. EFFECTIVE DATE OF PLAN.

The Plan will become effective immediately prior to and contingent upon the Effective Date. No Purchase Rights will be exercised unless and until the Plan has been approved by the stockholders of the Company, which approval must be within 12 months before or after the date the Plan is adopted (or if required under Section 12(a) above, materially amended) by the Board.

#### 15. MISCELLANEOUS PROVISIONS.

- (a) Proceeds from the sale of shares of Common Stock pursuant to Purchase Rights will constitute general funds of the Company.
- **(b)** A Participant will not be deemed to be the holder of, or to have any of the rights of a holder with respect to, shares of Common Stock subject to Purchase Rights unless and until the Participant's shares of Common Stock acquired upon exercise of Purchase Rights are recorded in the books of the Company (or its transfer agent).
- (c) The Plan and Offering do not constitute an employment contract. Nothing in the Plan or in the Offering will in any way alter the at will nature of a Participant's employment or amend a Participant's employment contract, if applicable, or be deemed to create in any way whatsoever any obligation on the part of any Participant to continue in the employ of the Company or a Related Corporation or an Affiliate, or on the part of the Company, a Related Corporation or an Affiliate to continue the employment of a Participant.

- (d) The provisions of the Plan will be governed by the laws of the State of Delaware without resort to that state's conflicts of laws rules.
- **(e)** If any particular provision of the Plan is found to be invalid or otherwise unenforceable, such provision will not affect the other provisions of the Plan, but the Plan will be construed in all respects as if such invalid provision were omitted.
- **(f)** If any provision of the Plan does not comply with Applicable Law, such provision shall be construed in such a manner as to comply with Applicable Law.

#### 16. DEFINITIONS.

As used in the Plan, the following definitions will apply to the capitalized terms indicated below:

- **(a)** "423 Component" means the part of the Plan, which excludes the Non-423 Component, pursuant to which Purchase Rights that satisfy the requirements for an Employee Stock Purchase Plan may be granted to Eligible Employees.
- **(b)** "Affiliate" means any entity, other than a Related Corporation, whether now or subsequently established, which is at the time of determination, a "parent" or "subsidiary" of the Company as such terms are defined in Rule 405 promulgated under the Securities Act. The Board may determine the time or times at which "parent" or "subsidiary" status is determined within the foregoing definition.
- (c) "Applicable Law" means shall mean the Code and any applicable securities, federal, state, foreign, material local or municipal or other law, statute, constitution, principle of common law, resolution, ordinance, code, edict, decree, rule, listing rule, regulation, judicial decision, ruling or requirement issued, enacted, adopted, promulgated, implemented or otherwise put into effect by or under the authority of any Governmental Body (or under the authority of the NASDAQ Stock Market, the New York Stock Exchange or the Financial Industry Regulatory Authority).
  - (d) "Board" means the board of directors of the Company.
- **(e)** "Capitalization Adjustment" means any change that is made in, or other events that occur with respect to, the Common Stock subject to the Plan or subject to any Purchase Right after the date the Plan is adopted by the Board without the receipt of consideration by the Company through merger, consolidation, reorganization, recapitalization, reincorporation, stock dividend, dividend in property other than cash, large nonrecurring cash dividend, stock split, liquidating dividend, combination of shares, exchange of shares, change in corporate structure or other similar equity restructuring transaction, as that term is used in Financial Accounting Standards Board Accounting Standards Codification Topic 718 (or any successor thereto). Notwithstanding the foregoing, the conversion of any convertible securities of the Company will not be treated as a Capitalization Adjustment.
  - (f) "Code" means the U.S. Internal Revenue Code of 1986, as amended, including any applicable regulations and guidance thereunder.

- **(g)** "Committee" means a committee of one or more members of the Board to whom authority has been delegated by the Board in accordance with Section 2(c).
  - **(h)** "Common Stock" means the Class A common stock of the Company.
  - (i) "Company" means Origin Materials, Inc., a Delaware corporation.
- **(j)** "Contributions" means the payroll deductions and other additional payments specifically provided for in the Offering that a Participant contributes to fund the exercise of a Purchase Right. A Participant may make additional payments into his or her account if specifically provided for in the Offering, and then only if the Participant has not already had the maximum permitted amount withheld during the Offering through payroll deductions and, with respect to the 423 Component, to the extent permitted by Section 423 of the Code.
- **(k)** "Corporate Transaction" means the consummation, in a single transaction or in a series of related transactions, of any one or more of the following events:
- (i) a sale or other disposition of all or substantially all, as determined by the Board in its sole discretion, of the consolidated assets of the Company and its subsidiaries;
  - (ii) a sale or other disposition of more than 50% of the outstanding securities of the Company;
  - (iii) a merger, consolidation or similar transaction following which the Company is not the surviving corporation; or
- (iv) a merger, consolidation or similar transaction following which the Company is the surviving corporation but the shares of Common Stock outstanding immediately preceding the merger, consolidation or similar transaction are converted or exchanged by virtue of the merger, consolidation or similar transaction into other property, whether in the form of securities, cash or otherwise.
  - (I) "Designated 423 Corporation" means any Related Corporation selected by the Board to participate in the 423 Component.
- **(m)** "Designated Company" means any Designated Non-423 Corporation or Designated 423 Corporation, provided, however, that at any given time, a Related Corporation participating in the 423 Component shall not be a Related Corporation participating in the Non-423 Component.
- **(n)** "*Designated Non-423 Corporation*" means any Related Corporation or Affiliate selected by the Board to participate in the Non-423 Component.
  - **(o)** "*Director*" means a member of the Board.
- **(p)** "Effective Date" means the effective date of this Plan, which is the date of the closing of the transactions contemplated by the Agreement and Plan of Merger by and among Artius Acquisition Inc., Zero Carbon Merger Sub Inc. and Micromidas, Inc., dated as of February 16, 2021, provided that this Plan is approved by the Company's stockholders prior to such date.

- (q) "Eligible Employee" means an Employee who meets the requirements set forth in the document(s) governing the Offering for eligibility to participate in the Offering, provided that such Employee also meets the requirements for eligibility to participate set forth in the Plan.
- **(r)** "*Employee*" means any person, including an Officer or Director, who is "employed" for purposes of Section 423(b)(4) of the Code by the Company or a Related Corporation, or solely with respect to the Non-423 Component, an Affiliate. However, service solely as a Director, or payment of a fee for such services, will not cause a Director to be considered an "Employee" for purposes of the Plan.
- **(s)** "Employee Stock Purchase Plan" means a plan that grants Purchase Rights intended to be options issued under an "employee stock purchase plan," as that term is defined in Section 423(b) of the Code.
  - (t) "Exchange Act" means the U.S. Securities Exchange Act of 1934, as amended and the rules and regulations promulgated thereunder.
  - (u) "Fair Market Value" means, as of any date, the value of the Common Stock determined as follows:
- (i) If the Common Stock is listed on any established stock exchange or traded on any established market, the Fair Market Value of a share of Common Stock will be the closing sales price for such stock as quoted on such exchange or market (or the exchange or market with the greatest volume of trading in the Common Stock) on the date of determination, as reported in such source as the Board deems reliable. Unless otherwise provided by the Board, if there is no closing sales price for the Common Stock on the date of determination, then the Fair Market Value will be the closing sales price on the last preceding date for which such quotation exists.
- (ii) In the absence of such markets for the Common Stock, the Fair Market Value will be determined by the Board in good faith in compliance with Applicable Laws and regulations and, to the extent applicable as determined in the sole discretion of the Board, in a manner that complies with Sections 409A of the Code
- (v) "Fully-Diluted Common Stock" means, as of any date, the aggregate number of (i) shares of Common Stock issued and outstanding and (ii) securities convertible into or exercisable for shares of Common Stock (whether vested or unvested).
- (w) "Governmental Body" means any: (a) nation, state, commonwealth, province, territory, county, municipality, district or other jurisdiction of any nature; (b) federal, state, local, municipal, foreign or other government; (c) governmental or regulatory body, or quasi-governmental body of any nature (including any governmental division, department, administrative agency or bureau, commission, authority, instrumentality, official, ministry, fund, foundation, center, organization, unit, body or entity and any court or other tribunal, and for the avoidance of doubt, any tax authority) or other body exercising similar powers or authority; or (d) self-regulatory organization (including the NASDAQ Stock Market, the New York Stock Exchange and the Financial Industry Regulatory Authority).
- (x) "Non-423 Component" means the part of the Plan, which excludes the 423 Component, pursuant to which Purchase Rights that are not intended to satisfy the requirements for an Employee Stock Purchase Plan may be granted to Eligible Employees.
- **(y)** "Offering" means the grant to Eligible Employees of Purchase Rights, with the exercise of those Purchase Rights automatically occurring at the end of one or more Purchase Periods. The terms and conditions of an Offering will generally be set forth in the "Offering Document" approved by the Board for that Offering.

- (z) "Offering Date" means a date selected by the Board for an Offering to commence.
- (aa) "Officer" means a person who is an officer of the Company or a Related Corporation within the meaning of Section 16 of the Exchange Act.
  - (bb) "Participant" means an Eligible Employee who holds an outstanding Purchase Right.
- (cc) "Plan" means this Origin Materials, Inc. 2021 Employee Stock Purchase Plan, as amended from time to time, including both the 423 Component and the Non-423 Component.
- **(dd)** "*Purchase Date*" means one or more dates during an Offering selected by the Board on which Purchase Rights will be exercised and on which purchases of shares of Common Stock will be carried out in accordance with such Offering.
- **(ee)** "*Purchase Period*" means a period of time specified within an Offering, generally beginning on the Offering Date or on the first Trading Day following a Purchase Date, and ending on a Purchase Date. An Offering may consist of one or more Purchase Periods.
  - (ff) "Purchase Right" means an option to purchase shares of Common Stock granted pursuant to the Plan.
- **(gg)** "Related Corporation" means any "parent corporation" or "subsidiary corporation" of the Company whether now or subsequently established, as those terms are defined in Sections 424(e) and (f), respectively, of the Code.
  - (hh) "Securities Act" means the U.S. Securities Act of 1933, as amended.
- (ii) "Tax-Related Items" means any income tax, social insurance, payroll tax, fringe benefit tax, payment on account or other tax-related items arising out of or in relation to a Participant's participation in the Plan, including, but not limited to, the exercise of a Purchase Right and the receipt of shares of Common Stock or the sale or other disposition of shares of Common Stock acquired under the Plan.
- (jj) "Trading Day" means any day on which the exchange(s) or market(s) on which shares of Common Stock are listed, including but not limited to the New York Stock Exchange, Nasdaq Global Select Market, the Nasdaq Global Market, the Nasdaq Capital Market or any successors thereto, is open for trading.

July 1, 2021

Securities and Exchange Commission 100 F Street, N.E. Washington, DC 20549

## Commissioners:

We have read the statements made by Origin Materials, Inc. (formerly Artius Acquisition Inc.), under Item 4.01 of its Form 8-K filed July 1, 2021. We agree with the statements concerning our Firm under Item 4.01, in which we were informed of our dismissal on June 25, 2021, effective immediately. We are not in a position to agree or disagree with other statements of Origin Materials, Inc. (formerly Artius Acquisition Inc.) contained therein.

Very truly yours,

/s/ Marcum LLP

Marcum LLP

## ORIGIN MATERIALS, INC.

List of Subsidiaries

Subsidiary	Jurisdiction
Micromidas, Inc.	Delaware
Origin Materials Canada Holding Limited	New Brunswick
Origin Materials Canada Pioneer Limited	New Brunswick
Origin Materials Canada Polyesters Limited	New Brunswick
Origin Materials Canada Research Limited	New Brunswick



Origin Materials, Market Leader in Disruptive Materials Technology, Completes Business Combination With Artius, Creating First Publicly Traded Pure Play Carbon Negative Materials Company

June 25, 2021

Origin's common stock to begin trading on the Nasdag under symbol "ORGN" on June 25, 2021

WEST SACRAMENTO, Calif.—(BUSINESS WIRE)—Jun. 25, 2021— <u>Origin Materials, Inc.</u>, the world's leading carbon negative materials company, announced today that it has completed its <u>previously announced business combination</u> with Artius Acquisition Inc. ("Artius") (NASDAQ: AACQ, AACQW).

This press release features multimedia. View the full release here: https://www.businesswire.com/news/home/20210625005071/en/

The transaction was unanimously approved by Artius' Board of Directors and was approved at an extraordinary general meeting of Artius shareholders on June 23, 2021. In connection with the closing, Artius changed its name to Origin Materials, Inc. ("Origin Materials," "Origin," or the "Company"). The ticker symbols for the common stock and warrants of Origin will change from "AACQ" and "AACQW" to "ORGN" and "ORGNW", respectively and will begin trading as such on the Nasdaq on June 25, 2021.

## **Company Background**

Founded in 2008, Origin is the world's leading carbon negative materials company with a mission to enable the world's transition to sustainable materials. Origin's patented drop-in technology, economics and carbon impact are supported by a growing list of major global customers and investors. Origin's technology has been further supported by an ISO-compliant Life Cycle Assessment (LCA) which concluded that Origin's products are expected to be carbon negative when produced at commercial scale.

While an estimated 55% of global carbon emissions come from energy generation and transport, the other 45% come from the production of materials for consumer and industrial products. More than ten million barrels of oil per day are used to create materials, in the process releasing new carbon into the atmosphere. Origin's vision for the future is to replace this oil use with non-food feedstocks and materials while capturing carbon in the process.

Origin Materials' patented technology platform, which turns inexpensive, plentiful, sustainable wood residues into carbon negative materials, can help revolutionize the production of a wide range of end products, including clothing, textiles, plastics, packaging, car parts, tires, carpeting, toys, and more with a ~\$1 trillion addressable market. In addition, Origin Materials' technology platform is expected to provide stable pricing largely decoupled from the petroleum supply chain, which is exposed to more volatility than supply chains based on sustainable wood residues.

Since announcing the business combination with Artius on February 17, 2021, Origin Materials has increased its customer demand from signed offtake agreements and capacity reservations by 90% to \$1.9 billion from the \$1 billion previously disclosed. This increase also greatly expanded the Company's end markets beyond its Fortune Global 500 CPG customers – Danone, Nestlé Waters, PepsiCo – to include industrial and apparel textiles, automotive parts, insulating foams and other applications.

The Company also implemented the following new and expanded partnerships and customer relationships:

- June 15, 2021: <u>Strategic alliance with Palantir Technologies</u> (NYSE:PLTR) to accelerate the world's transition to net zero carbon focused on decarbonizing the global materials supply chain.
- June 10, 2021: <u>Launched Net Zero Automotive program with Ford Motor Company</u> (NYSE:F) focused on industrializing new materials to drive decarbonization in the automotive industry April 27, 2021: Partnership with Mitsubishi Gas Chemical to industrialize advanced carbon negative chemicals and materials for applications in the automotive, medical, food, information and communication, energy, and infrastructure sectors.
- · April 20, 2021: Partnership with Solvay to develop advanced carbon negative materials for the automotive industry.
- April 19, 2021: <u>Strategic alliance with PrimaLoft</u> to develop carbon negative insulating fiber for outdoor gear, bedding and apparel.
   PrimaLoft's iconic brand partners include Patagonia, Stone Island, L.L. Bean, Lululemon, adidas and Nike.
- April 12, 2021: <u>Partnership with Packaging Matters</u> to advance carbon negative packaging solutions, building on an existing 10-year supply agreement.
- April 6, 2021: Partnership with AECI Much Asphalt to develop low-carbon asphalt.
- April 5, 2021: Partnership with AEC SANS Technical Fiber to develop carbon negative materials for apparel and automotive applications.

## **Management Commentary:**

John Bissell, Co-Founder and Co-CEO of Origin Materials, commented, "We are excited to complete this transaction, which will further accelerate our growth and mission of enabling the world's transition to sustainable, carbon negative materials. The materials supply chain accounts for nearly half of all global carbon emissions, creating a significant opportunity to decarbonize materials that are used in products across a wide range of end markets. We believe our breakthrough technology provides countless companies that make physical products with the ability to drastically reduce their emissions in a cost-effective manner."

Rich Riley, Co-CEO of Origin Materials, commented, "Our strategic customer base has grown dramatically, driven by new partnerships with industry-leading companies that are at the forefront of decarbonization and corporate sustainability. With the capital from this transaction, we are well positioned to scale our technology platform and help our customers meet their decarbonization and ESG goals. We are energized by the exciting opportunities that lie ahead as we commercialize our technology and transition to a world of net zero carbon."

Boon Sim, Chief Executive Officer of Artius, commented, "We pursued a merger with Origin Materials because we wanted to make sure that we invested in a company with technology that has been tested and validated. We conducted significant due diligence evaluating their technology and their expanded customer base is further validation of their technology's ability to help the world transition to net zero carbon. With their breakthrough technology and significant customer base we are highly confident in Origin's ability to disrupt the materials supply chain and enable their customers' decarbonization goals, which we believe will deliver significant long-term shareholder value."

#### **Transaction Overview**

The transaction previously valued at approximately \$1 billion is supported by existing and new investors, including Danone, Nestlé Waters, PepsiCo, Mitsubishi Gas Chemical and AECI, as well as certain funds and accounts managed by Sylebra Capital, Senator Investment Group, Electron Capital Partners, BNP Paribas AM Energy Transition Fund and affiliates of Apollo. Origin Materials will use the proceeds to fund the Company's planned growth, including the completion and commissioning of its first commercial plant, which is expected to be operational in 2022, and of its second commercial plant, which is expected to be operational in 2025.

Origin's senior management team will continue to lead the combined company. The Board of Directors of the combined company will consist of John Bissell, Benno O. Dorer, Charles Drucker, Kathleen B. Fish, William Harvey, Pia Heidenmark Cook, Karen Richardson, Rich Riley and Boon Sim.

A more detailed description of the transaction can be found in the definitive proxy statement/prospectus filed with the SEC on May 27, 2021.

#### Advisors

BofA Securities served as exclusive financial advisor to Origin, and Cooley LLP served as legal advisor to Origin. Credit Suisse and Goldman Sachs & Co. LLC served as joint financial and capital markets advisors to Artius and served as co-placement agents. Cleary Gottlieb Steen & Hamilton LLP served as legal advisor to Artius.

## **About Origin Materials**

Headquartered in West Sacramento, Origin Materials is the world's leading carbon negative materials company. Origin Materials' mission is to enable the world's transition to sustainable materials. Over the past 10 years, Origin Materials has developed a platform for turning the carbon found in non-food biomass into useful materials, while capturing carbon in the process. Origin Materials' patented drop-in core technology, economics and carbon impact are supported by a growing list of major global customers and investors. Origin Materials' first commercial plant is expected to be operational in 2022 with a second commercial plant expected to be operational in 2025 and plans for additional expansion over the next decade.

For more information, visit www.originmaterials.com.

## **About Artius Acquisition Inc.**

Artius was a special purpose acquisition company formed for the purpose of effecting a merger, share exchange, asset acquisition, share purchase, reorganization or similar business combination with one or more businesses. Artius was co-founded by Charles Drucker, the former CEO of WorldPay, Inc., a leading payments company, and its predecessor company, Vantiv. Inc., and Boon Sim, the Founder and Managing Partner of Artius Capital Partners LLC.

For more information, visit <a href="https://www.artiuscapital.com/acquisition">https://www.artiuscapital.com/acquisition</a>.

# **Cautionary Note on Forward-Looking Statement**

This press release contains certain forward-looking statements within the meaning of the federal securities laws. Forward-looking statements generally are accompanied by words such as "believe," "may," "will," "estimate," "continue," "anticipate," "intend," "expect," "should," "would," "plan," "predict," "potential," "seem," "seek," "future," "outlook," and similar expressions that predict or indicate future events or trends or that are not statements of historical matters. These forward-looking statements include, but are not limited to, statements regarding Origin Materials' business strategy, estimated total addressable market, commercial and operating plans and product development plans. These statements are based on various assumptions, whether or not identified in this press release, and on the current expectations of the management of Origin Materials and are not predictions of actual performance. These forward-looking statements are provided for illustrative purposes only and are not intended to serve as, and must not be relied on as, a guarantee, an assurance, a prediction, or a definitive statement of fact or probability. Actual events and circumstances are difficult or impossible to predict and will differ from assumptions. Many actual events and circumstances are beyond the control of Origin Materials. These forward-looking statements are subject to a number of risks and uncertainties, including that Origin Materials may be unable to successfully commercialize its products; the effects of competition on Origin Materials' business; disruptions and other impacts to Origin Materials' business as a result of the COVID-19 pandemic and other global health or economic crises; changes in customer demand; failure to realize the anticipated benefits of the business combination; and those factors discussed in the definitive proxy statement/prospectus filed with the SEC on May 27, 2021 under the heading "Risk Factors," and other documents Origin Materials has filed, or will file, with the SEC. If any of these risks materialize or our assumptions prove incorrect, actual results could differ materially from the results implied by these forward-looking statements. There may be additional risks that Origin Materials presently does not know, or that Origin Materials currently believes are immaterial, that could also cause actual results to differ from those contained in the forward-looking statements. In addition, forward-looking statements reflect Origin Materials' expectations, plans, or forecasts of future events and views as of the date of this press release. Origin Materials anticipates that subsequent events and developments will cause its assessments to change. However, while Origin Materials may elect to update these forward-looking statements at some point in the future, Origin Materials specifically disclaim any obligation to do so. These forward-looking statements should not be relied upon as representing Origin Materials' assessments of any date subsequent to the date of this press release. Accordingly, undue reliance should not be placed upon the forward-looking statements.

View source version on <u>businesswire.com</u>: <u>https://www.businesswire.com/news/home/20210625005071/en/</u>

Origin Materials
Investors:

<u>ir@originmaterials.com</u>

Media:

media@originmaterials.com

Source: Origin Materials and Artius Acquisition Inc.

#### UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION

#### Introduction

In connection with the consummation of the business combination on June 25, 2021 pursuant to that certain Agreement and Plan of Merger and Reorganization (the "*Merger Agreement*"), dated as of February 16, 2021, by and among Origin Materials, Inc. (f/k/a Artius Acquisition Inc.) ("*Artius*" or the "*Company*"), Zero Carbon Merger Inc., a Delaware corporation and direct, wholly owned subsidiary of the Company ("*Merger Sub*"), and Micromidas, Inc., a Delaware corporation doing business as Origin Materials ("*Legacy Origin*") (the "*Business Combination*"), the registrant changed its name from Artius Acquisition Inc. to Origin Materials, Inc.

The following unaudited pro forma condensed combined balance sheet as of March 31, 2021 and the unaudited pro forma condensed combined statements of operations for the year ended December 31, 2020 and for the three months ended March 31, 2021 present the combination of the financial information of the Company and Legacy Origin after giving effect to the following transactions:

- the Business Combination, which involved the holders of 43,880,956 shares of the Company's Class A Ordinary Shares exercising their right to redeem their shares for cash at a redemption price of approximately \$10.0040;
- the purchase by certain purchasers of an aggregate of 20,000,000 shares of common stock of the Company ("Common Stock") for a purchase price of \$10.00 per share and pursuant to separate subscription agreements dated February 16, 2021 (the "PIPE");
- the purchase by certain funds managed by affiliates of Apollo Capital Management, L.P. of 3,000,000 shares of Common Stock for a purchase price of \$10.00 per share and pursuant to a purchase agreement dated June 14, 2021 (the "*Backstop Transaction*"); and
- the purchase by certain purchasers of an aggregate of 1,300,001 shares of Common Stock for a purchase price of \$10.00 per share and pursuant to a purchase agreement dated June 23, 2021 (the "Additional Subscription") and certain other adjustments.

Collectively these transactions are referred to as the pro forma adjustment, as described in the accompanying notes. The Company and Legacy Origin are collectively referred to herein as the "Companies," and the Companies, subsequent to the Business Combination, the PIPE, the Backstop Transaction and the Additional Subscription, are referred to herein as the "Combined Company".

On February 16, 2021, the Company, Merger Sub, and Legacy Origin entered into the Merger Agreement, which provided that Legacy Origin would be merged with and into Merger Sub, with Legacy Origin surviving the Merger. After giving effect to the Business Combination, the Company owned, directly or indirectly, all of the issued and outstanding equity interests of Legacy Origin and its subsidiaries and became the Combined Company. Legacy Origin equity holders hold a portion of the Common Stock of the Combined Company.

The unaudited pro forma condensed combined balance sheet as of March 31, 2021 combines the historical balance sheet of Artius as of March 31, 2021 with the historical balance sheet of Legacy Origin as of March 31, 2021 on a pro forma basis as if the Business Combination and the other related events contemplated by the Merger Agreement, summarized below, had been consummated on March 31, 2021.

The unaudited pro forma condensed combined statement of operations for the year ended December 31, 2020 and for the three months ended March 31, 2021 combines the historical audited condensed statement of operations of Artius for the period from January 24, 2020 (inception) to December 31, 2020 and the unaudited condensed statement of operations of Artius for the three months ended March 31, 2021 with the historical audited condensed consolidated statement of operations of Legacy Origin for the year ended December 31, 2020 and the unaudited condensed consolidated statement of operations of Legacy Origin for the three months ended March 31, 2021 on a pro forma basis as if the Business Combination and the other related events contemplated by the Merger Agreement, summarized below, had been consummated on January 1, 2020, the beginning of the earliest period presented.

The unaudited pro forma condensed combined financial information was derived from and should be read in conjunction with the following historical financial statements and the accompanying notes, which are included in the Company's final prospectus and definitive proxy statement, dated May 27, 2021 (the "*Prospectus*"), and filed with the Securities and Exchange Commission (the "*SEC*"):

- the historical unaudited consolidated financial statements of Legacy Origin as of and for the three months ended March 31, 2021;
- the historical unaudited financial statements of Artius as of and for the three months ended March 31, 2021;
- the historical audited condensed financial statements of Artius as of December 31, 2020 and for the period from January 24, 2020 (inception) to December 31, 2020;
- the historical audited consolidated financial statements of Legacy Origin as of and for the year ended December 31, 2020 and the historical audited consolidated financial statements of Legacy Origin as of and for the year ended December 31, 2019; and
- other information relating to Artius and Legacy Origin included in the Prospectus, including the Merger Agreement and the description of certain terms thereof set forth under the sections titled "The Merger Agreement and Related Agreements" and "The Business Combination."

The unaudited pro forma condensed combined financial information should also be read together with the section titled "Artius Management's Discussion and Analysis of Financial Condition and Results of Operations," "Origin Management's Discussion and Analysis of Financial Condition and Results of Operations," and other financial information included in the Prospectus.

#### Other Related Events in connection with the Business Combination

Other related events that took place in connection with the Business Combination are summarized below:

- the PIPE, the Backstop Transaction and the Additional Subscription;
- during the period the Earnout Shares (as defined in the Prospectus) may be issued, Artius Acquisition Partners LLC will subject 4.5 million shares issued and outstanding of common stock of the Combined Company ("Combined Company Common Stock"), comprised of three separate and equal tranches, to potential forfeiture to the Company for no consideration until the occurrence of the respective earnout triggering events as more fully described in the Prospectus (the "Sponsor Vesting Shares"); and
- during the period Earnout Shares may be issued, the Combined Company may issue to certain holders of Legacy Origin's securities up to 25 million shares of additional Combined Company Common Stock, comprised of three separate and equal tranches, upon the occurrence of the respective earnout triggering events as more fully described in the Prospectus. The issuance of the Earnout Shares would dilute all Combined Company common stock outstanding at that time.

## **Accounting Treatment of the Business Combination**

The Business Combination was accounted for as a reverse recapitalization in accordance with GAAP. Under this method of accounting, Artius is treated as the "acquired" company for financial reporting purposes. Accordingly, for accounting purposes, the financial statements of the Combined Company will represent a continuation of the financial statements of Legacy Origin with the Business Combination treated as the equivalent of Legacy Origin issuing stock for the net assets of Artius, accompanied by a recapitalization. The net assets of Artius will be stated at historical cost, with no goodwill or other intangible assets recorded. Operations prior to the Business Combination will be those of Legacy Origin in future reports of the Combined Company.

Legacy Origin has been determined to be the accounting acquirer based on evaluation of the following facts and circumstances:

- The board of directors of the Company initially has nine members, and Legacy Origin had the ability to nominate the three of the members
  of the board of directors;
- Legacy Origin's senior management comprises the senior management roles of the Company and is responsible for the day-to-day operations;
- the Company has assumed the "doing business as" name of Legacy Origin; and
- The intended strategy and operations of the Company continues Legacy Origin's strategy and operations as a carbon negative materials company with a mission to enable the world's transition to sustainable materials.

The Sponsor Vesting Shares and the Earnout Shares are accounted for as liability classified equity instruments that are earned upon achieving the applicable earnout triggering events, which include events that are not indexed to the common stock of Legacy Origin.

### **Basis of Pro Forma Presentation**

The unaudited pro forma condensed combined financial information has been prepared in accordance with Article 11 of Regulation S-X. The adjustments in the unaudited pro forma condensed combined financial information have been identified and presented to provide relevant information necessary for an illustrative understanding of the Combined Company upon consummation of the Business Combination in accordance with GAAP. Assumptions and estimates underlying the unaudited pro forma adjustments set forth in the unaudited pro forma condensed combined financial information are described in the accompanying notes.

The unaudited pro forma condensed combined financial information has been presented for illustrative purposes only and is not necessarily indicative of the operating results and financial position that would have been achieved had the Business Combination, the PIPE, the Backstop Transaction and the Additional Subscription occurred on the dates indicated, and does not reflect adjustments for any anticipated synergies, operating efficiencies, tax savings or cost savings. Any cash proceeds remaining after the consummation of the Business Combination and the other related events contemplated by the Merger Agreement are expected to be used for general corporate purposes. The unaudited pro forma condensed combined financial information does not purport to project the future operating results or financial position of the Combined Company following the completion of the Business Combination. The unaudited pro forma adjustments represent management's estimates based on information available as of the date of these unaudited pro forma condensed combined financial information and are subject to change as additional information becomes available and analyses are performed. Artius and Legacy Origin have not had any historical relationship prior to the transactions. Accordingly, no pro forma adjustments were required to eliminate activities between the companies.

The following summarizes the pro forma Combined Company Common Stock issued and outstanding immediately after the Business Combination, excluding the Sponsor Vesting Shares:

	Shares	%
Legacy Origin - Combined Company Common Stock	70,266,925	48.6%
Legacy Origin, options	7,945,956	5.5%
Total Origin Business Combination shares	78,212,881	54.1%
Artius public shareholders	28,569,044	19.7%
Holders of Artius sponsor shares	13,612,500	9.4%
PIPE Investors and Backstop Investors	24,300,001	16.8%
Pro Forma Common Stock	144,694,426	100.0%

# UNAUDITED PRO FORMA CONDENSED COMBINED BALANCE SHEET AS OF MARCH 31, 2021 (in thousands)

ASSETS		As of Mar	ch 31, 2021			
ASSETS		Artius		Transaction Accounting		Pro Forma
Carein Cash					Note 3	Combined
Cash						
Casa   Casa		ф 250	ф 0.070	ф <u>БО</u> 4 ББО	( )	ф. 450, 400
Restricted cash	Cash	\$ 359	\$ 8,8/2			\$ 473,423
Restricted cash						
Restricted cash						
Restricted cash						
Content receivable	Restricted cash	<u>—</u>	565	(555,565)	(1)	565
Total Current Assets		_		<u>_</u>		27
Pepale spesses and other current assets		_		_		
Total Current Assets		242	147	_		389
Property plant and equipment net			9 611	464 192		474,404
Intangible assets, net						46,356
Cash and securities held in Trust Account         724,779 (26,730)         C(24,779) (26)         C(25,300)         56,217         (260,307)         521           LIABILITIES, REDEEMABLE CONVERTIBLE PREFERRED STOCK AND SHAREHOLDERS' EQUITY (DEPICIT)         Current Liabilities           Accrued expenses         166         1,443         (359)         ()         1           Accrued expenses         166         1,443         (359)         ()         0           Derivative liability         61,020         3,826         (3,856)         ()         0           Shareholder convertible note payable         2,738         (2,742)         ()         0           Pomissory note - related party         2         (2,742)         (4,9433)         62           Total Current Liabilities         25,358         20,742         (4,9433)         62           PPD Ison         -         906         -         -           Canadian Government Research and Development Program Liability         -         6,271         -         6           Related party other liabilities, long-term         -         5,189         -         5           Related party other liabilities, long-term         -         2,500         -         2           Camout Liabilitie		<u> </u>		<u> </u>		250
Total Assets		724 779		(724 779)	(a)	
LABLITIES, REDERMABLE CONVERTIBLE PREFERRED STOCK AND SHAREHOLDERS' EQUITY (DEFICIT)   Current Liabilities					(u)	521,010
STOCK AND SHAREHOLDERS' EQUITY (DEFICIT)   Carrent Liabilities   2		723,300	50,217	(200,307)		321,010
Accounts payable 2 2,731 (2,556) (c) Account expenses 166 1,443 (359) () 1 Account expenses 166 1,443 (359) () 1 Derivative liability 6 1,020 38,26 (3,826) () 61 Shareholder convertible note payable — 12,742 (12,742) () Promisory note related party — — — — — — — — — — — — — — — — — — —	STOCK AND SHAREHOLDERS' EQUITY (DEFICIT)					
Accused expenses 166 1,443 (359) () 1 1 (90) (c) C			0.50	/2 == C		
Derivative liability						177
Derivative liability   61,020   3,826   (3,826)   (1)   61	Accrued expenses	166	1,443			1,160
Shareholder convertible note payable   -   12,742   (12,742   0)   Promissory note - related party   -   -   -   -     -	Decl. (1919)	C1 020	2.026			C1 020
Promissory note - related party   25.358						61,020
Deferred underwriting fee payable         25.58 b (20.742) (24.931) (24.931) (25.38) (20.742) (44.931) (26.271) (20.742) (26.271) (26			12,/42	(12,/42)	(J)	_
Total Current Liabilities			_	(25.250)	4.)	_
PPP Loan         —         906         —         6           Canadian Government Research and Development Program Liability         —         6,271         —         6           Redeemable convertible preferred stock warrants liability         —         67,342         (K)         (K)           Stockholder note         —         5,558         —         5           Earnout Liability         —         —         206,746         (I)         244           melated party other liabilities, long-term         —         —         387         (o)         —         2           Cother liabilities, long-term         —         2,5500         —         —         2         2         7         2         7         2         7         2         7         2         7         2         32         7         32         7         32         7         32         32         7         32         32         7         32         32         7         32         32         7         32         32         4         32         4         32         4         32         4         32         4         4         32         4         4         32         4         4					(b)	
Canadian Government Research and Development Program Liability         — 6,271         — 6         6 6,342         (6)         (8)         6 6,342         (8)         (8)         5 66         1 5 89         — 5         5 568         — 5         5 568         — 5         5 568         — 5         5 568         — 5         5 568         — 206,746         (1)         2 44         (10)         2 44         (10)         2 44         (10)         2 44         (10)         2 42         2 26,746         (1)         2 44         (10)         2 42         2 26,746         (1)         2 44         (10)         2 44         (10)         2 44         (10)         2 44         (10)         2 44         (10)         2 44         2 43         3 27         3 24         4 1,125         1 1,125         1 1,125         1 1,125         1 1,125         1 1,125         1 1,125		86,546		(44,931)		62,357
Redeemable convertible preferred stock warrants liability		_		_		906
Stockholder note   —   5,189   —   5   5   5   5   5   5   5   5   5						6,271
Related party other liabilities, long-term		_		(67,342)	(k)	
Earmout Liability         —         —         206,746         (1)         244           Cher liabilities, long-term         —         2,500         —         2           Chat Liabilities         86,546         108,518         313,574         327           Committees           Class A ordinary share subject to possible redemption         633,834         —         (633,834)         (e)           Redeemable convertible preferred stock, Series A         —         31,478         (f)         —           Redeemable convertible preferred stock, Series B         —         41,125         (41,125)         (f)         —           Redeemable convertible preferred stock, Series C         —         23,380         (23,380)         (f)         — <td></td> <td>_</td> <td></td> <td>_</td> <td></td> <td>5,189</td>		_		_		5,189
Common stock		_	•	200 540	(1)	5,568
Other liabilities, long-term         —         2,500         —         2           Total Liabilities         86,546         108,518         132,574         327           Commitments         Class A ordinary share subject to possible redemption         633,834         —         (633,834)         (e)           Redeemable convertible preferred stock, Series A         —         31,478         (f)         —           Redeemable convertible preferred stock, Series B         —         41,125         (f)         —           Redeemable convertible preferred stock, Series C         —         23,380         (g3,380)         (f)         —           Total Redeemable convertible preferred stock         633,834         95,983         (729,817)         —           Shareholders' Equity (Deficit)         —         1         —         7         (e)         —           Class A ordinary share         1         —         7         (e)         —	Earnout Liability					244,847
Other liabilities, long-term         —         2,500         —         2           Total Liabilities         86,546         108,518         132,574         327           Commitments           Class A ordinary share subject to possible redemption         633,834         —         (633,834)         (e)           Redeemable convertible preferred stock, Series A         —         31,478         (31,478)         (f)           Redeemable convertible preferred stock, Series B         —         41,125         (41,125)         (f)           Redeemable convertible preferred stock, Series C         —         23,380         (23,380)         (f)         —           Total Redeemable convertible preferred stock, Series C         —         23,380         (23,981)         — <td></td> <td>_</td> <td>_</td> <td></td> <td></td> <td></td>		_	_			
Total Liabilities	Other liabilities long term		2 500	007	(0)	2,500
Commitments		00.540		122.574		
Class A ordinary share subject to possible redemption   633,834   — (633,834)   (e)		86,546	108,518	132,5/4		327,638
Redeemable convertible preferred stock, Series A	Commitments					
Redeemable convertible preferred stock, Series B		633,834	_	· · · · · ·		_
Redeemable convertible preferred stock, Series C         —         23,380         (23,380)         (5)           Total Redeemable convertible preferred stock         633,834         95,983         (729,817)         7         (e)           Shareholders' Equity (Deficit)           Class A ordinary share         1         —         7         (e)         2         (f)         2         (g)         2         (d)         (i)         2         (g)         2         (d)         (i)         (i)         (i)         2         —         2         (g)         (g)         2         —	Redeemable convertible preferred stock, Series A	_	31,478	(31,478)	(f)	_
Redeemable convertible preferred stock, Series C         —         23,380         (23,380)         (5)           Total Redeemable convertible preferred stock         633,834         95,983         (729,817)         7         (e)           Shareholders' Equity (Deficit)           Class A ordinary share         1         —         7         (e)         2         (f)         2         (g)         2         (d)         (i)         2         (g)         2         (d)         (i)         (i)         (i)         2         —         2         (g)         (g)         2         —			44.405	(44.405)	(0)	
Total Redeemable convertible preferred stock		_				_
Shareholders' Equity (Deficit)   Class A ordinary share		<u> </u>			(1)	
Class A ordinary share 1 — 7 (e)		633,834	95,983	(/29,81/)		_
Class B ordinary share   2 (f)   2 (g)   2 (d)   (4) (i)   (4) (i)   (5)   (4) (i)   (5) (i)   (6) (i)		1		7	(-)	10
Class B ordinary share   2	Class A ordinary snare	1	_			10
Class B ordinary share   2						
Class B ordinary share 2 — (2) (g)  Common stock — — — — — — — — — — — — — — — — — — —						
Class B ordinary share       2       —       (2)       (g)         Common stock       —       —       —         Additional paid in capital       14,690       3,324       (36,600)       (c)       342         633,827       (e)       95,981       (f)       (e)       (g)						
Common stock  Additional paid in capital  14,690 3,324 33,827 (e) 95,981 (f) (9,693) (h) 199,998 (d) 67,342 (k) 12,742 (j) (206,746) (l) (37,214) (m) 1,380 (n) (887) (o) (395,979) (i)	Class B ordinary share	2	_			_
Additional paid in capital 14,690 3,324 (36,600) (c) 342 633,827 (e) 95,981 (f) (9,693) (h) 199,998 (d) 67,342 (k) 12,742 (j) (206,746) (l) (37,214) (m) 1,380 (n) (887) (o) (395,979) (i)	•	۷	_	(2)	(8)	_
633,827 (e) 95,981 (f) (9,693) (h) 199,998 (d) 67,342 (k) 12,742 (j) (206,746) (l) (37,214) (m) 1,380 (n) (887) (o) (395,979) (i)		14 600	2 224	(26 600)	(a)	— 342,165
95,981 (f) (9,693) (h) 199,998 (d) 67,342 (k) 12,742 (j) (206,746) (l) (37,214) (m) 1,380 (n) (887) (o) (395,979) (i)	Additional paid in Capital	14,050	3,324			342,103
(9,693) (h) 199,998 (d) 67,342 (k) 12,742 (j) (206,746) (l) (37,214) (m) 1,380 (n) (887) (o) (395,979) (i)						
199,998 (d) 67,342 (k) 12,742 (j) (206,746) (l) (37,214) (m) 1,380 (n) (887) (o) (395,979) (i)						
67,342 (k) 12,742 (j) (206,746) (l) (37,214) (m) 1,380 (n) (887) (o) (395,979) (i)						
12,742 (j) (206,746) (l) (37,214) (m) 1,380 (n) (887) (o) (395,979) (i)						
(206,746) (l) (37,214) (m) 1,380 (n) (887) (o) (395,979) (i)						
(37,214) (m) 1,380 (n) (887) (o) (395,979) (i)						
1,380 (n) (887) (o) (395,979) (i)						
(887) (o) (395,979) (i)						
(395,979) (i)						
Accumulated deficit (5,055) (152,455) 5,055 (11) (145	Accumulated deficit	(9,693)	(152,459)	9,693	(h)	(149,654)
4,185 (j)		( , )	, ,,			, , , , , , ,

			(1,3	880) (n)	
Accumulated other comprehensive income (loss)		851		<u>—</u>	851
Total Shareholders' Equity (Deficit)	5,000	(148,284)	336,6	556	193,372
TOTAL LIABILITIES, REDEEMABLE CONVERTIBLE					
PREFERRED STOCK AND SHAREHOLDERS' EQUITY					
(DEFICIT)	\$725,380	\$ 56,217	\$ (260,5	i <u>87</u> )	\$ 521,010

# UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENT OF OPERATIONS FOR THE YEAR ENDED DECEMBER 31, 2020

(in thousands, except share and per share data)

	As of Decer Artius (Historical) (Restated)	mber 31, 2020 Legacy Origin (Historical) (Restated)	Transaction Accounting Adjustments	Note 3	Pro Forma Combined
Operating Expenses:					
Research and development	\$ —	\$ 4,138	\$ 3,956	(cc)	\$ 8,094
General and administrative	3,029	6,563	5,530	(cc)	15,120
			(2)	(ee)	
Depreciation and amortization	_	479	_		479
Total operating expenses	3,029	11,180	9,484		23,693
Loss from operations	(3,029)	(11,180)	(9,484)		(23,693)
Other income (expense)					
Interest expense	_	(342)	175	(aa)	(167)
Interest income	213	_	_		213
Change in fair value of derivative liability	(23,060)	(1,088)	1,088	(dd)	(23,060)
Change in fair value of redeemable convertible preferred stock warrants					
liability	_	(18,498)	19,233	(bb)	735
Other income, net	4	805			809
Total other income (expense)	(22,843)	(19,123)	20,496		(21,470)
Net loss	\$ (25,872)	\$ (30,303)	\$ 11,012		\$ (45,163)
Basic and diluted weighted average shares outstanding - Class A and Class B					136,748,470
Basic and diluted net loss per share - Class A and Class B					\$ (0.33)

# UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENT OF OPERATIONS FOR THE THREE MONTHS ENDED MARCH 31, 2021

(in thousands, except share and per share data)

Overtical Expression	As of Artius (Historical	March 31, 2021 Legacy Ori (Historica		ng		Pro Forma Combined
Operating Expenses:						4.00.4
Research and development	\$ —	\$ 1,3		25 (aa)	\$	1,834
General and administrative	91	2 3,9	48 8	55 (aa)		2,429
			(3,2	86) (ee)		
Depreciation and amortization	_	1	15 –	-		115
Total operating expenses	91	5,3	72 (1,9	06)		4,378
Loss from operations	(91)	2) (5,3)	72) 1,9	06		(4,378)
Other income (expense)						
Interest expense	_	(2)	79) 2	12 (bb)		(67)
Interest income	63	3 –		_		63
Change in fair value of derivative liability	17,029	9 (3)	91) 3	91 (cc)		17,029
Change in fair value of redeemable convertible preferred stock warrants		,	•	` ` `		
liability	_	(48,1	09) 48,1	09 (dd)		_
Other income, net	_	5	81			581
Total other income (expense)	17,09	2 (48,1	98) 48,7	12		17,606
Net loss	\$ 16,18	\$ (53,5)	<del>70</del> ) \$ 50,6	18	\$	13,228
Basic and diluted weighted average shares outstanding - Class A and Class B	-				1	36,748,470
Basic and diluted net loss per share - Class A and Class B					\$	0.10

### Notes to Unaudited Pro Forma Condensed Combined Financial Statements

#### 1. Basis of Presentation

The Business Combination is accounted for as a reverse recapitalization, with no goodwill or other intangible assets recorded, in accordance with GAAP. Under this method of accounting, Artius is treated as the "acquired" company for financial reporting purposes. Accordingly, for accounting purposes, the Business Combination is treated as the equivalent of Legacy Origin issuing stock for the net assets of Artius, accompanied by a recapitalization. The net assets of Legacy Origin will remain at historical cost.

The unaudited pro forma condensed combined balance sheet as of March 31, 2021 gives pro forma effect to the Business Combination as if it had been consummated on March 31, 2021. The unaudited pro forma condensed combined statements of operations for the three months ended March 31, 2021 and the year ended December 31, 2020 give pro forma effect to the Business Combination as if it had been consummated on January 1, 2020.

The unaudited pro forma condensed combined balance sheet as of March 31, 2021 has been prepared using, and should be read in conjunction with, the following:

- Artius's unaudited balance sheet as of March 31, 2021 and the related notes included in the Prospectus; and
- Legacy Origin's unaudited consolidated balance sheet as of March 31, 2021 and the related notes included in the Prospectus.

Management has made significant estimates and assumptions in its determination of the pro forma adjustments. As the unaudited pro forma condensed combined financial information has been prepared based on these preliminary estimates, the final amounts recorded may differ materially from the information presented.

The unaudited pro forma condensed combined financial information does not give effect to any anticipated synergies, operating efficiencies, tax savings or cost savings that may be associated with the Business Combination. The pro forma adjustments reflecting the consummation of the Business Combination are based on certain currently available information and certain assumptions and methodologies that Artius believes are reasonable under the circumstances. The unaudited condensed pro forma adjustments, which are described in the accompanying notes, may be revised as additional information becomes available and is evaluated. Therefore, it is likely that the actual adjustments will differ from the pro forma adjustments and it is possible the difference may be material. Artius believes that its assumptions and methodologies provide a reasonable basis for presenting all of the significant effects of the Business Combination based on information available to management at the time and that the pro forma adjustments give appropriate effect to those assumptions and are properly applied in the unaudited pro forma condensed combined financial information.

The unaudited pro forma condensed combined financial information is not necessarily indicative of what the actual results of operations and financial position would have been had the Business Combination taken place on the dates indicated, nor are they indicative of the future consolidated results of operations or financial position of the Combined Company. They should be read in conjunction with the historical financial statements and notes thereto of Artius and Legacy Origin.

## 2. Adjustments to Unaudited Pro Forma Condensed Combined Financial Information

The unaudited pro forma condensed combined financial information has been prepared to illustrate the effect of the Business Combination and has been prepared for informational purposes only. The historical financial statements have been adjusted in the unaudited pro forma condensed combined financial information to give pro forma effect to events that are (1) directly attributable to the Business Combination, (2) factually supportable, and (3) with respect to the statements of operations, expected to have a continuing impact on the results of the Combined Company. Artius and Legacy Origin have not had any historical relationship prior to the Business Combination. Accordingly, no pro forma adjustments were required to eliminate activities between the companies.

The pro forma combined provision for income taxes does not necessarily reflect the amounts that would have resulted had the Combined Company filed consolidated income tax returns during the periods presented.

## Transaction Accounting Adjustments to Unaudited Pro Forma Condensed Combined Balance Sheet

The adjustments included in the unaudited pro forma condensed combined balance sheet as of March 31, 2021 are as follows:

- (a) Reflects the reclassification of cash held in the Trust Account (as defined in the Prospectus) that becomes available following the Business Combination.
- (b) Reflects the settlement of \$25.4 million in deferred underwriting compensation.
- (c) Represents estimated combined transaction costs to be incurred by Artius and Legacy Origin of approximately \$39.2 million for legal, financial advisory and other professional fees incurred in consummating the Business Combination. Of this amount, \$1.9 million and \$3.3 million was expensed by Legacy Origin and Artius during the year ended December 31, 2020 and period ended March 31, 2021, respectively, and \$1.8 million and \$2.6 million was accrued by Legacy Origin and Artius in Accounts Payable and Accrued Expenses, respectively, as of March 31, 2021.
- (d) Reflects proceeds of \$200.0 million from the issuance and sale of 20,000,000 shares of Artius Class A Common Stock at \$10.00 per share in the PIPE pursuant to the Subscription Agreements.
- (e) Reflects the reclassification of \$634 million of Artius Class A Common Stock subject to possible redemption to permanent equity.
- (f) Reflects the conversion of 21,070,663 shares of Legacy Origin Preferred Stock and 1,285,164 shares of Legacy Origin Common Stock plus conversion of accumulated dividends on redeemable convertible preferred stock and convertible notes payable into 78,212,881 shares of Combined Company Common Stock.
- (g) Reflects the conversion of Artius Class B Common Stock held by the Sponsor into Combined Company Common Stock.
- (h) Reflects the elimination of Artius's historical retained earnings.
- (i) Represents (i) the redemption of 43,880,956 shares of Artius Class A Common Stock by holders validly exercising their right to have such shares redeemed for a full pro rata portion of the trust account holding the proceeds from Artius' initial public offering, calculated as of two business days prior to June 25, 2021, representing \$438,983,333 in the aggregate and (ii) the purchase by certain investors of 4,300,001 shares of Artius Class A Common Stock at a price of \$10.00 per share.
- (j) Reflects conversion of the bridge notes into Class A common stock upon the consumation of the Merger, net of debt discount, elimination of the related derivative liability and accrued interest.
- (k) Reflects the reclassification of Legacy Origin's redeemable convertible preferred stock warrant liability to APIC as a result of the Legacy Origin's redeemable convertible preferred stock warrants being net settled into shares of Legacy Origin common stock immediately prior to the Merger.
- (l) Reflects the preliminary estimated fair value of the Earnout Shares contingently issuable to holders of Origin Capital Stock (as defined in the Prospectus) and vested Origin Stock Options (as defined in the Prospectus) as of the Closing. The preliminary fair values were determined using the most reliable information available. The actual fair values could change materially once the final valuation is determined at the Closing. Refer to Note 4 for more information.
- (m) Reflects the preliminary estimated fair value of the Sponsor Vesting Shares contingently recallable from the Sponsor as of the Closing. The preliminary estimated fair values were determined using the most reliable information available. The actual fair values could change materially once the final valuation is determined at the Closing. Refer to Note 4 for more information.
- (n) Reflects the preliminary estimated fair value of the incremental compensation provided to holders of vested Origin Stock Options which under the terms of the Merger Agreement have been allowed to participate as holders of Origin Capital Stock for the rights to contingently receive a pro rata portion of the Earnout Shares upon achievement of the Earnout Triggering Events. There are no future service requirements related to the Earnout Triggering Events. The preliminary estimated fair values were determined using the Black-Scholes option-pricing model with the most reliable information available as the difference between the fair value of the modified award and the fair value of the original award immediately before it was modified. The actual fair values of the incremental compensation costs are subject to change as additional analyses are performed and such changes could be material once the final valuation as of the modification date is determined.
- (o) Reflects the contingent consideration payable to Legacy Origin's financial advisor a portion of which is triggered at each earnout threshold.

## Adjustments to Unaudited Pro Forma Condensed Combined Statement of Operations

The adjustments included in the unaudited pro forma condensed combined statement of operations for the year ended December 31, 2020 and for the three months ended March 31, 2021 are as follows:

- (aa) Reflects the preliminary estimated fair value of the incremental compensation to be provided to holders of vested Origin Stock Options in the form of certain accommodations made under the terms of the Merger Agreement to permit holders, without requiring exercise of their options, to participate as holders of Origin Capital Stock for the rights to contingently receive a pro rata portion of the Earnout Shares upon achievement of the Earnout Triggering Events. There are no future service requirements related to the Earnout Triggering Events. The preliminary estimated incremental compensation cost is based on the difference between the estimated fair value of the arrangements immediately after the Merger and the fair value of the awards immediately before the Merger based on their original terms. The fair values were estimated based on option-pricing models using the most reliable information and assumptions currently available which includes: volatility of 70%, risk free rate of approximately 0.17% and 0.35% at December 31, 2020 and March 31, 2021, respectively, expected term of 3 years and common stock value as of December 31, 2020 and March 31, 2021 of \$23.54 and \$21.56 per share, respectively. The actual fair values of the incremental compensation costs are subject to change as additional analyses are performed and facts and circumstances change to the closing date of the Merger, and such changes could be material once the final valuation as of the modification date is determined.
- (bb) Reflects the elimination of interest expense related to the bridge notes.
- (cc) Reflects elimination of the change in fair value of the derivative liability upon conversion of the bridge notes into Class A common stock on consumation of the Merger.
- (dd) Reflects the elimination of the redeemable convertible preferred stock warrant liability upon the consumation of the Merger.
- (ee) Reflects the elimination of nonrecurring transaction costs.

## 3. Loss per Share

Represents the net loss per share calculated using the historical weighted average shares outstanding, and the issuance of additional shares in connection with the Business Combination, assuming the shares were outstanding at the beginning of the periods presented. As the Business Combination is being reflected as if it had occurred at the beginning of the periods presented, the calculation of weighted average shares outstanding for basic and diluted net

loss per share assumes that the shares issuable relating to the Business Combination have been outstanding for the entire periods presented. Basic and diluted earnings per share are the same for each class of common stock because they are entitled to the same liquidation and dividend rights.

	Year-Ended December 31, 2020			eriod-Ended March 31, 2021
Pro Forma net loss (in thousands)	\$	(45,163)	\$	13,228
Weighted average shares outstanding, basic and diluted - Class A				
and Class B	13	6,748,470	1	36,748,470
Net loss per share, basic and diluted - Class A and Class B	\$	(0.33)	\$	0.10
Weighted average shares calculation, basic and diluted - Class A				
and Class B				
Artius public stockholders - Class A	2	28,569,044		28,569,044
Holders of Artius sponsor shares - Class A	1	13,612,500		13,612,500
PIPE Investors and Backstop Investors - Class A	2	4,300,001		24,300,001
Legacy Origin stockholders - Class A (1)	7	70,266,925		70,266,925
	13	136,748,470		36,748,470

### 4. Sponsor Vesting Shares and Earnout Shares

The Sponsor Vesting Shares and Earnout Shares are expected to be accounted for as liability classified equity instruments that are earned upon the achieving the Earnout Triggering Events (as defined in the Prospectus), which include events that are not indexed to the Combined Company Common Stock. The preliminary estimated fair value of the Sponsor Vesting Shares and Earnout Shares is \$37.2 million and \$206.7 million, respectively.

The estimated fair values of the Sponsor Vesting Shares and Earnout Shares were determined by using a Monte Carlo simulation valuation model using a distribution of potential outcomes on a monthly basis over the three-year Earnout Period (as defined in the Prospectus). The preliminary estimated fair values of Sponsor Vesting Shares and Earnout Shares were determined using the most reliable information available. Assumptions used in the preliminary valuation, which are subject to change at the Closing, were as follows:

Current stock price: the current stock price was set at the deemed value of \$23.54 per share for Combined Company Common Stock.

*Expected volatility*: the volatility rate was determined by using an average of historical volatilities of selected industry peers deemed to be comparable to our business corresponding to the expected term of the awards.

*Risk-free interest rate*: The risk-free interest rate is based on the U.S. Treasury yield curve in effect at the time of issuance for zero-coupon U.S. Treasury notes with maturities corresponding to the expected three year term of the Earnout Period.

Expected term: The expected term is the three year term of the Earnout Period.

*Expected dividend yield*: The expected dividend yield is zero as we have never declared or paid cash dividends and have no current plans to do so during the expected term.

The actual fair values of Sponsor Vesting Shares and Earnout Shares are subject to change as additional information becomes available and additional analyses are performed and such changes could be material once the final valuation is determined at the Closing.

### COMPARATIVE SHARE INFORMATION

The comparative per share information sets forth summary historical per share information for Artius and Legacy Origin and unaudited pro forma condensed combined per share information after giving effect to the Business Combination.

The selected unaudited pro forma condensed combined book value information as of March 31, 2021 and for the year ended December 31, 2020 give pro forma effect to the Business Combination as if it had occurred on March 31, 2021 and December 31, 2020, respectively. The selected unaudited pro forma condensed combined net income (loss) per share and weighted average shares outstanding information for the three months ended March 31, 2021 and for the year ended December 31, 2020 give pro forma effect to the Business Combination as if it had occurred on January 1, 2020.

This information is only a summary and should be read in conjunction with the historical financial statements of Artius and Legacy Origin and related notes included elsewhere in the Prospectus. The unaudited pro forma condensed combined per share information of Artius and Legacy Origin is derived from, and should be read in conjunction with, the unaudited pro forma condensed combined financial information and related notes included herein in the section titled "Unaudited Pro Forma Condensed Combined Financial Information."

The unaudited pro forma condensed combined income (loss) per share information below does not purport to represent the income (loss) per share which would have occurred had the companies been combined during the periods presented, nor earnings per share for any future date or period. The unaudited pro forma condensed combined book value per share information below does not purport to represent what the value of Artius and Legacy Origin would have been had the companies been combined during the periods presented.

As of and for the Year Ended December 31, 2020 (Restated) (3)	Legacy Artius Origin (Historical) (Historical)		Origin	Combined Pro Forma	Origin equivalent pro forma per share data (2)
Book value per share (1)	\$	0.17	(74.74)	1.20	2.54
Weighted average common shares outstanding—basic and diluted	21,242	2,273	1,285,202	136,748,470	2,869,441
Net loss per share—basic and diluted	\$ (	(1.23)	(23.58)	(0.33)	\$ (15.74)
As of and for the Period Ended March 31, 2021 (3)					
Book value per share (1)	\$	0.16	(110.58)	1.41	2.99
Weighted average common shares outstanding—basic and diluted	28,815	5,514	1,355,762	136,748,470	2,869,441
Net loss per share—basic and diluted	\$ (	(0.56)	(39.51)	0.10	\$ 4.61

- (1) Book value per share = Total equity/shares outstanding.
- (2) The equivalent pro forma basic and diluted per share data for Legacy Origin is based on the exchange ratio set forth in the Business Combination Agreement.
- (3) There were no cash dividends declared in the period presented.